D-PorCCA, a new tool to study the acoustic life of harbour porpoises



Melania Cosentino Department of Electronic and Electrical Engineering

University of Strathclyde

A thesis submitted for the degree of

Doctor of Philosophy

2020

In dedication to my younger self

Acknowledgements

I would like to express my sincere gratitude to my supervisor James Windmill for his support and encouragement to pursue a project I was happy with, and to extend my gratitude to Joe Jackson.

I would also like to thank my colleagues in the Centre for Ultrasonic Engineering, especially Francesco Guarato and Jose Guerreiro who were always patient and willing to help.

This project would not have been possible without the financial support of the Engineering and Physical Sciences Research Council and the European Research Council under the European Union's Seventh Framework Programme. I would like to thank David Nairn and the Clyde Porpoise Project, Joanna Sarnocinska, G. Patterson, P. Nichols, W. and M. Rudd, and the Namibian Dolphin Project for providing data for this project.

I would also like to thank Dr. Jakob Tougaard (Aarhus University, Denmark) and Dr. Mariano Coscarella (Centro Nacional Patagonico, Argentina) for their input, support, and enthusiasm. And to Paula Ramirez for her invaluable help.

Last but not least, I would like to thank my family and friends for their love and support. Sonita, gracias por estar ahi. Por escucharme. Por apoyarme. Por todo.

Abstract

Despite decades of research, the social life of harbour porpoises in their natural environment is almost entirely unknown, although recent data suggest they are very social. Harbour porpoises produce only highly stereotyped clicks (narrow-band highfrequency -NBHF-) that are ideal for monitoring purposes. These are emitted in click trains: a series of clicks with regular or gradually changing inter-click intervals. The patterns in repetition rates are indicative of the behaviour of the clicking porpoise, which can be broadly divided into three main categories: orientation or travelling, feeding, and socialising. This means passive acoustic monitoring (PAM) methods can be used to study the behaviour of harbour porpoises. In this project, a series of algorithms to study the behaviour of NBHF species from acoustic recordings were developed and put together in a standalone application, D-PorCCA. The algorithms were developed and tested using data from Scotland and Denmark collected using static and mobile PAM devices. D-PorCCA and all algorithms were developed in MATLAB. The application includes a user-friendly interface, a simple transientsound detector, a new porpoise click classifier (PorCC), and algorithms to identify behaviours. Additionally, the acoustic behaviour of harbour porpoises was studied while interacting with a solitary short-beaked common dolphin who inhabits the

Firth of Clyde (Scotland). The performance the algorithms in D-PorCCA was tested against manual labelling, varying from 69.2% to 98.5% of accuracy. These steps are automated and provide researchers with pre-selected data and summary data such as 'positive porpoise minute' and day/night activities. D-PorCCA has many functionalities and the user can easily inspect and verify the data. The main results of this project are four. First, evidence was found of interspecies communication between a solitary short-beaked common dolphin and harbour porpoises in the Firth of Clyde (Scotland). Second, a series of patterns consistent with social calls reported in the literature as well as new ones were found for the first time using PAM in the wild. The most striking patterns were those known as 'phrases', which are a series of similar calls produced in a short period of time. Third, the feeding and socialising patterns overlap, limiting our ability to distinguish between them. Lastly, the algorithms can be used in other species that produce NBHF clicks. These results suggest D-PorCCA can potentially be used for behavioural studies of wild harbour porpoises as well as other NBHF species, and to fill knowledge gaps in our understanding of the behaviour of these elusive species. Moreover, it has potential for application in large monitoring project, such as the Static Acoustic Monitoring of the Baltic Sea Harbour Porpoise, known as the SAMBAH Project.

Contents

1	Intr	roduction	1
	1.1	Graphical abstract	1
	1.2	Background	2
	1.3	Motivation for this work	3
	1.4	Aims and objectives of research	4
	1.5	Contribution to knowledge	6
	1.6	Dissemination of results	6
	1.7	Thesis outline	9
2	Lite	erature Review	12
2	Lit e 2.1	erature Review	12 12
2	Lite 2.1 2.2	erature Review Introduction	12 12 12
2	Lite 2.1 2.2 2.3	Prature Review Introduction	12 12 12 21
2	Lite 2.1 2.2 2.3 2.4	Prature Review Introduction	 12 12 12 21 51
23	Lite 2.1 2.2 2.3 2.4 Dat	Parature Review Introduction Introduction Introduction Background Introduction The harbour porpoise Introduction Available tools Introduction	 12 12 21 51 54
23	Lite 2.1 2.2 2.3 2.4 Dat 3.1	erature Review Introduction Introduction Introduction Background Introduction The harbour porpoise Introduction Available tools Introduction a description Introduction Study areas and data collection Introduction	 12 12 21 51 54 65

	3.3	Derived Data	71
4	Por	CC: the harbour porpoise click classifier	75
	4.1	Introduction	75
	4.2	Methods	78
	4.3	Results and Discussion	88
	4.4	Applications	96
	4.5	Summary	97
5	Pat	terns of click trains	98
	5.1	Introduction	98
	5.2	Material and Methods	100
	5.3	Results and discussion	108
	5.4	Limitations	118
	5.5	Summary	120
6	Acc	oustic behaviour	122
	6.1	Introduction	122
	6.2	Orientation	124
	6.3	Social calls	128
	6.4	Overlaps	136
	6.5	Summary and implications	137
7	Inte	erspecies communication	139
	7.1	Introduction	139
	7.2	Methods	141
	7.3	Results and discussion	146

	7.4	Summary	. 169
8	D-F	PorCCA - from acoustics to behaviour	170
	8.1	Introduction	. 170
	8.2	Description	. 171
	8.3	Other applications	. 181
9	Con	nclusions	190
	9.1	Overview of the Thesis	. 190
	9.2	Summary of Findings	. 192
	9.3	Implications of the Research	. 194
	9.4	Recommendations for Future Research	. 196
Bi	bliog	graphy	198
A	ppen	dix A	241
Al	ppen	dix B	250
Aj	ppen	dix C	257

Chapter 1

Introduction

1.1 Graphical abstract



Figure 1.1: Schematic of this project.

1.2 Background

Harbour porpoises (*Phocoena phocoena*) are found in temperate and cold waters in the Northern Hemisphere, in both coastal and off-shore areas [1]. Despite decades of research in the wild and in captive settings, their social lives is still a mystery. They are small and are usually seen in groups of two or three individuals or alone, rarely gathered in large groups [2, 3]. This seemingly solitary life and the fact that they spend 95% of their time underwater [4] limit our ability to study harbour porpoises in the wild using typical visual methods.



Figure 1.2: Harbour porpoises (*Phocoena phocoena*) (©Solvin Zankl)

Luckily for researchers, harbour porpoises are very vocal and produce basically one type of sound: highly stereotyped narrow-band high-frequency (NBHF, *sensu* [5])) clicks, which are ideal for monitoring purposes. These NBHF clicks have peak and centroid frequency between 100 and 160 kHz, centred at 130 kHz, no spectral energy below 100 kHz, and duration between 50 and 175μ s [6, 7]. These clicks are emitted in 'click trains', a series of clicks with regular inter-click interval or that changes gradually [8]. The repetition rate is indicative of the behaviour of the porpoise, which can be divided into three broad categories: orientation, foraging, and socialising. While the patterns for orientation (*i.e.*, repetition rates below 100 clicks per second) and foraging (*i.e.*, low repetition at first, ending in a high-repetition buzz) are known and understood [*e.g.*, 9], the vocalisation patterns during social interactions is almost entirely unknown.

A few studies conducted with animals in captivity and in the wild suggest harbour porpoises have specific calls (*i.e.*, click repetition rates patterns) for different social situations. Authors described several calls, including distress calls, mating calls, and contact call between a mother and her calf [10, 11, 12].

1.3 Motivation for this work

Recent data indicate that harbour porpoises are more social than visual surveys would suggest, spending almost all of their time within hearing distance of other porpoises [13]. Additionally, new technological advancements have made it possible to collect large amounts of acoustic data in a non-invasive and cost-effective manner. This opens the door for studies focused on the social lives of harbour porpoises, for which new analytical tools are necessary.

Harbour porpoises belong to a group of species commonly known as 'NBHF' cetaceans that includes at least 16 species of whales, dolphins, and porpoises that produce NBHF clicks. These species have in common their seemingly lack of social lives. Harbour porpoises are the best studied of these species and thus make an excellent model to understanding these less known species. In addition, parties of the 'Agreement on the Conservation of Small Cetaceans of the Baltic, North East Atlantic, Irish and North Seas' (ASCOBANS), including the UK, and are obliged to develop and adopt measures to protect harbour porpoises, as well as to acquire

knowledge about their occurrence, abundance, and distribution. Moreover, the harbour porpoise is listed in the Annex II of the Habitats Directive (Council Directive 92/43/EC), which requires member states to set up Special Areas for Conservation (SACs) for them [14].

Effective conservation measures for harbour porpoises, require reliable data on distribution and abundance to minimise or eliminate the overlap with anthropogenic activities that threaten harbour porpoise survival. In other words, we need to know where they are and how many there are. Furthermore, knowledge of how porpoises use their environment is crucial to identifying key areas for them, such as breeding areas, that require special protection.

There is a large amount of acoustic recordings of harbour porpoises being collected throughout their distribution range that could be used to fill gaps in our understanding of their ecology and social behaviour in the wild. New analytical tools that are able to detect and classify harbour porpoise clicks accurately, as well as finding patterns in their repetition rate that can be used to identify underlying behaviours are warranted.

1.4 Aims and objectives of research

The necessary tools to study harbour porpoise behaviour from acoustic recordings must be able to automatically inspect hundreds of hours of recordings and provide researchers access to pre-selected acoustic events for further analysis. Such tools would highly reduce the time needed for data analysis and will allow to carry out behaviour studies. For this reason, the overall aim of this project is to develop a standalone application, D-PorCCA, with a user-friendly interface and the necessary

functions and algorithms to study the behaviour of harbour porpoises in their natural environment from acoustic recordings.

To that end, the first objective was to develop a harbour porpoise click classifier (PorCC) based on machine learning techniques (*i.e.*, logistic regression). Identifying patterns in repetition rates require detecting as many clicks as possible. Therefore, the classifier works for clicks of both high and low quality (*e.g.*, clicks produce when the animal moves away from the recording device).

These clicks must be grouped and potential click trains produced by harbour porpoises identified. Therefore, the second objective was to develop a series of algorithms to identify these click trains, which are then cleaned from expected undesired sources of noise.

Within this topic, the third objective was to develop a series of algorithms to classify acoustic events into three main behavioural categories: 'Orientation' or inspection of the environment (or travelling), 'Foraging' or feeding, and 'Socialising'. Patterns that do not fit within known types would be classified as 'Unknown'.

Harbour porpoises interact with animals of the same species, but also with individual of other species. The fourth objective, therefore, was to study the acoustic behaviour of harbour porpoises during interactions with another species: a solitary female short-beaked common dolphin (*Delphinus delphins*) who has inhabited the Firth of Clyde (Scotland) since the early 2000s. Lastly, the fifth objective was to test D-PorCCA in recordings from another NBHF species: the Heaviside's dolphin (*Cephalorhynchus heavisidii*) from Namibia.

Due to time constrains, it was not within the objectives of this project to use D-PorCCA for behavioural studies, however, click train types and patterns found in the data are described in detail.

1.5 Contribution to knowledge

The major output of this project is a series of analytical tools that would allow for behavioural studies of harbour porpoises in their natural environment for the first time. In this context, the two most important contributions to knowledge are:

1.5.1 Harbour porpoise communication

Using the algorithms developed in this project a series of social calls were found in data from Danish and Scottish waters. Some of these calls had been described in the literature before, but others were not.

1.5.2 Interspecies communication

The interactions between a solitary short-beaked common dolphin (*Delphinus del-phis*) and harbour porpoises in the Firth of Clyde was known by locals and the scientific community. Acoustic data available during three interactions showed the dolphin changes her sounds to produce signals similar to those of the porpoises. Moreover, the sounds of both animals were produced in patterns that suggested vocal communication, including turn taking behaviour.

1.6 Dissemination of results

A series of scientific publications were published as a result of this project. This includes 1 peer-reviewed publication with the results of the first objective (Cosentino et al., 2019 [15]) and four poster presentations at international conferences. Additionally, two manuscripts are in preparation, onw with the results of the fourth objective and another one with the results described in Chapter 6.

Conference presentations

- Cosentino, M., Schwarzbach, P., Tougaard, J., Nairn, D., Guarato, F., Jackson, J.C., Windmill, J. F. C. (2019). D-PorCCA, a new tool to study the behaviour of harbour porpoises. Poster presentation. World Marine Mammal Conference. Barcelona, Spain.
- Cosentino, M., Guarato, F., Tougaard, J., Nairn, D., Jackson, J.C., Windmill, J. F. C. (2019). PorCC: A new high-accuracy click classifier to study harbour porpoises in the wild. Poster presentation. Danish Marine Mammal Symposium. Odense, Denmark.
- Schwarzbach1, P., Cosentino, M., Guarato, F., Nairn, D., Jackson, J.C., Dahne, M., Windmill, J. F. C. (2019). New detector / classifier to study harbour porpoise behaviour in the wild. Workshop paper. Presented at the annual SAMBAH meeting. Turku, Finland.
- Cosentino, M., Nairn, D., Jackson, J., Windmill, J. (2018). I beg your pardon? Vocal communication between a wild solitary short-beaked common dolphin and a harbour porpoise. Poster presentation. European Cetacean Society annual meeting. La Spezia, Italy.

Peer-reviewed publications

Cosentino, M., Tougaard, J., Guarato, F., Nairn, D., Jackson, J.C., Windmill, J. F. C. (2019). PorCC: A new high-accuracy click classifier to study harbour porpoises in the wild. Journal of the American Society of Acoustics. See Appendix A - Derived data supporting the findings of this study and the classification algorithm are available at the Pure Data Repository of the University of Strathclyde here.

1.6.1 In preparation

- Cosentino, F., Nairn, D., Jackson, J.C., Windmill, J. F. C. (2019). I beg you pardon? Vocal communication between a solitary common dolphin and harbour porpoises.
- Cosentino, F., Nairn, D., Jackson, J.C., Windmill, J. F. C. (2019). The social lives of harbour porpoises.

Public outreach

2018

In early 2018, as a result of sharing highlights of this PhD project on Twitter, I was invited to participate in a science radio show from Australia to talk about the PhD as well as my previous experience as a marine biologist. The interview is now available at the RRR Australian Radio Station website and the TuneIn site (Einstein A Go Go).

In the summer of 2018, the *BBC The One Show* crew came to Scotland to film Kylie, a solitary dolphin in the Firth of Clyde who is occasionally seen with harbour

porpoises, a topic of my project. The story was released in the One Show and the University of Strathclyde drafted a Press release concurrently, which attracted attention from all over the world. Kylie and preliminary results of this project made the front page of the BBC news (What did the dolphin say to the porpoise? and were featured in international popular sites such as "I fucking love science" (A Scottish Dolphin Has Learned How To "Speak" Porpoise) and "Forbes" (A Lonely Dolphin Has Learnt To Talk To Porpoises), as well as other sites in France and Germany.

2019

In 2019 I was featured in a documentary about the Hunterston building site, located close to where Kylie is usually seen (The Hunterston Proposal). Additionally, in April 2019 I was invited to participate in an event organised by the Royal Society for the Protection of Birds (RPBS). Although I was not able to attend I recorded the presentation, which was shown during the event at the University of Aberdeen (Scotland).

2020

Lastly, I gave a 15 minute presentation in the Argyll Community Centre in Saltcoats in January 2020 with some of the results of this thesis. The audience comprised elderly from the local community.

1.7 Thesis outline

This dissertation is divided into 9 chapters.

Chapter 1 presents a general introduction to the topic of this project and describes

the goal and main objectives, as well as the main results and outreach efforts.

Chapter 2 presents a review of the current state of knowledge of several aspects of harbour porpoise ecology and natural behaviour, as well as the available tools to study them in their natural environment. The chapter also discusses aspects of animal communication within and between species, and the challenges of carrying out research of acoustic behaviour of cetaceans.

The datasets used in this thesis are described in Chapter 3, as well as the methodologies used for data collection in both study areas, Scotland and Denmark.

Chapter 4 presents a detailed description of the methods used to develop a new click classifier for harbour porpoises (PorCC). It also presents the results applied to the Scottish database in comparison to the performance of an existing classifier.

Chapter 5 presents the results of using a series of algorithms developed to group and classify click trains, clean them from undesired sources of noise, and extract existing patterns to identify the underlying behaviour of the harbour porpoise. Examples of each click train type and behaviour are presented, as well as general results such as the total number of click train types and behaviours found in the data. The algorithms are attached in Appendix C). The algorithms developed in the previous chapters were used in harbour porpoise data from Scotland and Denmark and the results are presented in Chapter 6. This chapter is focused on the behavioural patterns. The results showed that each of the three categories defined here (*i.e.*, orientation, foraging, and socialising) comprised more than one pattern. These are described and discussed.

Chapter 7 presents a description of the long-term interactions between harbour porpoises and a solitary short-beaked common dolphin who inhabits the Firth of Clyde. This include a general discussion of the types of interactions observed over

the years, as well as the results of the acoustic interactions.

Chapter 8 introduces D-PorCCA, a stand-alone application that incorporates the algorithms I developed for this project, which allows researchers from anywhere in the world to analyse the data that has already been collected to learn more about their species, including all porpoise species and the additional seven species known to produce NBHF clicks. At the end of Chapter 8, the results of using D-PorCCA in acoustic recordings of Heaviside's dolphins, another species known to produce narrow-band high-frequency clicks, similar to those of the harbour porpoises.

Finally, Chapter 9 presents the general conclusions of this thesis and recommends future work.

Chapter 2

Literature Review

2.1 Introduction

This Chapter presents a review of the current state of knowledge of several aspects of harbour porpoise ecology and natural behaviour, as well as the available tools to study them.

2.2 Background

Humans have been fascinated by cetaceans for thousands of years, evidenced by the numerous paintings, sculptures, and myths that exist around them since prehistoric times. Despite this fascination, there are still many aspects of their biology, ecology, and natural behaviour that are poorly understood. Studying cetaceans at sea is a difficult task, they spend most of their time underwater and so our ability to see them depends on many factors, from weather conditions to observer experience

[16, 17]. In fact, even in good visibility conditions animals can remain visually undetected due to their behaviour. This is especially true for cryptic species, such as the harbour porpoise (*Phocoena phocoena*).



Figure 2.1: Sixteenth Century interpretation of a 1st Century description by Pliny the Elder of killer whales (orcha) attacking whales (balena) and their calves near Cadiz (Spain). Illustration by Olaus Magnus in his "Carta Marina" (1539).

Our knowledge of the lives of cetaceans in their natural environment has grown exponentially over the past few decades thanks to technological advances. It is now possible to collect large amounts of data in a cost-effective manner. At the same time, this requires long periods and many resources dedicated to data analysis, which often results in leaving data unanalysed. Therefore, tools that improve our ability to analyse data with minimum manual work will open doors for studies that were not possible before. When studying cetaceans, progress is inevitably linked to technological advancement.

2.2.1 Cetacean acoustics

Cetaceans evolved from land mammals that returned to the oceans, the hippopotamus being the closest relative alive today [18, 19, 20]. As they re-entered the water, cetaceans evolved a wide range of adaptations for the transition to a fully aquatic life. Their bodies became more hydrodynamic, losing the posterior limbs and evolving a smooth, hairless skin to reduce friction, and a thick blubber layer that allows them to maintain the body temperature around 36C, just like land mammals do. In addition, their nostrils displaced to an upwards position at the top of the head, facilitating breathing during travel, specially at high speeds [21, 22, 23].

Cetaceans are divided into two groups: odontocetes or toothed whales, with 73 recognised species, and mysticetes or baleen whales, with 14 recognised species. As a group, cetaceans produce a wide range of sounds, within and outside the human hearing range, from very low frequencies (~ 10 Hz), such as 'songs' produced by baleen whales, to impulsive sounds with high frequency content (> 130 kHz) produced by toothed whales [6, 24, 25].

Odontocetes rely on sounds both via active production and passive hearing, to find preys and avoid predators, as well as for orientation, and to keep contact with other individuals and finding potential mates. The following are the three main types of sounds Odontocetes produce [26, 27]:

1) narrow-band tonal whistles: whistles are tonal sounds of long duration, usually few seconds long, with frequencies concentrated between 5 and 20 kHz [23]. Whistles are used for individual identification and communication purposes [e.g., 28].

2) short-pulsed sounds: these are known as 'echolocation clicks'. Echolocation is 'the process in which an animal obtains an assessment of its environment by emitting sounds and listening to echoes as the sound waves reflect off different objects in the environment' [29].

3) burst-pulsed calls: these are clicks produced at high repetitions rates, where the inter-click interval is too small to perceive them as separate clicks, and include sounds like 'cries' and 'barks', which are thought to be used for communication purposes[27, 30].

The term echolocation was coined in the 1950s by Griffin [31] to describe the ultrasonic impulsive sounds used by bats to forage and navigate their environment, and was later adapted for cetaceans [for a detailed review on echolocation in bats and dolphins see 32]. The temporal and spectral characteristics of echolocation clicks vary between species [33, 34, 35] and likely reflects the ecological adaptation to the habitat and local prey species [32]. Echolocation clicks can be divided into four groups:

1) broadband, short-duration clicks, produced by most delphinids, such as the bottlenose dolphin (*Tursiops truncatus*) [36]

2) narrow-band, high-frequency (NBHF), long-duration clicks, produced by at least16 species of whales, dolphins, and porpoises, including the harbour porpoise [6],

3) frequency-modulated up-sweep echolocation clicks produced by beaked whales, such as True's beaked whales (*Mesoplodon mirus*) [37], and

4) multipulsed, high powered low-frequency clicks produced by sperm whales (*Physeter macrocephalus*) [38].

Examples of these clicks are shown in Figure 2.2.



Figure 2.2: Echolocation signals of four odontocete species: (a) bottlenose dolphin (Tursiops truncatus) - from Evans (1973) [39] (b) harbour porpoise(Phocoena phocoena) - from this study (c) True's beaked whale (Mesoplodon mirus) - from DeAngelis et al. (2018) [40] (d) sperm whale (Physeter macrocephalus) - from Mohl et al. (2000) [38]

2.2.1.1 Narrow-band high-frequency - NBHF - clicks

There are at least 16 species from five different taxonomic groups, usually referred to as narrow-band high-frequency (NBHF) species, who produce these clicks almost exclusively (Figure 2.3). These species are all seven *Phocoenids* [6, 41, 42], all four members of the genus *Cephalorhynchus* [*e.g.*, 43], two species of the genus *Lagenorhynchus* [44, 45, 46, 47], the pygmy (*Kogia breviceps*) [5] and dwarf sperm whales (*Kogia sima*) [48], and the franciscana (*Pontoporia blainvillei*) [49].

The temporal and spectral characteristics of these NBHF clicks are remarkably similar in all species where they have been described (Figure 2.4), even in species



Figure 2.3: NBHF species. Known species that produce narrow-band high-frequency (NBHF) clicks.

with a wider acoustic repertoire, such as the Heaviside's and the Commerson's dolphins [47, 50]. NBHF clicks have peak and centroid frequencies between 100 and 160 kHz, little or no spectral energy below 100 kHz, duration between 57 and 212 μ s, and -3 dB bandwidths between 8 and 20 kHz [5, 6, 7, 41, 42, 43, 47, 50, 51, 52]. These NBHF clicks are emitted in trains in a narrow, forward-oriented beam [53, 54]. A click train can be loosely defined as a 'series of clicks separated by gradually or cyclically changing inter-click interval suggesting a unit during an echolocation event or a communication signal' [8].

The production of NBHF clicks evolved independently in these five groups [55], and thus comprises species with different distribution, habitat preferences, and diets. What all these species have in common, apart from the characteristics of their clicks, is their small size (compared to other odontocetes species) and their social



Figure 2.4: Waveforms of narrow-band high-frequency (NBHF) clicks of four species (a) Heaviside's dolphin (Cephalorhynchus heavisidii) - from Morisaka et al. 2011 [51]). (b) Burmeister's porpoise (Phocoena spinipinnis) - from Reyes Reyes et al. 2018 [42] (c) Dwarf sperm whale (Kogia sima) - from Merkens et al. 2018 [48] (d) Pygmy sperm whale (Kogia breviceps) - from Madsen et al. 2005.

lives, more specifically, the seemingly lack of it. In this context, it would seem that the characteristics of the NBHF clicks make them unsuitable for communication purposes, as these are stereotyped clicks, which individually likely carry little information, and because high frequencies attenuate fast, limiting communication to few hundred meters at most. However, a series of studies on harbour porpoises and Hector's (*Cephalorhynchus hectori*) and Heaviside's dolphins have found patterns in the repetition rates of NBHF clicks in relation to specific behaviours (*e.g.*, aggression), suggesting this is how information is transmitted to conspecifics [*e.g.*, 9, 11, 12, 13, 43, 50]. A summary of the NBHF clicks characteristics of all species where they have been described is shown in Table 2.2.

Table 2.1	: NBHF specie	es and the ϵ	licks characteris	tics (mean	$i \pm SD$	
SPECIES	PF(kHz)	-3dB BW	Duration (μs)	Pattern	Other sounds	References
Harbour porpoise [*]	$137(\pm 6)$	$17(\pm 5)$	$54(\pm 8)$	Yes	No	[6, 7, 41]
$(Phocoena\ phocoena)$						
Dall's porpoise	$137(\pm 4)$	$11(\pm 5)$	$104(\pm 37)$	ż	No	[41]
$(Phocoenoides \ dalli)$						
Burmeister's porpoise	$138(\pm 11)$	$8(\pm 2)$	$144(\pm 64)$	ż	No	[42, 56]
$(Phocoena\ spinipinnis)$						
Vaquita	128 - 139	10-28	79-193	ż	No	[57]
$(Phocoena \ sinus)$						
Finless porpoise	$121(\pm 4)$	$17(\pm 3)$	$80(\pm 11)$	ż	N_{O}	58
$(Neophocaena\ phocaenoides)$						
Narrow-ridge finless porpoise	$125(\pm 7)$	$20(\pm 4)$	$68(\pm 14)$	ż	N_{O}	[59, 60]
$(Neophocaena\ asiae orientalis)$						
Spectacled porpoise ^{**}	ż	<i>~</i> ·	ż	ż	ż	
$(Phocoena\ dioptrica)$						
Heaviside's dolphins ^{***}	122 and 130			\mathbf{Yes}	Yes	[50, 51, 61, 62]
$(Cephalorhynchus\ heavisidii)$						
*The authors analysed clicks fro	om two popula	tions and t	he values varied	slightly be	etween them [41]	
**Nothing is known about the s	sounds of Spec	tacled porp	oises, although i	t is thoug	ht they would be	e similar
to those of other porpoises [63].						
*** Heaviside's dolphins produc	two different	t types of cl	icks, ones with]	peak arour	nd 122kHz anoth	ier at 130kHz

				Ę		ر ل
SPECIES	PF(KHZ)	-3dB BW	Duration (μs)	Pattern	Uther sounds	Keterences
Commerson's dolphins	$132(\pm 6)$	$21(\pm 3)$	$78(\pm 1)$	ż	Y_{es}	[64, 65]
$(Cephalorhynchus\ commersonii)$						
Chilean dolphins	$126(\pm 2)$	$18(\pm 5)$	$82(\pm 30)$	ż	N_{O}	[44]
$(Cephalorhynchus\ eutropia)$						
Hector's dolphins	$129(\pm 5)$	$20(\pm 3)$	$57(\pm 6)$	Yes	No	[43, 46]
$(Cephalorhynchus \ hectorii)$						
Hourglass dolphins	$126(\pm 2)$	$8(\pm 2)$	$115(\pm 24)$	Yes	N_{O}	[46]
$(Lagenorhynchus\ cruciger)$						
Peale's dolphins	$123(\pm 3)$	$15(\pm +4)$	$92(\pm 2)$	ż	N_{O}	[65]
$(Lage nor hynchus \ australis)$						
Dwarf sperm whale*	$127(\pm 1)$	$9(\pm 1)$	$199(\pm 54)$	ż	N_{O}	[48]
$(Kogia\ sima)$						
Pygmy sperm whale	$130(\pm 1)$	$8(\pm 2)$	$119(\pm 19)$	ż	N_{O}	[5, 66]
$(Kogia \ breviceps)$						
Franciscana	139	19	$212(\pm 56)$	ż	\mathbf{Yes}	[49, 67]
$(Pontoporia\ blainvillei)$						
*The authors analysed clicks from	a two popul	ations and t	the values varied	l slightly b	etween them [48	<u></u>

Two main hypotheses have been proposed about the evolutionary advantages of these clicks: reduced masking and acoustic crypsis [41, 50, 68]. Masking is the summation of ambient noise with the emitted signals, which reduces the probability of detection. In these species masking is reduced or avoided entirely because there are few natural sounds in that frequency range [27]. Acoustic crypsis is achieved as the effective hearing range of their main predator, the killer whale (*Orcinus orca*) [69, 70, 71], is below 100 kHz [72]. The latter might be partly incorrect, however. Galatius et al. (2019) showed that only *Lagenorhynchus* and *Cephalorhynchus* species evolved after the killer whales did, and therefore NBHF clicks in these species could have evolved in response to killer whale predation. For the other groups, NBHF could have evolved to avoid being heard by other predators that existed at the time, especially the 'killer sperm whale', an animal similar to modern sperm whales [55, 73].

Due to their cryptic behaviour, little is known about these species, some of which are rarely seen alive. An exception is the harbour porpoise, which has been studied thoroughly both in captive settings and in the wild, and even for them there is still many aspects of their natural behaviour that remain unknown.

2.3 The harbour porpoise

The harbour porpoise is one of the seven extant porpoise species and it is the most abundant and best-known species of all. They are dark grey on the dorsal area and lighter on the ventral side, and have a small, rounded head with no noticeable beak. The dorsal fin is small and triangular, positioned in the middle part of the body (Figure 2.5). Much is known about their life history, which was derived from data

collected from stranded animals and individuals incidentally caught in fishing nets [e.g., 74, 75]. Harbour porpoises are short-lived, compared to other Odontocetes, reaching usually between 10 and 12 years of age, with only about 7.5% of the population reach older ages, up to 20 years [74, 75]. The record is at 24 years for a male found stranded in the North Sea [74].



Figure 2.5: Harbour porpoises off Cumbrae, in the inner Firth of Clyde (Scotland) in the summer of 2018. They were part of a larger group of 20-25 individuals. Photo: Melania Cosentino.

Harbour porpoises are found exclusively in the northern hemisphere, in both sides of the Atlantic and Pacific Oceans, as well as in the Baltic and the Black Seas [76]. They are rarely seen in the Mediterranean Sea, although recent sightings suggest that a small population may be found off the coasts of Málaga, in southern Spain [77, 78]. There are four recognised subspecies, two in the North Pacific (P. p. vomerina and an unnamed one), one in the North Atlantic (P. p. phocoena), and one in the Black Sea (P. p. relicta) [76]. The Baltic Sea population is a genetically distinct population but does not qualify as a subspecies [76].

Although absolute abundance numbers throughout their entire distribution range are unknown, some harbour porpoise populations have been assessed. In the North Sea and adjacent waters abundance was estimated through large-scale surveys conducted in 1995, 2005, and 2016 [1, 79, 80, 81]. The most recent survey puts this

population in over 424,000 animals (CV = 0.17; 95% CI = 313,151 - 596,827) [81]. However, not all populations are so abundant. The Baltic Sea population is critically endangered, with fewer than 500 individuals remaining [82].

Because of their small size and cryptic behaviour, harbour porpoises are specially difficult to observe at sea (Figure 2.6). They surface for a few seconds and spend only about 5% of their time at the surface, depending on geographic location, sex of the individual, time of day, and season [4]. They are shy and rarely approach vessels, diving for about 4 min [83], and are seen travelling in groups of two or three animals, or alone [17, 79]. Although they are occasionally seen in large groups, these observations are rare [e.q., Nairn, unpublished data, 3], thus sightings are restricted to short distances and good weather conditions. Therefore, using typical visual monitoring methods, such as photo-identification and visual surveys from air, research vessels, or whale-watching platforms, to study the behaviour of harbour porpoises is limiting. In fact, few researchers have attempted to use photo-identification [e.g.,84, 85], as the shape and colouration of their dorsal fin are similar between individuals, and have low levels of scaring compared to other species. These scars are the basis for individual identification in many species [e.q., 86]. Maybe porpoises do not scar as easily as other species do or maybe they do not usually engage in playful or aggressive behaviour that leave scars, which are common behaviours in other species [87]. An additional reason for not so many scars in porpoises may be the very small blunt teeth of this species.

Several environmental factors influence harbour porpoise distribution and habitat use. The main are water depth (between 50 and 200 m) and slope [e.g., 79, 88, 89, 90], which are in turn related to the distribution of their prey [91, 92]. Individual move-



Figure 2.6: Harbour porpoise in the distance. Firth of Clyde, Scotland. Photo: Melania Cosentino

ments vary greatly, however, as some porpoises travel dozens of km a day moving to far locations, while others stay in the same area for weeks [88, 91]. When looking at entire populations (*e.g.*, the North Sea), variations in sighting rates have been observed, within and between years [2, 17, 75, 79]. Evidence of coordinated migrations has been found in historic catch data of harbour porpoises in the Baltic Sea [*e.g.*, 93], and more recently in studies using passive acoustic monitoring [82]. Similarly, migration patterns have been identified in the population off West Greenland. Several individuals carrying satellite tags provided the first record of seasonal migration of harbour porpoises to offshore, deep waters. Some individuals moved hundreds of km to open waters, reaching deep areas in the middle of the North Atlantic Ocean, and performing dives of up to 410 m before returning to West Greenland [94]. These results were unexpected given what was known about porpoises. Another example of the importance of technological advances in our understanding of cryptic species.

2.3.1 Acoustics

The first recordings of harbour porpoises were made in the early 1960s using tape recorders. These recordings suggested harbour porpoises produce narrow band clicks with peak frequencies centred at 2 kHz and were emitted at different repetition rates, which seemed to be related to different behaviours, such as feeding and communication [10]. The high-frequency (HF) component described in subsubsection 2.2.1.1 were recorded and described a few years later [6]. A low-frequency (LF) component has been reported in other NBHF species, such as the Heaviside's dolphin [61] and the Yangtze finless porpoise (*Neophocoena phocoenoides asiaeorientalis*) [60]. There have been discussions within the scientific community about whether the LF component is real or just an artefact of the recording device, and if it is real, whether it has enough energy to be important for communication purposes [*e.g.*, 7], as low frequencies travel farther than high frequencies [95].

The discussion has settled now. The click production process delivers a click with both components [11, 96], but the LF component is between 50 and 60 dB weaker than the HF component [7] and thus unlikely to have any significant value for communication. Interestingly, however, the LF component is stronger than the HF one during the first 4 days after birth [96].



Figure 2.7: A section of a click train from a 1 year-old male, showing the waveform of the low frequency (LF) and high frequency (HF) components. The two components were recorded with different hydrophones, on two separate tracks of the tape recorder. The LF component track has been low-pass filtered at 20 kHz, and the HF component has been high-pass filtered at 10 kHz. From Amundin 1991, p67 [11]

2.3.1.1 Sound production

Unlike humans and other animals, toothed whales lack vocal cords in their larynx, and their respiratory system (in principle) is not connected to their digestive system (but see [97] and [98]), which means they have a different sound production mechanism. Toothed whales have one nasal opening located at the top of their heads, known as the 'blowhole'. A few centimetres below the blowhole, there is a series of structures, including two nasal passages each with a pair of 'phonic lips' in association with fat bodies. Sound is produced by air passing through the phonic lips, which are put in motion when the nasal passages are pressurised. The sound is then



guided via the air sacs and the anterior side of the skull into the melon Figure 2.8.

Figure 2.8: Transverse MRI scans of a young harbour porpoise. The caudal part of the melon (yellow) abuts against layers of connective tissue, muscles, and tendons (red) forming a dense theca, which, along with the skull and a collection of nasal air sacs, reflect the vibrations that originate in the phonic lips (light brown) into the melon. The melon is under control of highly developed facial musculature. Adapted from Wisniewska et al. 2015 [99]

The melon is a fatty tissue that acts as a lens and an impedance matcher, which creates a highly directional sound beam that is emitted into the water [11, 100]. The width and height of the sound beam is different between species but, in the case of click production, it is consistently narrow. In harbour porpoises, the -3dB beamwidth in both the horizontal and vertical planes is approximately 16 degrees [53]. The peak frequency is concentrated between 100 and 160 kHz (*i.e.*, wavelength approximately 1-1.5 cm), centred at 130 kHz [6]. Mean peak frequency values can vary between populations [*e.g.*, 41].

Because Odontocetes have two pairs of phonic lips (except for the three sperm whale species, which have one) it was proposed that sound is produced simultaneously by both phonic lips. However, a series of experiments with suction cups hydrophones and video cameras showed phonic lips and associated structures act independently during sound production. Moreover, each pair seems to be specialised in producing one type of sound, either tonal or pulsed sounds, which means that dolphins can whistle and click at the same time. Whistle production is associated with the movement of the left nasal passage (although the right passage may be able to produce whistles as well) while click production is associated with movement of the right nasal passage [52, 101, 102]. This sound production mechanism has been found in all species studied so far. In harbour porpoises (and likely other non-whistling species), clicks are produced exclusively by the right pair of phonic lips [52, 102].

2.3.2 Behaviour

Many aspects of the behaviour of harbour porpoises have been studied, although not all of them are well understood. Hall (2011) presented an exhaustive review of harbour porpoise behaviour on her PhD dissertation, which she divided into 12 categories, based on 9 categories proposed by Scott in his book Animal Behaviour (1958, cited by [3]). These categories are:

1) ingestive: foraging and feeding [e.g., 9],

 shelter-seeking: habitat selection, including side fidelity and migration [e.g., 88, 94],

3) sexual: mating behaviour and as a form of play (e.g., between males), and spatial segregation of the sexes [e.g., 87, 103, 104],
4) agonistic: conflict or aggressive behaviour [e.g., 11, 12],

5) epimeletic: caring for another individual that is injured or need helps in some way,

6) et-epimeletic: the calling or signalling for care and attention [e.g., 8],

7) allelomimetic: behaviour leading other individuals to copy one's behaviour, such as synchronised surfacing during feeding or resting [e.q., 3],

8) eliminative: excretion of waste,

9) investigative: examine their surroundings or specific objects in their environment [e.g., 105],

10) avoidance: moving away from conspecifics, potential predators, or anthropogenic disturbances (*e.g.*, noise [106]),

11) social: engaging in social behaviour with conspecifics [e.g., 10, 11], and

12) rest and sleep [e.g., 107].

Social behaviour in Hall (2011) [3] refers specifically to socialising with individuals of the same species, including gathering in large groups and showing playful behaviour, such as surfing the waves. However, and although Hall mentions approaching vessels and discusses the existence of hybrids between harbour and Dall's porpoises, she does not mention social interactions with other species, which could be considered another type of behaviour (see subsection 2.3.5).

Our current understanding of harbour porpoise vocal behaviour suggests they have specific vocalisation pattern for some of the behaviours mentioned above. Because this project is focused on identifying behaviours based on patterns of click repetition rates, behaviours are re-organised into three broad categories: orientation or travelling (*i.e.*, 2 and 9), foraging or feeding (*i.e.*, 1), and social behaviour or communication (*i.e.*, 3, 4, 6, and 11). The behaviours not included in these categories

(i.e., 5, 7, 8, 10, and 12) either lack information about the corresponding acoustic emissions or the evidence is too scarce to be included (e.g., sleep).



Figure 2.9: **Behaviours**. Examples of representative behaviours based on repetition rate patterns: orientation or travelling (left), foraging or feeding (middle), and socialising or communication (right)

Technological advances had made it possible to study aspects of their lives that take place underwater that would not be possible with observations alone, from diving [94] and feeding abilities [108] to acoustic communication [13]. Although nearly 200 harbour porpoises have been tagged over the years in different areas of the North Atlantic [87], this represents a small number compared to population sizes. Although tags are relatively inexpensive and less invasive than they used to be, and can stay on the animals for long periods, it is not possible to tag every porpoise. On the other hand, tag data has provided information that can be used for behavioural studies using static acoustic monitoring (SAM) systems, which are cost-effective and non-invasive methods.

2.3.2.1 Orientation

Toothed whales use echolocation clicks to investigate or orient themselves in the environment they move through, and harbour porpoises are not an exception. Harbour porpoises vocalise almost continuously [13, 109], with 90% of silent intervals being no longer than 10 seconds [109], except during certain periods when they seem to be sleeping [107]. When they are travelling or navigating, the rate of click production is usually between 20 and 50 clicks per second, and not exceeding 100 clicks per second. During orientation or travelling, the relationship between the inter-click interval (ICI) is directly proportional to the distance to the target. In other words, the interval between clicks is long enough to allow for the echo to return and be processed before another click is produced, which is known as the two way travel time (TWTT). When the animal approaches a target object (*e.g.*, a rock), the ICI decreases because the signal and its echo travel smaller distances, thus arriving faster to the porpoise.

Verfuss et al. (2005) showed with a series of experiments with porpoises in captivity, that they use echolocation to orient themselves in their environment, even when they had been there for several years. The study found that porpoises echolocated continuously and that they used objects in their environment, which the authors called 'landmarks', to find their way to the target location. This behaviour is evident when looking at the variations of the ICI as the animal approaches the target (Figure 2.10). The ICI changed from ~60 ms to ~20 ms with decreasing distance. The ICI is always higher than the TWTT, which includes a 'lag time' necessary for processing the incoming information [110]. The lag time estimated in these experiments was between 26 and 36 ms when faced with difficult tasks and between 14 and 19 ms for simple ones [105].



Figure 2.10: Landmark. The median click interval (75% and 25% quartile) for each 1-metre bin distance to reference for each task. The two-way-transit time line of the click–echo pair is indicated (from Verfuss et al. (2005) [105]).

Although the lag time appears to be a constant value, as if the animals 'lock' their clicking rate to maintain this lag time, they have control over the repetition rate and the intensity of the clicks, which they adjust depending on environmental conditions (*e.g.*, noisy background) [111].

This information can be used to estimate the active space of the animal, that is, the distance up to which porpoises can make up their environment. Although the actual distance will depend on the source level of the signals and the background noise level, we can make a rough estimate using this equation:

Distance = ((TWTT/2) + lag time) * speed of sound in water

So, for example, with an ICI of 40 ms, and assuming a lag time of 20 ms, porpoise are 'seeing' up to a distance of ~45 m [109]. This is valid during orientation and exploration of the environment, but it cannot be applied to all cases, as porpoises are able to produce well over 1000 clicks per second (*i.e.*, ICI below 1 ms) [10, 11, 12, 13]. It is not yet understood how porpoises process information when clicking at high repetition rates.

2.3.2.2 Foraging

Data from stomach contents of stranded and bycaught animals suggest that harbour porpoises feed mainly on small demersal and pelagic fish, especially sand eel and cod, and whiting and herring, respectively. However, they can feed on many other fish, crustaceans, and cephalopods species. In addition, the importance of each prey type varies depending on season, area, and other factors [104, 112, 113]. Our understanding of what feeding looks like acoustically comes from experiments in captivity and data from wild animals carrying acoustic and behavioural tags. These data suggests porpoises use echolocation clicks to find and capture their prey in what can be thought of as a foraging event.

A successful foraging event can be separated into three phases: search, approach, and feeding buzz. These stages are common to all echolocating mammals, including toothed whales and bats [32]. During the search phase, the porpoise produces fewer than 100 clicks per second (*i.e.*, ICI > 10 ms) in search of a potential prey. When the prey is located, the porpoise approaches it and the ICI becomes smaller and smaller, as the echo reaches the porpoise faster and faster. The final stage, the feeding buzz, is a period of high repetition rates (*i.e.*, the ICI can be <2 ms) [*e.g.*, 9, 83, 109]. During these buzzes, the amplitude of the clicks decreases, becoming 6 to 10 dB less intense than typical echolocation clicks [9].

2.3.2.3 Social behaviour and communication

The social lives of harbour porpoises in their natural environment is a mystery. When using visual methods, harbour porpoises are usually observed travelling alone or in small groups of two or three individuals [1, 17, 79, 114]. Large congregations do occur and have been reported in the literature [e.g., 3, 87] and in social media, but remain rare. This seemingly solitary life might be just the result of how a cetacean 'group' is understood and defined during surveys. Most studies do not provide a definition of what a group is [e.g., 115], and when they do, a group is usually defined as a sample of individuals that interact or showed coordinated behaviour or both [e.g., 47]. Interaction refers to interactions seen by the observer, which hinders our understanding of social behaviour in cryptic species. Harbour porpoises spend 95% of their time underwater and although their phonations can travel a few hundred meters, animals can be interacting acoustically at large distances that the observer would not perceive. This is recognised by researchers, but it is rarely mentioned. Given group size numbers reported in most studies, 'group' is usually defined as it is understood for other species [e.q., 76, 79, 81, 115]. There are exceptions, however [85, 116].

In fact, data from wild animals carrying acoustic tags have shown that harbour porpoises are more social than visual surveys tell us, spending much of their time at hearing distance of conspecifics [e.g., 13]. The longest period recorded without sounds from other porpoises was ~ 3 hours. Furthermore, the results were consistent with previous studies, showing that harbour porpoises produce two main types of click trains that are qualitatively different: foraging buzzes and social or communication calls. The distribution of the ICI of calls is bimodal, with peaks at ~ 125 and ~ 1000 clicks per second, with little overlap with feeding buzzes, which have the peak around 400 clicks per second [13]. The key difference between feeding buzzes and social calls, as a general rule, is that feeding buzzes show a gradual increase of repetition rates, while calls lack a preceding period of low repetition rates [13] [but see 12].

The use of the term 'communication' calls may prove problematic because, even though the field of animal communication has been active for many decades, there is not yet a definition of communication everyone agrees with. It is often defined as the process of conveying information from senders to receivers by means of signals, where signals are behaviours or structures that senders evolved in order to convey information [117]. This way of thinking about communication is not restricted to acoustic signals. Most animals, as well as plants, and cells, communicate using chemical, visual, and tactile signals, as well as acoustic ones [118, 119, 120, 121, 122]. For some authors, however, not all acoustic signals constitute a form of communication. For example, Au and Hastings (2009), in their book Principles of Marine *Bioacoustics* when talking about sounds produced by toothed whales, state that 'Social sounds are those that are used by odontocetes in a social context and could be, but not necessarily, used for communications' [123]. Given virtually nothing is known about the social lives of harbour porpoises in their natural environment, 'communication' calls here refer to burst pulses, which lack a preceding period at low clicking rates. Whether these calls carry information or if they are used for communication, is difficult to assert given the few existing studies on this topic [8, 10, 11, 12, 13].

Because the low-frequency (LF) component is produced at the same time as the high-frequency (HF) component, it is possible to use observations of early studies based on the LF component to learn about harbour porpoises calls. Busnel and Dziedzic (1966) and Amundin (1991) described a series of calls that seemed to occur in specific contexts, including courtship, dominance, and distress. The repetition rates of these calls were different: while 'courtship' calls were long (\sim 1 s) and had high repetition rates (>500 clicks per second), 'dominance' calls looked like two or three consecutive short calls at an interval of 200 ms, with variable repetition rates of up to 1000 clicks per second [10]. 'Distress calls' were described in different studies, but the characteristics of the calls differed. Busnel and Dziedzic (1966) described these as calls with repetition rates between 130 and 250 clicks per second and duration between 0.4 and 1.55 s [10], while Amundin (1991) described them based on the HF component as calls with repetition rates between 100 and 500 clicks per second, but with repetition rates between 300 and 400 clicks per second based on the LF component [11].

Other calls include 'push threat call' (400-900 clicks per second), 'signals of pain' (500-800 clicks per second), 'contact call' between mother and calf (increasing from 10 to over 600 clicks per second - Figure 2.11), and 'aggression' (200-1000 clicks per second) [9, 10, 11, 12]. Additionally, Amundin found what he called 'phrases', comprising two or more (in general similar) calls repeated in a short period of time. He also found what he believed to be information that could be used for individual recognition, specifically variations in repetition rates at the end of the communication calls [11].

Clausen et al. (2010) found the duration of the click trains also varied markedly depending on the behaviour, for example contact calls are less than 1 s long, while



Figure 2.11: Contact call. Click repetition rate pattern for contact calls produced during mother-calf interactions (extracted from Clausen et al. (2010) [12]).

grooming can last up to 16 s. Moreover, the duration of the click itself varied slightly between behaviours [12].

Communication calls have been studied in wild porpoises for the first time in 2018, in a study based on six porpoises carrying acoustic tags [13]. Calls from studies using static acoustic monitoring devices, however, are rare. One example is Koschinski et al (2008). The authors witnessed a calf that was entangled in a gillnet in Clayoquot Sound (Canada) and were able to record it during this period. They found that 92% of the echolocation events had ICIs below 10 ms, suggesting these were communication calls. The patterns resembled those of 'distress calls' described by Amundin (1991), although the repetition rates were different [8, 11]. The authors also found calls with repetition rates at ~160 clicks per second, which falls within the bimodal distribution found by Sorensen et al. [8, 13].

At the time of writing, no study using static or towed continuous recorder have reported communication calls in wild harbour porpoises.

2.3.3 Calves

Harbour porpoises are small, reaching a maximum of about 2 m in length and 72 kg in weight, and although there are variations between populations, females are larger and heavier than males. Sexual maturity is reached at about 5 years of age, depending on sex, and varies between populations. Pregnancy rates are high, with 60-75% of mature females being pregnant at any given time of the year. Females become pregnant every year or every second year and gestation lasts between 10 and 11 months. Newborns are less than 80 cm long and 7 kg at birth. In the North Sea and adjacent areas, birth occurs between May and August, and weaning between February and May, when the porpoises are between 8 and 11 months old [63, 74, 75, 124, 125].



Figure 2.12: Harbour porpoise calf (and mother) - born in August 2007 in Fjord and Belt facilities in Kerterminde (Denmark).

Little is known about the ontogeny of harbour porpoise clicks. The only study focused specifically on the vocal behaviour of harbour porpoise calves was carried out with data from three individuals born in the Fjord and Belt facilities in Kerteminde (Denmark) [96]. The calves were born on August 2007, July 2013, and July 2014 respectively, but only the first one survived long enough (*i.e.*, 9 months) to study how the temporal and spectral characteristics of the clicks changed over time [96]. Delgado found that calves can vocalise just seconds after being born, as the first clicks reliably coming from the calf were recorded 40 seconds after birth. For the first 3 or 4 days these clicks had a high (HF) and a low frequency (LF) component, with the latter being stronger. The HF component is similar to the clicks of adult harbour porpoises, with a peak between 100 and 150 kHz, while the LF component has its main peak centred at 2 kHz [96]. For the first 9 months, the clicks (Figure 2.15) have higher peak frequencies than those of adults, decreasing from nearly 150 kHz to 130 kHz over this period, having as well higher bandwidths [96].

The higher peak frequencies in young individuals was also reported by Goodson et al. [126], who described the spectral and temporal characteristics of two captive porpoises in Harderwijk Aquarium (The Netherlands), one of which was 8 months old at the time and the other 20 months old. The authors found that the younger individual had a higher -3dB bandwidth and peak frequency values (14 kHz and 148 kHz, respectively) than the older one (12.5 kHz and 144 kHz, respectively) [126]. Similarly, Clausen et al. reported centroid frequencies at 136 kHz and -3dB bandwidth at 16 kHz for a calf of age between 3 to 9 months [12].

Both the LF and HF components are part of the same click production process being produced simultaneously [11, 60, 96]. How the LF component is produced and whether it has a function during this crucial period (*e.g.*, for the calf to keep contact with the mother) is still unknown, but it seems to be present in other neonatal porpoise species as well, such as the Yangtze finless porpoises (*Neophocaena phocaenoides asiaeorientalis*) [60].

Because the LF component is long in duration, it cannot be detected by existing



Figure 2.13: Click train produced by a harbour porpoise calf - from Delgado 2016 [96]

impulsive sounds detectors. Additionally, some researchers filter the data collected with broadband hydrophones before digitalisation, leaving out sounds with frequencies lower than 2 kHz. Moreover, much of the background noise is in low frequencies, thus a classifier for calves focused on the LF component would be ineffective. So far, no classifier exist to find calves based on acoustic information. Such classifier, would be crucial for conservation purposes as it would allow for identification of nursing areas that need protection.

2.3.4 Conservation status

The harbour porpoise, as a species, is listed as Least Concern by the International Union for Conservation of Nature (IUCN) [1], however, the European population is listed as 'Vulnerable' and the Baltic Sea population is 'Critically Endangered', with fewer than 500 individuals remaining [82]. In European waters, concerns for its conservation status due to, especially, lethal incidental catches in fishing nets, led to the Agreement on the Conservation of Small Cetaceans of the Baltic and North Seas (ASCOBANS). The agreement was signed in 1992 and the area was extended in 2008. ASCOBANS parties are obliged to develop and adopt measures to protect harbour porpoises, as well as to acquire knowledge about their occurrence, abundance, and distribution. The harbour porpoise is also listed in the Annex II of the Habitats Directive (Council Directive 92/43/EC) [14], which requires member states to set up Special Areas for Conservation (SACs) for them.

The survival of harbour porpoises is threatened by human activities, especially fishing and underwater noise. Thousands of harbour porpoises get accidentally caught in fishing nets and die every year in the North Sea and adjacent waters, especially in gillnets [e.g., 127, 128, 129]. Additionally, noise-producing activities, such as Navy operations, pile driving in windfarm constructions, and seismic surveys, can indirectly lead to death, for example by distracting the animals who swim towards fishing nets, but also by forcing them away from productive areas, resulting in starvation [106, 130, 131].

The designation of SACs requires identifying habitats that porpoises use for important life functions such as breeding (*i.e.*, identifying areas with mother-calf pairs) to minimise overlap with human activities. Given it is not possible to equip every animal with a tag, we can take advantage of recent technological developments and develop tools to use passive acoustic monitoring systems to fill gaps in our understanding of harbour porpoise behaviour and communication in the wild.

2.3.5 Interspecific interactions

Non-predatory interspecies interactions between cetaceans have been widely documented, of both aggressive and non-aggressive nature. Non-aggressive interactions usually consist of temporary feeding aggregations of two or more toothed (Odontocetes) or baleen whales (Mysticetes) species [132, 133, 134, 135], occasionally including other marine mammals and sea birds [136]. Social heterospecific interactions are infrequent compared to other non-predatory interactions, although they have been observed between groups of different species [137, 138, 139], between a group of one species and an individual of another species [140], and between two individuals of different species [141, 142]. These interactions are also usually temporary and involve either toothed or baleen whales or both [141, 142, 143].

Differences in body size do not seem to present a constraint for these interactions: in 2004 and again in 2006, a humpback whale (*Megaptera novaeangliae*) was seen what seemed like playing with a bottlenose dolphin off the coasts of Hawaii [141]. In both cases the dolphin was 'lifted' by the whale on top of its mouth and the interaction lasted just a few minutes. Interestingly, the observations were made two years apart (to the day) off different islands and neither the whale nor the dolphin were the same individual [141].

Interactions between a group of one species and an individual of another species are also usually temporary and have been reported for odontocetes as well as mysticetes. Species that in some areas are prey-predator or engage in interactions of aggressive nature can occasionally engage in social interactions [69, 143]. One of these unusual observations was made by myself in 2008, when a group of killer whales was seen socialising with a striped dolphin (*Stenella coeruleoalba*) in the Strait of Gibraltar, Spain.

The only known case of an individual of one species interacting having with a group of another species for a long period is Billie. She is an adult female bottlenose dolphin who lives in the Algeciras Bay, in the Strait of Gibraltar, with short-beaked common dolphins (*Delphinus delphis*) [144]. Despite bottlenose dolphins being resident in the area [145], she seems to spend of all her time with the common dolphins and was also seen caring for common dolphin calves and new-borns, which is a rarely observed type of interspecific interaction. Billie is thought to have recently give birth to a hybrid calf, although no genetic analysis have been done for confirmation yet [144].

2.3.5.1 Predatory interactions

The main natural predator of harbour porpoises is the killer whale [69, 70, 71], a cosmopolitan species and the most widely distributed marine mammal [146]. Although their distributions overlap, not all killer whale populations feed on marine mammals. In the North Pacific as well as in the North Atlantic, only one of the three populations feed on marine mammals [e.g., 147, 148, 149]. Further, killer whales predating on harbour porpoises in the North Sea and adjacent waters are scarce [79, 81, 150, 151].

More recently, a new predator has been identified in the North Sea: the grey seal (*Halichoerus grypus*). Grey seals feeding on harbour porpoises have been reported for many European countries, including Belgium, France, the Netherlands, and the UK. These observations include reports of scavenging as well as active predation [152, 153, 154, 155]. Despite grey seals being present in West Scotland and Danish waters [156], no predation events have been reported in these areas so far. The impact of grey seals attacks on the population of harbour porpoises in the North Sea and surrounding areas is unknown.

2.3.5.2 Aggressive interactions

For reasons that are not well understood, harbour porpoises are killed by bottlenose dolphins in some areas around the UK [143] and in Californian waters [157]. In the UK, deaths due to bottlenose dolphin attacks are recorded along the entire Scottish coast and constitute the most common cause of death for the species, representing about a quarter of all cases [158]. The impact on the population is unknown.

Lethal interactions with other species have also been recorded. Baird et al. (1998) described an event that took place in the west coast of Canada, off British Columbia in 1994. A pair of Pacific white-sided dolphins (*Lagenorhynchus obliquedens*) were seen interacting with a neonate harbour porpoise for several hours. Although the dolphins were not visibly aggressive, they harassed the porpoise, which eventually died [159].

Another example of a harbour porpoise calf who died as a result of interacting with dolphins in Canadian waters is described in Larrat et al. (2012) [160]. In this case there were no observations of the event, as the report is based on rake marks found on the body of a young calf (< 1 year old) found stranded off Quebec. The large number of markings on its body and their characteristics indicated the calf was a victim of aggressive behaviour from Atlantic white-side dolphins (*Lagenorhynchus acutus*) [160].

2.3.5.3 Affiliative interactions

Although rare, non-aggressive interactions between harbour porpoises and individuals or groups of other species have been reported throughout their distribution range. Recently, a juvenile harbour porpoise was seen travelling with a group of bottlenose dolphins in the Marmara Sea, spending much of its time in the echelon position [140]. This interaction is surprising as in other areas, as mentioned above, bottlenose dolphins are responsible for dozen of harbour porpoise deaths every year [143].

On the west coasts of the United States and Canada, harbour porpoises are sympatric with Dall's porpoises (*Phocoenoides dalli*), although there seems to be a temporal separation in their distribution. They have never been seen in mixed groups [3, 161], and yet hybrids of these two species have been confirmed in the area, which seem to occur at relatively high levels [162]. Willis et al. (2004) reported 20 cases of hybrid individuals within Dall's porpoise groups, of which nine were confirmed via genetic analyses. In all cases where tests were carried out (n = 5), the mother was a Dall's porpoise and the father a harbour porpoise, which explains why hybrids are observed in Dall's but not in harbour porpoise groups [162]. Unfortunately, no stranding of hybrids have been detected so far, which would provide opportunities to learn more about these individuals [3].

The only known case of 'ongoing' interspecific interaction between an individual of another species and a harbour porpoise is Kylie, a solitary female short-beaked common dolphin (*Delphinus delphis*) who lives in the Firth of Clyde [142]. Kylie inhabits the Firth of Clyde since at least the early 2000s and because she moves between different areas, she has received several names over the years, including Kylie, Donna, and Colin. Her sex was unknown until 2019, when was confirmed via underwater images taken by a local diver. She has lived a solitary life since she was first seen in the area, although she is known to be occasionally accompanied by a harbour porpoise [142]. Ryan et al. (2017) reported two observations of Kylie and a harbour porpoise, made in two different areas of the Firth of Clyde, four years

apart. The authors were lucky enough to have access to underwater photographs of the porpoise, which showed that it was the same individual in both cases [142]. Opportunistic observations by many locals suggest Kylie spends time with a harbour porpoise regularly (Nairn, Unpublished data). To date, her vocal behaviour has not been studied, neither when she is alone nor when she is with a harbour porpoise. The interactions between Kylie and harbour porpoises provide a unique opportunity to study inter-specific vocal communication in wild cetaceans.

2.3.5.4 Short-beaked common dolphins

Short-beaked common dolphins are found in temperate and warm waters around the globe, except the Indian Ocean. Around the UK, common dolphins are mostly found off Wales in the Celtic Sea and off the Northwest coast of Scotland [79, 80, 81], rarely close to shore, but are also known to visit the Moray Firth [163] in the east and the Minch in Northwest Scotland where they are seen in coastal waters [164]. This area is over 400 km away from where Kylie now spends most of her days, and where common dolphins are rarely seen (Nairn, unpublished data). Common dolphins live and travel in large groups that regularly exceed 50 individuals and can reach thousands of animals [165].

Common dolphins produce a wide range of sounds, within and outside the human hearing range, that they use for foraging, orientation, and communication purposes. Communication calls include whistles, buzzes, barks, yelps, squeals, and burst pulsed calls. Studies on common dolphin sounds have been mainly focused on whistles and how these vary depending on behavioural context and time of day, as well as geographic location [165, 166].

So far, there are no studies focused specifically on common dolphin clicks. Gurevich

(1969) (cited in Evans 1973 [39, 167]) reported that common dolphin clicks have energy between 100 and 150 kHz but, unfortunately, the study was in Russian and details, such as peak frequency values, were not reported by Evans (1973) [39, 167]. Fish and Turl (1976) then reported that Evans said (in a personal communication) that the source level 'and most of the other data for *Delphinus* and *Globicephala* reported in Evans (1973) could have been an editorial error [168]. Although it is not clear which of the numbers they refer to, other non-NBHF delphinids species are known to produce short-duration clicks with energy in these frequencies. Unlike NBHF clicks, short-duration clicks are broadband and thus cover a wide range of frequencies, sometimes extending over 100 kHz [123]. With increased source levels and as the recording devices gets closer to the centre of the echolocation beam, the centroid frequency of these clicks increase [*e.g.*, 102, 169].

Since then, several studies have directly or indirectly investigated the temporal and spectral characteristics of common dolphin echolocation clicks. In these cases, the authors either used the studies discussed above as references or used methods in-adequate for detecting high-frequency (> 100 kHz) sounds, recording at sampling frequencies between 80 kHz and 200kHz, rendering a maximum detectable frequency of 40 kHz and 100 kHz, respectively [168, 170, 171]. Because of the reported high levels of energy in frequencies below 100 kHz, these clicks are fundamentally different from NBHF clicks. Further, common dolphin clicks have been described as oligocyclic (*i.e.*, waveform with few peaks) and of short duration, not exceeding 50μ s.

The fact that information about the energy content is inconsistent between studies likely reflects the species ability to change the spectral characteristics of the clicks. Roch et al. showed that long-beaked common dolphin (*Delphinus capensis*) can

produce clicks with a wide range of peak frequency, from about 20 kHz to 65 kHz [172], as shown in Figure 2.14. It is expected that short-beaked common dolphins have similar abilities, especially as new data suggests short-beaked and long-beaked common dolphins may be the same species [20].



Figure 2.14: Long-beaked common dolphin. Concatenated spectrogram of longbeaked common dolphins (Delphinus capensis) clicks. Clicks were sorted by peak frequency which are highlighted by black points - Extracted from Roch et al. (2011) [172]

The characteristics of Kylie's clicks are not known or whether the sounds she produces differ to those of conspecifics who live in groups. There are a few records of toothed whales leading a solitary (temporarily or permanently), most of which are bottlenose dolphins [173, 174]. Kylie is one of the seven cases of solitary common dolphins ever reported worldwide [173, 175]. In most cases it is unknown why or how these highly social animals become solitary [173, 176, 177, 178].

2.3.6 Vocal learning and interspecies communication

The ability to learn new behaviours as a result of interacting with other individuals is known as social learning, whether it occurs by interacting with individuals of the same or other species [179]. When it comes to vocal behaviour, learning can be divided between contextual and production learning. Contextual learning occurs when the use (or understanding) of a signal is associated with a novel context, and production learning when an individual modifies the acoustic characteristics of an existing signal producing a signal that was not previously part of its repertoire [180, 181].

Janik and Slater argue that these vocal modifications refer specifically to the frequencies of the signal and result from exposure to acoustic inputs, either artificial sounds or by interacting with individuals of the same or other species [181]. Thus, cases in which animals change the amplitude or duration of the signal in response to high levels of background noise (*i.e.*, the Lombard effect) would not constitute production learning.

In captive settings it is possible to design experiments to test the animal's ability for vocal learning, as Richards et al. [182] put it: 'To establish a capability for vocal mimicry [...] the following two conditions are necessary and sufficient: (a) The vocalization produced by the animal in response to a presented "model" sound must resemble that model, and (b) the model must not resemble sounds present in the established baseline pretraining repertoire of the animal'. In that study, bottlenose dolphins were successfully trained to imitate artificial sounds and to use these sounds to label objects, such as 'person' and 'pipe' [182]. Killer whales have also been trained to imitate different sounds, including human speech in the form of simple words such as 'hello' and 'bye bye' [183]. Even more impressive results were recently seen in grey seals (*Halichoerus grypus*). A group of researchers were able to teach wild seals kept temporarily in captive setting to produce a wide range of sounds and imitate melodies of songs they could have never heard in the wild [184]. The ability to imitate human speech has been observed as well in marine mammals who did not receive any training. A trainer at the Vancouver Aquarium in Canada heard someone telling him to get out of the pool. To his surprise, it was a beluga (*Delphinapterus leucas*) in the pool who utter the words [185]. One of the most wellknown cases is Hoover, a harbour seal (*Phoca vitulina*) who spontaneously started to produce sounds that resembled human speech, with a Boston accent [186].

Inter-specific interactions sometimes lead to changes in the acoustic behaviour of the animals involved. A study conducted with captive animals found that killer whales who spent time with bottlenose dolphins, had learned to produce sounds that were similar to bottlenose dolphins' sounds, in what is, presumably, an attempt to communicate with them [187]. Similarly, a beluga who spent time with bottlenose dolphins were recorded producing whistles similar to those of bottlenose dolphins sounds within months of being introduced in the same environment. Moreover, the beluga favoured those sounds over the ones she used before the interactions occurred [188].

Production learning as a result of interspecies interactions in the wild, however, have rarely been observed [189]. One of the few reported cases is a solitary killer whale who was recorded producing sounds similar to the 'barks' of California sea lions (Zalophus californianus) [190]. Tyack (2008) argues that the definition by Janik and Slater [180, 181] is limiting our understanding of vocal learning to cases where we can perform experiments in captive settings, which does not necessarily reflect what happens in their natural environment. Tyack argues that vocal convergence is a form of production learning, in which the animals match the sounds of conspecifics or individuals of another species [191]. However, developing new tools are necessary to study vocal learning (both contextual and production learning) and communication in wild cetaceans.

2.4 Available tools

Using non-invasive techniques to study the acoustic behaviour of harbour porpoises in their natural environment requires a series of tools, some of which already exist. These include an acoustic recorder that can stay underwater for long periods with minimum or no supervision, a detector of impulsive sounds (*i.e.* potential harbour porpoise clicks), a click classifier, and a series of algorithms to group acoustic events and extracts patterns to identify behaviours. The existing tools are described below.

2.4.1 Passive Acoustic Monitoring - PAM - devices

Visual surveys are the default method to study cetaceans. These range from land and boat-based surveys focused on individual animals using photo-identification [85, 192], to large-scale boat-based and aerial surveys carried out in pre-determined transects to estimate density and abundance [81, 89]. The use of acoustic devices, generally known as passive acoustic monitoring (PAM) systems, to study cetaceans began in the 1960s [193] and became widely used from the 1990s onwards, when many research groups started developing their own devices [see 194]. PAM systems have several advantages over visual surveys as they can be used during bad weather conditions (e.g., high waves, fog) as well as at night, and data collection can be done with no supervision, which means their performance is not affected by the experience or levels of alertness of the observer. Moreover, because cetaceans spend up to 95%of their lives underwater, animals sometimes are only detected acoustically or are detected acoustically before they are detected visually [e.g., 195].

There are now numerous PAM systems to choose from, and the choice depends on the research objectives, location, activity, and species to be monitored, as well as budget and time constrains. Acoustic recording devices can be mounted on vessel hulls [196], towed behind vessels [195, 197, 198, 199, 200], or moored to the seabed, which are also known as static acoustic monitoring (SAM) devices [201, 201, 202, 203]. Other systems include drifting buoys [204, 205] and small acoustic tags that can be attached to animals [e.g., 206]. Furthermore, depending of the objective of the study, one can use a single hydrophone [207, 208] or an array, which in turn can be as simple as two hydrophones, or very complex with several hydrophones (or devices) arranged in a strategic configuration. For example, Thomson et al. [209] used an array of static devices (*i.e.*, C-PODs, Chelonia Ltd., UK) in the Moray Firth, Scotland, to investigate whether harbour porpoises changed their distribution as a result of seismic surveys in the area and Malinka et al. [204] studied the movement of harbour porpoises in tidal areas using PLA-Buoy (Porpoise Localising Array Buoy), a self-contained buoy-based system with a vertical hydrophone array comprising eight hydrophones [204]. The duration of the monitoring period also depends on the type of PAM used, ranging from a few hours (e.q., towed systems [210]) to a couple of days or weeks (e.g., D-tags [206]), to up to several weeks or months (e.g., D-tags [206]), to up to several weeks or months (e.g., D-tags [206]). C-PODs [211]).

Existing SAM devices that can be used specifically to study harbour porpoises can

be divided into two main types: devices that only log data about acoustic events and those that record continuously. The most used data logger is the C-POD and its predecessor the T-POD (Chelonia Ltd., Cornwall, UK). 'T' stands for timing and 'C' for cetacean. These are self-contained devices that can be deployed on the seabed for long periods without supervision. Additionally, they include a built-in classifier (*i.e.*, KERNO) for clicks and to separate click trains produced by harbour porpoises and other NBHF species, from those produced by dolphins, and those of non-animal origin (*e.g.*, sonar) [211]. This separation is based on a series of parameters estimated for each impulsive sound detected, including estimates of peak-to-peak amplitude and peak frequency, as well as inter-click intervals. Both C-PODs and T-PODs have been successfully used to monitor many NBHF species including the vaquita, the harbour and Burmeister's porpoises, and the Heaviside's and Hector's dolphins [56, 106, 212, 213, 214], as well as other odontocetes such as the bottlenose dolphin [*e.g.*, 201].



Figure 2.15: CPOD (top) and DeepC-POD (bottom) - from https://www.chelonia.co.uk

The PODs are used for a wide variety of studies, including seasonal and geographical changes in distribution [215] and response to anthropogenic noise [106, 203, 216], as well as porpoise acoustic behaviour, including diurnal variations in echolocation rates and click train patterns [202]. The embedded algorithms highly reduce the time needed for data analysis and provide researchers with pre-selected data. On the other hand, because the signals are not recorded and the algorithms are not publicly available, verification of the acoustic events is limited and subjective. Furthermore, although the false alarm levels seem to be rather small [e.g., 82], the level of missed clicks is not well understood [217], especially in the context of high background noise levels [218].

Despite this, they showed potential for studying communication in harbour porpoises. Koschinski et al. (2008) used T-PODs to record and describe vocalisation patterns consistent with social calls of a harbour porpoise calf entangled in a fishing net off the coasts of Canada. It is worth noting, however, that the calls were searched for as the authors knew where to look in advance [8]. A new POD is being developed, the F-POD, which would potentially be able to detect social calls [219]. A new alternative to the PODs is the SoundTrap (Ocean Instruments, New Zealand). SoundTraps are self-contained acoustic devices and, unlike PODs, can record continuously, which means that not only verification is possible but also they can be used to study how noise types and levels affect the behaviour of harbour porpoises. Moreover, recordings made with SoundTraps could be used to study harbour porpoise social behaviour and communication. However, recording continuously at high sampling rates (> 300 kHz) required to record harbour porpoise clicks, generates enormous amount of data. Researchers are then faced with two challenges: limitations of storage capacity and available time necessary to analyse the data. Because SoundTraps include a click (*i.e.*, impulsive sound) detector that saves them in small clips, these can be classified directly and the potential harbour porpoise clicks extracted for further analysis. To that end, an automated classification system that can accurately and reliably identify harbour porpoise clicks is required (especially for studies without concurrent visual information) as well as algorithms to group clicks, extract patterns, and identify behaviours.

2.4.2 Detectors and classifiers

In principle, the terms 'detector' and 'classifier' can be used interchangeably. In signal detection theory, the output of a detector is a yes/no answer as to whether the input is a known signal. In this sense, a detector is a classifier ([see 220]). However, when studying phonations of cetaceans, specifically echolocation clicks produced by toothed whales, using these terms interchangeable can be problematic. In much of the literature, the term 'click detector' is used to mean 'click classifier', when in practice, the detection and classification processes are consecutive steps that complement each other. What is usually referred to as 'click detectors' are energy detectors that detect transient sounds (*i.e.*, 'clicks') in the .wav files. These are sounds of short duration with abrupt onset and rapid decay and represent potential echolocation clicks, as the great majority are of non-animal origin and only a small proportion are produced by cetaceans.

One example is the PAMGuard's Click Detector Module. PAMGuard is a modular, open source software aiming to detect and classify marine mammal sounds. It was released in 2006 [221] and it is today one of the most used software for real time monitoring and off-line data analysis. PAMGuard has been shown to be crucial for real-time monitoring of harbour porpoises around underwater noise-producing activities, such as seismic surveys [e.g., 222], as well as for others studies such as fine-scale distribution [90], and monitoring around risk areas for them [204, 223]. This click detector detects impulsive sounds with a signal-to-noise ratio (SNR) over a given threshold (e.q., 6 dB) and extracts them as individual audio clips of very short duration, which includes the signal and a number of samples (e.q., 40 samples)before and after the signal. These clips are then classified as porpoise/no porpoise. The standard settings of the classifier in PAMGuard include a pre-filter (4th order digital Butterworth IIR 10 kHz high-pass filter) and a trigger filter (4th order digital Chebyshev IIR 100-150 kHz band-pass filter, pass band ripple). Clicks are classified as produced by porpoises by comparing the test band (110-150 kHz) to control bands (40-90 kHz and 160-190 kHz), with a 6 dB threshold ('general configuration file – porpoise click detection', available at www.pamguard.com). As an open source software, PAMGuard is regularly improved, and although the user can manage the settings, there is no available information about the performance of the classifier. The precision (*i.e.*, percentage of individual clicks correctly classified as porpoise clicks) reported for an earlier version of this classifier was between 37% and 74%, depending on the settings and background noise, while the proportion of missed clicks was not reported [224]. The user can manually verify the potential porpoise clicks identified by the classifier, and then group them into echolocation events in order to be extracted for further analysis ([92, 116]).

In simple terms, a classification system (*i.e.*, a discriminant function in statistics) assigns a given signal X to one of k pre-defined classes or categories according to a series of parameters or functions, where X is a multivariate random variable. For example, the multivariate random variable X (*i.e.*, person's appearance) comprises

a series of variables such as x_1 = height, x_2 = weight, and x_3 = hair colour. When we talk about harbour porpoise clicks, these variables would be those that can be used for classification purposes, such as peak frequency and duration. It is clear, therefore, that the classification process takes place after the 'detection'. It is crucial to make the distinction between the detection and classification steps, as each is associated to performance errors which must be corrected for/taken into account using different approaches. Moreover, the settings of the detector impact the performance of the classifier, for example, if the detector uses a high SNR threshold, some clicks would not be detected and thus not classified, regardless of how accurate the classification system is.

Classifying cetacean signals have been a research topic of interest for some decades, as many sounds seem to be species specific. At first, detection and classification was restricted to certain species and conditions due to the limitations of the technology available at the time. One of the first systems was developed to study sperm whales, which produce loud, regular, broadband clicks that they emit approximately every second as they dive [38, 225]. The authors were not able to use any temporal information of the signal (*e.g.*, waveform) for classification purposes, but the rate at which sperm whales click is highly regular and this, together with the direction where the signal was coming from (*i.e.*, bearing), allowed them to separate sperm whale clicks from those signals that were likely noise [225]. This simple detector/classifier was improved since then and it is still used to study sperm whales [226]. As technology improves, both in terms of PAM devices and computational capabilities, automatic classifiers for cetacean sounds are regularly developed [*e.g.*, 170, 172, 221, 224, 227]. Classification methods for toothed whales sounds have been developed for several species, for echolocation clicks as well as whistles and other tonal sounds. For example, Real-time Odontocete Call Classification Algorithm (ROCCA) is a MATLAB-based tool that extracts, measures, and classifies whistles. These are described using ten variables, including start and end frequencies, maximum and minimum frequencies, and duration. Subsequently, a multivariate discriminant function analysis (DFA) is used to classify whistles to known groups based on orthogonal linear functions derived from the ten selected variables, and regression tree analysis (CART) to create decision trees that separates data into groups through a series of binary splits [228]. ROCCA can achieve high correct classification rates (*i.e.*, precision) for individual clicks for several toothed whale species, with variations between species, ranging from 14.7% to 63.8%, with an overall correct classification of 33.5% [228].

Similarly, Oswald et al. (2003) [229] used DFA to classify whistles within and between nine different odontocete species. The authors used the spectrograms of the whistles and measured 12 variables that were then incorporated into the model. The classifier achieved between 29.9% to 91.2% accuracy within species, however, the performance between species was lower, between 6.7% and 66% [229]. And Romeu et al. (2017) [230] used DFA to classify whistles within the same species (*i.e.*, bottlenose dolphins - *Tursiops truncatus*) in different behavioural contexts. Dolphins from this population are known to forage cooperatively with fishermen in Laguna, in southern Brazil [231]. The authors used seven variables in this case and used DFA to successfully distinguish between whistles the dolphins produced when engaged in cooperative foraging with fishermen and those produced in other scenarios [230, 231].

Kyhn et al. 2013 [41] studied the spectral characteristics of harbour and the Dall's porpoises, which are sympatric NBHF species off west Canada, and harbour por-

poises in Danish waters. The authors found enough differences in the spectral characteristics of their signals to distinguish between sympatric species (off the west coast of Canada) and animals of the same species but from different geographic areas (Canada and Denmark). To that end, they used a canonical discriminant analysis, using centroid frequency, root-mean-square bandwidth, and duration as explanatory variables [41].

Classification rates are generally higher than what is expected by chance, although there is high variability, which can be explained at least in part by the existing variability in whistles and clicks characteristics even within species, depending on geographic and behavioural contexts. The perfect classifier can hardly exist, however, as not only the temporal and spectral characteristics of the signal might be affected depending on the direction where the signals impinges on the hydrophone [232], but also the performance of the classifier also depends on how the data was collected and extracted.

2.4.3 Pattern identification algorithms

Unlike other sounds produced by cetaceans and other animals, harbour porpoise echolocation clicks are of very short duration (< 100 μ s) and it is thus possible to treat each click as a discrete point in time, much as footsteps. When we hear footsteps we have information about the inter-steps intervals and the amplitude (*i.e.*, loudness) of the sound. If the separation between steps is regular, then we assume these belong to just one person, but if the inter-step intervals are irregular, then it is likely that there are more than one person. Similarly, it is possible to use the relative amplitude of the steps to distinguish between the person closest to us, and whether they are approaching or moving away.

With this in mind, a simple method has been developed to separate overlapping click trains, in this case of bottlenose dolphin clicks [233]. Lepper et al. (2005) first estimated the mean and standard deviation of the distribution of the logarithm of the ICI and amplitude difference between consecutive clicks of their dataset. Then used the standard deviation values as a threshold to decide whether a click belonged to the same click train (*i.e.*, animal - Figure 2.16).



Figure 2.16: Overlapping click trains. Figure extracted from Lepper et al. (2005) [233] showing two overlapping click trains produced by bottlenose dolphins (Tursiops truncatus) and the result of their algorithm (red line)

Another similarly simple algorithm was developed by Starkhammar et al. (2011) [234], which uses the level of similarity between the power spectrum of consecutive clicks. Unlike the previous work, this algorithm begins at a random place of overlapping click trains and estimates the correlation of the power spectrum of consecutive clicks in an iterative process. Clicks with a correlation of 0.95 or more are assumed to belong to the same click train [234].

Both manuscripts reported success in the separation of overlapping click trains, although no specific numbers are provided, these algorithms have shown potential for other species [233, 234]. Both methods were developed using data collected by one hydrophone. Amplitude and inter-click intervals are always available to researchers, whether they are using data loggers or continuous recorders, and in the latter case also the power spectrum can be produced. Both methods can in principle, with minor modifications, work if the click trains are incomplete and if other impulsive sounds and multipath have not yet been removed, which is the case for most recordings made in the wild.

2.4.4 Challenges

There are many sources of noise in the ocean, some of which have energy within the frequency range of the porpoise clicks, therefore using acoustic events or click trains is more reliable than using individual porpoise clicks [219]. This is crucial as even though impulsive noise may be classified as a NBHF click, most noise sources are not patterned as NBHF click trains are. Additionally, background noise levels can degrade the porpoise click if the animal are at a limit of detectable distance from the recorder. The most important challenges are discussed below.

2.4.4.1 Directionality

Echolocation clicks are highly directional and as the main axis of the head of the porpoise moves away from the recording system, the signal is filtered and distorted. Although this is especially true for broadband echolocation clicks produced by delphinids, it does affect NBHF clicks [53, 235].

2.4.4.2 Masking

Certain noise types in the background can mask clicks with low signal-to-noise ratio, which can be missed by either the detector or the classifier. This is important because understanding the underlying behaviour of click train patterns, require access to most, if not all, clicks, which can have low amplitudes at the end (e.g., feeding buzzes).

2.4.4.3 WUTS

WUTS refers to 'weak unknown train sources', and as the name indicates, the sources of these trains are unknown. Although they have been found only using T-PODs and C-PODs, WUTS have been reported in many places, including Australia mangroves, Gulf of Maine, Gulf of Alaska, Gulf of California, and the Baltic Sea. They have not been reported in studies using continuous recordings and everything we know about them comes from PODs [219]. These WUTS seem to have complex patterns of repetition rates, that can range from about 7,000 clicks per second to 100 clicks per second, although never below 40. The key feature is the variation in the peak frequency, which seems to be more random than expected in a NBHF click train. Because the POD does not record the sounds, it is not clear how the energy is concentrated, although they seem to have a minimum bandwidth [219].

2.4.4.4 Sonar

In many areas, fishermen use sonar to find fish. The frequencies are usually set at 50 kHz, 100 kHz, and/or 150 kHz. The last two can be a real problem for click train classification as the energy is within the porpoise range and because the inter-pulse intervals of sonars are highly regular. On the other hand, the regularity is relatively easy to identify and the frequency changes are small compared to variations in click trains.

2.4.4.5 Multipath

Multipath is the propagation phenomenon that results in the same signal to reach the hydrophone from different paths. In the case of harbour porpoise clicks, this can occur when the porpoise is facing the water surface (or the seabed) and the click reaches the hydrophone from the direct path as well as after being reflected in the water surface (or the seabed). Depending on the situation, this phenomenon can impact the performance of the classifier (e.g., the multipath signal retains the same spectral characteristics), or the ability to group clicks belonging to the same echolocation event (e.g., the multipath signal retains the amplitude levels or arrives at intervals that could overlap with the signal arriving through the direct path).

Chapter 3

Data description

This Chapter describes the study areas and how the acoustic data were collected and extracted. In Section 'Study areas and data Collection', both study areas in Scotland and Denmark are described as well as how the data were collected in each. In Section 'Data Extraction', the method used to extract the samples used to developed the algorithms throughout this project is explained. Lastly, in Section 'Derived Data' a description of the main variables used and how they were estimated is presented.

One additional dataset was used in this project. See Chapter 8 - D-PorCCA - from acoustics to behaviour for more information.
3.1 Study areas and data collection

3.1.1 Scotland

The Firth of Clyde (55.5254 N, 4.9333 W) in the West coast of Scotland is a large inlet with high freshwater input, characterized by deep channels and unique sediment types (Figure 3.1). It is an area with high biological diversity, and where at least 15 species of whales, dolphins, and porpoises have been observed in the area.



Figure 3.1: Study area - Scotland. Firth of Clyde.

Harbour porpoises are the most abundant and most widely distributed small cetacean species in the UK [2, 79, 80, 89, 236], and is a resident species in the Firth of Clyde [90], which is one of the areas with higher density of harbour porpoises in Europe, with a density of about 1 animal per square km [80, 81, 237]. The only

abundance estimate for the Firth of Clyde was made over a decade ago from sightings made from July to September 2004, and resulted in 1,645 individuals (CV = 0.351) with a density of 0.823 animal per square km [237]. However, these numbers are likely an underestimation, as Brown (2018) found hot spots where porpoise density is up to 4 animals per square km [90].

3.1.1.1 Data Collection

The acoustic data from the Firth of Clyde used in this project were collected by the CIC Clyde Porpoise during systematic and opportunistic surveys. These were carried out under sail or engine from the 'Saorsa', a 40-foot sailing vessel. In the Firth of Clyde, data were collected in 25 days throughout all seasons, between 2016 (n = 20) and 2017 (n = 5), totalling over 210 hours of recordings. Additionally, data off the north-east coasts of Northern Ireland were collected on the 19th of September 2019 (16 hours). Systematic acoustic surveys were carried out in predetermined transect lines aiming to provide equal coverage probability to the entire study area, which were designed using the software Distance 7. These transect lines were surveyed at a speed between 5 and 7 knots. Data collection was carried out in different weather and background noise conditions (*e.g.*, vessel traffic, sonar activity), during both day and night times. If the sea state reached ≥ 5 in the Beaufort scale (Table 3.1), the survey was terminated. No concurrent visual efforts were made.

Data were collected using a towed hydrophone array connected to a computer running the software PAMGuard [221]. Different versions of PAMGuard were used, starting with version 1.15.10. The hydrophone array included four omnidirectional

WIND	Wind	DESCRIPTION			
Force	Speed (kn)				
0	1	Calm, sea like a mirror.			
1	1-3	Light air, ripples only.			
2	4-6	Light breeze, small wavelets (0.2 m) .			
		Crests have a glassy appearance.			
3	7-10	Gentle breeze, large wavelets (0.6 m).			
		Crests begin to break.			
4	11-16	Moderate breeze, small waves (1 m),			
		some white horses.			
5	17-21	Fresh breeze, moderate waves (1.8 m),			
		many white horses.			
6	22-27	Strong breeze, large waves (3 m),			
		probably some spray.			

Table 3.1: The Beaufort scale (0-6 only), developed by Sir Francis Beaufort in 1805.

hydrophones, comprising one low frequency hydrophone, one mid-frequency hydrophone, and two broadband hydrophones (Figure 3.2). These were two Magrec HP03 hydrophone units, each comprising a spherical ceramic and a HP02 preamp, with a preamp high-pass filter set at 2 kHz. The hydrophones had a sensitivity of -201 dB re $1V/\mu$ Pa at 150 kHz, and a flat response between 2 kHz and 150 kHz. The array was connected to a computer and towed behind the vessel using a Kevlar-strengthened 100 m long cable. The Magrec HP03 hydrophone units were 25 cm apart, enough to obtain direction of arrival of impulsive sounds [238]. Only recordings made with the broadband hydrophones were used in this project.

Recordings were digitised through a St Andrews Instrumentation Ltd. data acquisition card with 16-bit ADC resolution, at a sampling frequency of 500 kHz. Due to the large amounts of data collected at such high sampling frequencies using 4 channels, recordings were saved in 9-min long audio files (size of 2 GB) in *.wav* format. Additionally, the computer was connected to a GPS, and latitude and longitude

were recorded every second.



Figure 3.2: Hydrophone array. Schematic of the hydrophone array system used to collect acoustic data in Scotland.

3.1.2 Denmark

The Great Belt (55.3615 N, 10.9655 E) is the only deep water entrance to the Proper Baltic Sea and thus a high-traffic area, containing the shipping lane known as "Route T", with around 27,000 vessels passing by annually. The Little Belt (55.5295 N, 9.7535 E) is heavily trafficked, with maximum depth of 80 m. The seabed in the area is composed of mud and sand with no kelp. Both areas are known to have high harbour porpoise density in spring and summer months [239].



Figure 3.3: Study area - Denmark. Left: Denmark. Right: Little Belt (LB) and Great Belt (GB). Image © 2015 Google and GeoBasis-DE/BKG (Data: SIO, NOAA, U.S. Navy, NGA, GEBCO).

3.1.2.1 Data collection

Data were collected in Danish waters in 2015 using SoundTrap ST300 (Ocean Instruments, New Zealand), in two different locations (Figure 3.3). The data were obtained from a total of 7 deployments, 5 in the Great Belt area, in August, September, and November, and 2 in the Little Belt area in August. The deployments in the Great Belt were 1 km from the lane "Route T". SoundTraps are self-contained and recordings are made in *.wav* format, at a sampling rate of 576 kHz. The devices were set up in a structure with an anchor (*i.e.*, a cement block), an acoustic release unit (Sonardyne, UK) to recover the devices, and a submerged buoy to counteract the negative buoyancy of the SoundTrap [?].

According to SoundTrap User Guide, "in contrast to traditional hydrophone systems, with SoundTraps there is no need to be concerned with sensitivity in voltage terms. Because SoundTraps integrate the recorder and hydrophone in a single package, there is a fixed relationship between sound pressure and the resultant wav file data, thereby simplifying calibration and eliminating the need for voltage calculations". However, in order to estimate amplitude values in PAMGuard, it is necessary to provide these values. In this project the following were used: sensitivity of -182 $1V/\mu$ Pa, and peak to peak voltage range of 2.0 V as recommended by Ocean Instruments. For more details on how the data were collected see [217].

3.2 Data Extraction

For the completion of this project, two types of data were used: audio files (*.wav* files) which were recorded through PAMGuard, and small audio clips containing individual transient sounds, generated and saved by PAMGuard's 'Click Detector Plug-In' (see below). These are sounds of short duration with abrupt onset and rapid decay and represented potential harbour porpoise echolocation clicks (n = 37,760,282). The *.wav* files were inspected using different versions of Audacity, looking at both the amplitude variations in the time domain (*i.e.*, waveform) and the spectrogram using the default settings (*i.e.*, Hanning window, FFT: 512).

PAMGuard is a modular, open-source software developed to detect and classify marine mammal sounds [221]. The Click Detector Plug-In detects transient sounds above a given signal-to-noise ratio threshold, selected by the user (default = 6 dB). The detected sound is then saved as an individual audio clip, which also includes a very short recording period before and after the impulsive sound detected (default = 40 samples before and after the signal). By default, all impulsive sounds detected in a given hour of recording are individually saved in one *.pgdf* file. PGDF stands for <u>PAMG</u>uard <u>Data File</u> [240].

Individual audio clips from Scotland and Denmark were extracted using a MAT-LAB algorithm developed by J. Macaulay (available here) and adapted for this project. For each audio clip a series of parameters are provided such as date and



Figure 3.4: Top: harbour porpoise clicks (in red) over a period of 20 s as seen in PAMGUARD's 'Click classifier' display (x = time, y = bearing). Bottom: temporal and spectral characteristics of a selected click (grey circle). Left: waveform in both channels. Centre: frequency domain (FFT - 256 samples). Right: Wigner plot.

time and, when there are more than one hydrophone, time of arrival difference (*i.e.*, delay) with respect to the reference hydrophone, and direction of arrival, estimated using trigonometric methods based on said delay [224].

The audio clips extracted from Scottish and Danish data were used to test the different algorithms developed in this project. The entire Scottish and Irish dataset comprises over 3 million clips and the entire Danish dataset over 34 million clips.

3.3 Derived Data

A series of parameters were used throughout the project in audio clips extracted as mentioned in the previous Section. The following paragraphs describe these parameters and how they were estimated where applicable.

- **ID** A unique identification number was assigned to each click in chronological order per day.
- Date and time For clips extracted using PAMGuard, the programme provides date and time in numerical format, which was used like that or converted into text format (*e.g.*, 25 Jul 2017 15:28:34). In case of clips created from *.wav* files using a click detector developed for this project, date and time was obtained from the *.wav* file metadata and estimated for each clip using sampling frequency and number of samples.
- Start sample Sample number where the click begins per day (in PAM-Guard) or within the *.wav* file for the detector developed in this project. The sample number is used to estimate the inter-click interval.
- Duration (μs) The duration of the signal was estimated as the 80% energy of the clip that contains the signal [238]. The samples PAMGuard saves contain the signal and samples before and after, which may sometimes include an echo. Moreover, the echo might arrive at the same time as the signal and thus the energy in the sample would be greater than that of the signal itself, therefore using 80% instead of 95 or 97% provided a more accurate approximation of the duration of the click of interest for modelling and comparison purposes.
- Amplitude (dB re 1 μ Pa) Peak amplitude of the signal.
- Peak frequency (PF, kHz) Frequency with the highest amplitude in the frequency domain (FFT = 256 points) [238].
- Centroid frequency (CF, kHz) The point dividing the spectrum in halves

of equal energy, derived by the squared pressure over a 256-point (128 μ s) window, symmetrical around the peak of the signal envelope [241].

- -3dB Bandwidth (BW_{-3dB}, kHz) The -3 dB bandwidth is defined as the bandwidth around the peak frequency that contains half of the signal power [238].
- **RMS Bandwidth** (*BW_{RMS}*, **kHz**) The root mean square bandwidth is defined as the spectral standard deviation around the centroid frequency on a linear scale. (Algorithm courtesy of J. Tougaard) [238].
- Ambiguous bearing (degrees) Direction to the sound source estimated using time of arrival differences (*i.e.*, delays). This information is available in the output of PAMGuard's Click Detector [224]. It is ambiguous because there is no information on where in the water column the source is situated, nor which side of the hydrophone array the signal is coming from.
- Cross Correlation (XC) Maximum cross-correlation (also known as 'matched filtering') coefficient between the clip and a typical harbour porpoise click. The click used as a 'template' is a click extracted from the snip samples, selected based on the waveform characteristics and peak-to-peak amplitude (~162dB re: 1μPa) [235, 238]. Additionally, the waveform was consistent in both hydrophones and the time of arrival difference between them was 0 (*i.e.*, the orientation of the animal was likely towards the array).
- Q_{RMS} Relative width of the signal, and is estimated as the ratio between centroid frequency and BW_{RMS} [33]
- Inter-click interval (ICI, ms) Time difference between the click and the

previous one. The inter-click interval for the first click in a click train (click ID = 1) is 0.

• Clicks per second (CPS) - The number of clicks per seconds is estimated as 1000 divided the inter-click interval.

These variables and parameters are regularly used in studies on harbour porpoise acoustics characteristics and behaviour, and they are used throughout the thesis for different purposes. In Chapter 4, Q_{RMS} , duration, XC, CF, ratio, and BW_{-3dB} are used to develop a harbour porpoise click classifier for high and low-quality harbour porpoise clicks. In Chapter 5, the repetition rate (*i.e.*, clicks per second) and CF are used to identify click train types, subsequently the repetition rate and the amplitude are used to clean click trains from unwanted sources of sound, and lastly repetition rate is used to identify behaviours. Finally, these are used in different ways in D-PorCCA, the standalone application developed in this project, described in detail in Chapter 8. These parameters and variables are used for visualisation purposes as well as to generate summary data the user can download for further analysis.

Chapter 4

PorCC: the harbour porpoise click classifier

4.1 Introduction

The available acoustic continuous recordings made using passive acoustic monitoring (PAM) systems could be used to fill gaps in our understanding of harbour porpoise behaviour and communication in the wild. To that end, however, a classification system that can accurately and reliably identify as many harbour porpoise clicks as possible within an echolocation event is required. A classification system, in simple terms, assigns a given signal x to one of k pre-defined classes according to a series of parameters or functions. For continuous recordings, one of the most used harbour porpoise detector/classifier systems is PAMGuard's Click Detector and Classifier Module, which is used for real-time as well as offline detections and classifications [92, 116, 242]. PAMGuard is a modular, open-source software designed and devel-

oped specifically to detect and classify marine mammal sounds [221]. The standard settings of the porpoise classifier includes a pre-filter (4th order digital Butterworth IIR 10 kHz high-pass filter) and a trigger filter (4th order digital Chebyshev IIR 100-150 kHz band-pass filter, pass-band ripple). Clicks are classified by comparing the test band (110-150 kHz) to control bands (40-90 kHz and 160-190 kHz), with a 6 dB threshold ('general configuration file – porpoise click detection', available at PAM-Guard's website.) As an open source software, PAMGuard is regularly improved, and although the user can manage many of the settings of the detector/classifier, there is no available information about its performance. The precision (*i.e.*, percentage of individual clicks correctly classified as porpoise clicks) reported for an earlier version of this classifier was between 37% and 74%, depending on the settings and background noise, although the proportion of missed clicks was not reported [224]. The performance of the current version remains unquantified (at the time of writing). The user can verify the classifications through the Click Detector Display (Figure 3.4) after the identified clicks have been highlighted and extract clicks for further analysis by either selecting everything or by selecting individual echolocation events [92, 116]. Alternatively, some researchers use custom-built classifiers, of which neither the algorithm, nor the performance are publicly available. As acoustic recordings continue to accumulate, assessing the performance of available classifiers for comparison purposes and automating these processes becomes essential.

Recently, different modelling and machines learning methods have been developed and applied to classify cetacean sounds [e.g., 170, 172, 221, 224, 227]. One of such methods is the discriminant function analysis (DFA), which can predict a categorical response variable from a group of 4 or more numeric explanatory variables. Multivariate DFA has been used to classify cetacean sounds, both for odontocetes and

mystecetes species, with varied results [e.g., 228]. If the response variable is binary (e.g., yes/no), there is an approach that makes fewer assumptions than DFA and it is more appropriate. This method is the generalised linear model (GLM), more specifically, the logistic regression model. GLMs are parametric because a probability distribution (binomial, in this case) is specified for the response variable. The logistic regression model can be used to estimate the probability that a given signal is a known signal, based on a series of predictor variables. For example, the multivariate random variable X (*i.e.*, harbour porpoise click) comprises a series of variables such as x_1 = peak frequency, x_2 = duration, and x_3 = -3dB bandwidth. The model to be fitted is

$$g(\mathbf{x}) = \beta_0 + \beta_1 * x_1 + \beta_2 * x_2 + \beta_3 * x_3 \dots$$

Where β_i are the coefficients to be estimated; x_i are the values of the explanatory variables; g(x) is the logarithm base 10 (log) of the odds ratio of $y_i = 1$ vs $y_i = 0$ (*i.e.*, the log of the odds of a signal being a porpoise relative to not being one); and β_0 is the intercept or constant (*i.e.* the log of the odds of a signal being a porpoise relative to not being one when all explanatory variables are zero). Thus logistic regression models divide the data in two, on one side what we are looking for and on the other, everything else. Machine learning methods are based on consecutive logistic regression models.

Thus, logistic regression models make an excellent candidate to identify harbour porpoise clicks, which are highly stereotyped sounds [6, 7, 11].

4.2 Methods

4.2.1 Data collection and extraction

The data used for this Chapter were collected in the Firth of Clyde (55.5254 N, 4.9333 W), Scotland, by the Porpoise Clyde CIC. Surveys were conducted in 25 days throughout all seasons, between 2016 (n = 20) and 2017 (n = 5), totalling over 210 hours of recordings. No concurrent visual efforts were made. Surveys were carried out under sail or engine from the 'Saorsa', a 40-foot sailing vessel. See Chapter 3 - Data description for a detailed description of data collection and extraction.

4.2.2 Data exploration

To develop the classifier, a total of six potential explanatory variables were chosen, four of which are regularly used to describe harbour porpoise clicks. These are duration, centroid frequency (CF, kHz), -3dB bandwidth (BW_{-3dB} , kHz), and Q_{RMS} . Additionally, the ratio between the peak and centroid frequencies (Ratio) and the peak value of a cross correlation (XC) performed against a typical harbour porpoise click were used. This typical click was extracted from the original dataset, and was selected based on the characteristics of the waveform, the power spectrum, and the spectrogram, as well as the peak-to-peak amplitude (162dB re: 1 μ Pa). Additionally, the waveform was consistent and overlapping in both hydrophones, as the time of arrival difference between them was 0, therefore the orientation of the animal was perpendicular to the array (See Chapter 3 - Data description for more details on the estimation of these variables).

Before developing the logistic regression models, the following assumptions were

tested when required:

-*Linear relationship*: The logistic regression model does not require a linear relationship between the response and explanatory variables.

-Normality: The error terms (residuals) do not need to be normally distributed.

-*Homoscedasticity*: the variance of the explanatory variables are not required to be similar for the logistic regression model.

-*Independent observations*: the observations should not come from repeated measurements or matched data. To address this requirement, clicks were selected using a random number generator.

-*Multicollinearity*: Multicollinearity is when the explanatory variables are correlated with each other. The logistic regression model requires there to be little or no multicollinearity. Multicollinearity was tested using the Pearson $\tilde{\chi}^2$ coefficient. None of the variable pairs had a correlation coefficient higher than ± 0.36 , except for Q_{RMS} and XC that had a correlation of 0.49 Figure 4.1. 'Multicollinearity was also tested using the generalised variance inflection factor (GVIF) and all pairs of variables had a GVIF value below 2, indicated they were not correlated.

-*Linearity*: The logistic regression assumes linearity of explanatory variables and log odds. Although this analysis does not require the response and explanatory variables to be related linearly, it does require that the explanatory variables are linearly related to the log odds.

-Sample size: This model requires a large sample size (n), which can be estimated as $n = 100 + x^* i$, where x is 50 for logistic regression models, and i is the number of explanatory variables [243]. Thus, having a maximum of six explanatory variables, n must be at least 400. A total of 5,000 samples were used per model, 500 signals of interest against 4,500 unwanted signals.



porpoise click classifier. The diagonal contains the histograms of each of the variables. The numbers represents the Pearson's $\tilde{\chi}^2$ correlation coefficient. Figure 4.1: Multicollinearity. Pair plots of predictor variables used in the logistic regression models for the harbour

4.2.3 Training data

Three categories of signals were defined for the development of the harbour porpoise click classifier (PorCC - for Porpoise Click Classifier): high quality porpoise clicks (HQ), low-quality porpoise clicks (LQ), and high-frequency noise (N), as shown in Figure 4.2. HQ are polycyclic signals with peak frequency between 100 and 160 kHz, no spectral energy below 100 kHz, and duration around 100 μ s (Figure 4.2a), matching the description of on-axis harbour porpoise clicks [6, 7, 235]. LQ are signals slightly different to HQ (Figure 4.2b), for example presenting notches in the power spectrum, or no clear beginning or end of the signal when looking at the waveform. The need for this distinction is clear when investigating click trains, as these almost always contain a series of consecutive clicks with consistent the interclick interval and amplitude variations, indicating that these are indeed clicks within the click train, but that are not of enough quality. Thus, LQ clicks comprise both off-axis and low amplitude clicks. In the decision-making path, clips with peak and centroid frequencies outside the range of interest (100-160 kHz) are discarded before the classification process takes place. Noise clips (N) are, therefore, signals with peak and centroid frequencies between 100 and 160 kHz but that do not share other characteristics with harbour porpoise clicks (e.g., oligocyclic, do have energy below 100 kHz) (Figure 4.2c).

Over 2,500,000 impulsive sounds were detected and saved as small audio clips by PAMGuard's Click Detector Module from the entire survey period (> 210 hours of recordings). The training data selected to develop the classifier consisted of a subsample of 125,416 (5% of the data) clips selected using a random number generator to ensure independence of the samples. Of these, 10,000 clips were further



Figure 4.2: Categories. Examples of the categories defined to develop the harbour porpoise click classifier (PorCC). A) High-quality harbour porpoise click (HQ). B) low-quality harbour porpoise click (LQ). C) high-frequency noise (N).

selected to develop a simple logistic regression model, which was used to aid in finding appropriate signals to develop the final logistic regression models for PorCC. This pre-step was necessary as it was not realistic to individually check all of the 125,416 clips to find samples that were representative of each of the defined categories. One volunteer (P. Martinez) and MC labelled these 10,000 clips individually, based on the characteristics described above (Figure 4.2). Because the samples were not consecutive, and because opening large (~ 2 GB) audio files in search of a specific click in a spectrogram is a time-consuming task, the spectrogram of the original recordings were not used in this step. This could have led to mistakes when labelling clicks, however, it is expected that such mistakes were not significant for the overall

training method, especially for the HQ category, as there are no other animal or human activity carried out in the area capable of producing sounds similar to porpoise clicks.

The logistic regression model estimates coefficients for each explanatory variable in the model and the error term, from which a probability of a given signal belonging to a specific category can be derived, in this case the probability that a given clip is a harbour porpoise (HQ or LQ) click. Thus, each of the 125,416 audio clips was assigned a probability of being a harbour porpoise click using the coefficients of the pre-classifier model. The probabilities ranged from 0 to 1. Those with high probability (≥ 0.9) were considered potential HQ, those with a probability between 0.5 and 0.9 were considered potential LQ, and those with a probability < 0.5 were considered potential N clips.

For the final step, a total of 5,500 signals were randomly selected using a random number generator. From these, 500 samples were selected from the potential HQ samples, 500 from the potential LQ samples, and 4,500 from the potential N samples. Before using these clips for training of the logistic regression models, all samples (n = 5,500) were verified individually to ensure each selected clip was a good representative of its respective category, removing unrepresentative clips and replacing them with clips randomly selected, again, from the original subsample for that particular category.

4.2.4 Logistic regression - PorCC Classification algorithm

All models and the final algorithm of the harbour porpoise click classifier (PorCC) were written in MATLAB 2017a (The Math Works TM, Inc., Natick, MA, USA).

The response variable for 'Model HQ' is binomial, with the possible outcomes being HQ and N, and it was built using 500 HQ and 4,500 N clips. The response variable for 'Model LQ' is also binomial with the possible outcomes being LQ and N, and it was built using 500 and 4,500 clips of each, respectively. The same N clips were used for both models to ensure that the training of the models was consistent. For each logistic regression model, a series of 63 reduced models were tested, which were built using all possible combinations of explanatory variables, without repetitions (*i.e.*, starting from all 6 variables combined, reducing until using just one variable at a time - see Appendix B). The best 'Model HQ' and 'Model LQ' were identified as the one with the lowest Akaike Information Criteria (AIC) value [244].

Although PorCC runs on clips previously saved by PAMGuard's Click Detector in *.pgdf* binary files, the coefficients and the decision making pathway can be applied to audio clips generated by any impulsive sound detector. The algorithms to extract the data from *.pgdf* using MATLAB, available online, were modified to function within PorCC. If the recordings were made using two or more hydrophones, the classification process takes place only in the sounds recorded by the first hydrophone on which they impinged, as these are expected to be of higher quality (*i.e.*, closer to the sound source). The decision-making pathway for each clip within PorCC is show in Figure 4.3.

The PorCC algorithm is included in D-PorCCA (see Chapter 8) where the user can choose threshold values for high- and low-quality clicks.



Figure 4.3: Decision-making pathway. Flowchart illustrating the decision-making pathway of PorCC, the harbour porpoise click classifier. CF = centroid frequency. PF = peak frequency. Th = threshold. Prob = Probability.

4.2.5 Testing data

To test the performance of PorCC against signals manually labelled, two datasets were created. The first dataset contained all clips (n = 265,918) extracted from 5% of the *.pgdf* files, which were selected using a random number generator. The second dataset contained all clips (n = 284,231) generated from all the recordings made on the 28th of August 2017, corresponding to 8 *.pgdf* files (*i.e.*, each *.pgdf* file stored all impulsive sounds detected in one hour of recording). Clips with peak and centroid frequencies between 100 kHz and 160 kHz and $Q_{RMS} \ge 4$ represented potential harbour porpoise clicks. These clips (n = 70,689) were extracted and manually labelled according to the three categories previously defined (Figure 4.2), based on the characteristics of the waveform, power spectrum, and spectrogram.

The overlap between the training (n = 5,500) and the testing (n = 70,689) datasets was of 442 clips only. The probability-threshold values used in PorCC were 0.9999 for HQ and 0.55 for LQ (Figure 4.3).

Confusion matrices and receiver operational characteristics (ROC) curves were used to assess the performance of PorCC against manual labelling. One harbour porpoise researcher labelled 500 clips of all three categories defined before and the level of agreement was 95%, therefore it was decided that using manual labelling as the reference was appropriate. Confusion matrices show how clips labelled manually are classified by PorCC. From the matrix it is possible to calculate hit rates, as well as the rate of misclassification (*i.e.*, false alarm and missed clicks) and precision levels. The hit rate in the strict criterion is the proportion of HQ (or LQ) clips classified as HQ (or LQ). The relaxed criterion is the proportion of HQ and LQ clips classified as either HQ or LQ. Three analyses were performed: one with only HQ clips, one with only LQ clips, and one with both HQ and LQ clips, all against the N clips. The false-alarm rate is the number of N clips classified as HQ (strict criterion), or HQ or LQ (relaxed criterion) divided by the total number of N clips. The missed-clicks rate is defined as 1 minus the hit rate. The precision is defined as the number of clips correctly classified divided by the total number of clips classified into that category. In total, this results in six different points of operation, from which the ROC plot is generated.

4.2.6 PorCC vs PAMGuard

A subset of the testing dataset for PorCC was also used to assess the performance of the porpoise click classifier built-in in PAMGuard and compared it to that of

PorCC. This dataset subset contained all clips from the 28th of August 2017 (n =284,231) of which 30,897 clips had already been manually labelled. PAMGuard's classifier highlights potential harbour porpoise clicks as well as potential echoes (i.e.,multipath) that the user can verify and group into 'acoustic events' to later extract them for further analysis. For the purpose of this study, all highlighted clicks (including echoes) for the entire day were selected without verification, assigned to a unique acoustic event, and exported to an SQL database. Potential echoes were included because it was previously noted by Cosentino that PAMGuard's classifier sometimes misidentifies real harbour porpoise clicks as echoes (and vice versa) as well as for comparison purposes, as PorCC also identifies potential echoes (LQ). PAMGuard creates a table within the SQLite database, where information for each of the extracted potential harbour porpoise clicks is provided, including date, time, and an identification number within the .pgdf file where the waveform is saved. Using a custom-built script, and using the identification number, potential harbour porpoise clicks were extracted and saved in a MATLAB structure array for further analysis. Subsequently, clips that were highlighted by PAMGuard's classifier but were discarded by PorCC (*i.e.*, clips with peak and centroid frequencies outside the range of interest) were manually labelled.

Confusion matrices and receiver operational characteristics (ROC) curves were used to assess the performance of PAMGuard's classifier and to compare it against the performance of PorCC. False alarm, hit rates, and precision levels were also estimated, as well as the detectability index (d') [220, 245].

4.3 Results and Discussion

This section presents the result of this Chapter and discusses the relevance in context.

4.3.1 Logistic regression models

According to AIC values, the best 'Model HQ' was that with only Q_{RMS} and duration as explanatory variables. This was expected, as the histogram and distribution of each of the explanatory variables has shown that these were the only variables where there was little overlap for the three categories (see the upper left and bottom right panels in Figure 4.4).



Figure 4.4: Histogram and distribution of the variables used to develop PorCC. Black line: high quality clicks (HQ). Black dashed line: low quality clicks (LQ). Gray line: noise clips (N). The variables in the lower panel do not have a unit. Y axis truncated for the cross-correlation coefficient and duration

The best 'Model LQ' included all six explanatory variables: Q_{RMS} , Duration, Ratio, XC, CF, and BW_{-3dB} . This was also expected and for similar reasons, as

there was significant overlap for all variables in LQ and N categories (Figure 4.4). The best five models are show in Table 4.1. The table with all the results is in Appendix B. It is worth noting that in both cases, when looking at models with only one explanatory variable, the model with the cross-correlation coefficient (XC) as the only explanatory variable appears in the second position after Q_{RMS} . There is enough variation in the temporal and spectral characteristics of porpoise clicks, which are not blueprints of each other, and so a cross-correlation would not perform well. The cross-correlation coefficient value, in fact, ranged from 0.0038 to 4.5655, and thus using a simple threshold for a XC coefficient in a decision-making process would inevitably include HQ as well as LQ and N samples, thus rendering a low performance. In this case the samples were not scaled before the cross correlation was performed, which may also have affected the performance. The training and testing data were both obtained from wild individuals that were engaged in natural behaviour, and it included off and on-axis clicks and thus there is more variability than could be expected if the classifier had been developed using clicks recorded from captive individuals in controlled environments.

Table 4.1: Logistic regression models. Series of logistic regression models for 'Model HQ' and 'Model LQ'. Only the best five models are shown here. See text for description of the variables used. The outcomes of the response variable for 'Model HQ' are high-quality harbour porpoise clicks or high frequency noise, and for 'Model LQ' are low-quality harbour porpoise click or high-frequency noise. AIC = Akaike's Information Criterion.

ID	Explanatory variables		
	Model HQ		
1	Q_{RMS} + Duration	-	
2	Q_{RMS} + Duration + Ratio	1.64	
3	Q_{RMS} + Duration + BW_{-3dB}	1.67	
4	Q_{RMS} + Duration + CF	1.78	
5	Q_{RMS} + Duration + XC	1.96	
	Model LQ		
1	Q_{RMS} + Duration + Ratio + XC + CF + BW_{-3dB}	-	
2	Q_{RMS} + Duration + Ratio + XC + BW_{-3dB}	1.19	
3	Q_{RMS} + Duration + Ratio + CF + BW_{-3dB}	19.19	
4	Q_{RMS} + Duration + XC + CF + BW_{-3dB}	20.07	
5	Q_{RMS} + Duration + Ratio + BW_{-3dB}	20.87	

4.3.2 PorCC performance

The perfect classifier cannot exist, as detection always will be limited by noise, either external from the environment, or internal. For electronic systems this internal noise is in amplifiers and hydrophones, and for biological systems, this noise will be in the form of spontaneous activity in the neurons. In real-world applications, noise also comes in the form of substantial variation in the temporal and spectral characteristics of acoustic signals. These are affected by many factors, including background noise and the direction from where the signals impinge on the hydrophone, as well as by how the data were collected (*e.g.*, hydrophone own noise, frequency characteristics of the hydrophones) [26].

PorCC classification process, including estimating all necessary parameters, takes

approximately 1 ms per clip. Harbour porpoises produce between fewer than 10 and a few hundred clicks per second depending on their behaviour [12, 13, 107], PorCC shows, therefore, potential for real time application. The performance values for PorCC are shown in Table 4.2. As precision increases hit rate decreases, that is, fewer clicks of the total available to the classifier are going to be identified, demonstrating the well-known trade-off between errors: false alarms vs. misses in signal detection and Type I vs. Type II errors in conventional statistics.



Figure 4.5: Receiver operating characteristics (ROC) curves. Dots represent false alarm rates and hit rates associated with detection of HQ-clicks (solid black line), LQ-clicks (black dashed line) and both types combined (grey line), all against a background of N-clicks. Curves are best fitting ROC-curves, generated under the assumption of Gaussian underlying distributions with equal variance. Figures show performance by PorCC under two different criteria: strict (only clicks classified by PorCC as HQ) and relaxed (all clicks classified as either LQ or HQ). Figures to the left and right contain same data, but right figures are plotted on double probit (probability) axes

PorCC is not exempt of errors, and attempts to increase the hit rate would lead

to an increase in the false alarm rate (Figure 4.5), as seen in the change in performance going from a strict (using only HQ clicks) to a relaxed criterion (using both HQ and LQ). In other words, the smaller the probability threshold for the classifier, the larger the number of clicks that would be identified and greater the error too. However, the ultimate goal in acoustic event detection and classification is not to avoid errors, but to manage them. Thus, PorCC provides the user with a general assessment of its performance through the ROC curves, as these show the changes in hit rate with false alarm variations [220], which results from using different threshold values to classify harbour porpoise clicks.

Users can, *a priori*, manage the level of error according to the project needs using the ROC curves as reference to understand the expected level of error. Furthermore, because PorCC distinguishes between high and low-quality clicks, depending on the objectives of the study, users can focus on either or both HQ and LQ clicks as well as decide when LQ clicks should be ignored (*e.g.*, single LQ clicks) or taken into account (*e.g.*, studies of click train patterns - see Chapter 8).

It is expected that when using mobile and static recorders, many clicks within a click train would be off-axis and therefore any study aiming to understand the repetition pattern of harbour porpoise clicks in the context of their behaviour must include both HQ and LQ clicks. Available data suggest in fact that the variation pattern of inter-click intervals within a click train is indicative of specific behaviours [8, 11, 12, 13, 246]. The relatively-well understood pattern is that of foraging. The porpoise first investigates the surroundings emitting between 10 to 50 clicks per second. The interval between consecutive clicks is related to the distance to the target, thus the closer the porpoise gets to the prey, so the repetition rate increases, as the echoes also return faster too. It is usually agreed that when the ICI reaches less

than 10 ms (*i.e.*, over 100 clicks per second), the porpoise is in the final stages close enough to the prey to make a catch, known as a feeding buzz.

The PorCC classification algorithm can be used with the output of any transientsound detector for continuous recordings and, given the low misidentification levels, it is suitable for behavioural studies of wild harbour porpoises and other species that produce similar sounds, as the variations in inter-click intervals can only be detected if the majority of clicks within a click train are identified. Moreover, these studies can be carried out in data that has already been collected using either towed hydrophone arrays or static devices, such as SoundTraps (Ocean Instruments, New Zealand). The PorCC algorithm, including the functions to estimate the different variables and the resulting coefficients, are part of D-PorCCA, an application developed specifically to study harbour porpoises in the wild. Additionally, they are publicly available at the Pure Data Repository of Strathclyde University (here) and can be coded in other programming languages, such as Python. It could also be incorporated into PAMGuard.

4.3.3 PorCC vs PAMGUARD

A total of 30,897 clips from the 28th of August met the criteria for potential harbour porpoise clicks, that is, having peak and centroid frequency between 100 and 160 kHz and Q_{RMS} higher than 4. The results of the comparison of the performance of PorCC (using Th1 \geq 0.9999 and Th2 \geq 0.55) and PAMGuad's classifier for HQ are shown in Table 4.2. Based on the detectability indexes, PorCC outperforms PAMGuard's classifier in all cases, but especially for HQ clicks (Figure 4.5).

rpoise arbour 19 that	Noise	382	2,490	533, 228			
nd the Pc quality h , anythir	PorCC LQ	1,710	2,982	4,303	had ere		
orCC a = high- oise (i.e	Н	4,475	519	60	they we		
between H rCC. HQ equency n	Total	6,567	5,991	537, 591	poise click , therefore	ck).	lips).
ion levels nce of Por and low-fr	Noise	113^{b}	1,283	280,034	rbour por kHz range	of HQ cli	ber of N c
identificat verforman = high-	PorCC LQ	565	436	2,093	tential ha and 160 ł ep.	l number	otal num
ınd misi overall 1 3. Noise	Н	1,209	23	25^c	d as pot the 100 first ste	the tota	by the t
of correct c tard, and poise clicks	Noise	$1,269^{b}$	I	279, 355	PAMGuar putside of t ded at the	ivided by t	cs divided
Comparison (Le in PAMGu y harbour por _l	PAMGuard LQ	1	375	I	ighlighted by l frequencies c ey were discar	tsified as N d	ed as HQ click
matrices. fier Modu low-qualit :).	Н	564^a	I	$477 + 1601^{a,c}$)17 clips h id centroic CC, as the) clicks cla	ps classifi
Confusion ctor/Classi icks. LQ - rpoise click	Total	1,833	965	281,433	total of 3,(nd peak ar red by Por	clicks (HG	larm (N cli
Table 4.2: Click Dete porpoise cli is not a po	Labelled	НQ	LQ	Noise	a) Of the Q_{RMS} i4 a not captu	b) Missed	c) False a

The overall precision for HQ for PorCC was 69.2% while it was 30.8% for PAM-Guard's classifier, assuming that PAMGuard's classifier correctly classified HQ and LQ as such in 100% of the cases, as once clicks are extracted from PAMGuard, there is no information of whether a clip was originally classified as a harbour porpoise click or an echo, which can be considered as equivalent to the HQ and LQ categories (Figure 4.6).



Figure 4.6: Receiver operating characteristics (ROC) curves. Dots represent false alarm rates and hit rates associated with detection of HQ-clicks (solid black line), LQ-clicks (black dashed line) and both types combined (grey line), all against a background of N-clicks. Curves are best fitting ROC-curves, generated under the assumption of Gaussian underlying distributions with equal variance. Figures to the left and right contain same data, but right figures are plotted on double probit (probability) axes

Although the data used to generate the ROC curves fall in the lower part of the curve (Figure 4.5 and Figure 4.6) and thus may be misleading, the false alarm rate, defined as N clips classified as HQ clicks, for PAMGuard's classifier was was between 19.08 (n = 477) and 83.1 (n = 2,078) times higher than PorCC's, when

using the same dataset (Table 4.2). Moreover, PorCC algorithm correctly classified 65.9% of HQ clicks (n = 1,209), while PAMGuard's did so in 30.7% of HQ clicks (Table 4.2).

4.4 Applications

The PorCC algorithm is one of the algorithms within the application D-PorCCA, developed during this project (see Chapter 8 D-PorCCA - from acoustics to behaviour). Within D-PorCCA, once all HQ and LQ are classified and information about them stored, clicks are grouped into click trains. A click train is defined as a series of clicks (default = a minimum of 16 clicks and a maximum of 1,500 clicks) separated by less than a given time period (default = 1 s). See Chapter 5 Patterns of click trains and Chapter 6 Acoustic behaviour for a description of how click trains are analysed.

Although PorCC was trained and tested in data collected using a towed hydrophone array, given the nature of the harbour porpoise signals, it was expected that PorCC could be used to classify acoustic data collected using other devices, in this case SoundTrap (Ocean Instruments, New Zealand). SoundTraps are self-contained recorders with one hydrophone that can be deployed for long periods of time moored to the seabed or deployed temporarily, for example hanging over the side of a still vessel or the shore. Additionally, it was expected that PorCC would perform equally well for clicks produced by porpoises from a different population, in this case Denmark. Lastly, given narrow-band high-frequency (NBHF) clicks produced by harbour porpoises are similar to NBHF clicks produced by other species, such as Heaviside's dolphins (*Cephalorhynchus heavisidii*), it was expected that PorCC could be used for these species too. A series of tests were performed to investigate these hypotheses and the results are shown in Chapter 5 - Patterns of click trains, in Chapter 6 - Acoustic behaviour, and Chapter 8 - D-PorCCA - from acoustics to behaviour.

4.5 Summary

PorCC is an acoustic classifier developed using machine learning techniques that has been shown to be highly accurate in identifying NBHF clicks produced by harbour porpoises, with improved performance over the currently available classifier in PAMGuard. The algorithm is fast and thus shows potential for real time applications. Moreover, PorCC can be applied to data collected using both mobile and static PAM systems and in recording from different harbour porpoise populations. Furthermore, it can be used to study other species that produce NBHF clicks including other porpoise and dolphin species.

The inclusion of high and low quality clicks in the classification process means that most, if not all, clicks within a click train would be identified. The result of this study suggest that PorCC is an ideal tool for behavioural studies based on vocalisation patterns for harbour porpoises and other NBHF species.

Chapter 5

Patterns of click trains

5.1 Introduction

Acoustic data from animals in captive settings as well as in their natural environment have shown harbour porpoises emit clicks almost continuously and that these clicks are not produced randomly. Click production is patterned differently depending on the behaviour of the porpoise [e.g. 10, 11, 12]. Because clicks are very short in duration (~ 100 μ s), they can be treated as discrete points in time, from where patterns of repetition rates can be identified. Data from acoustic tags are especially good for this purpose, as tags usually also collect data on animal movement as well as other environmental parameters such as water depths [13, 108].

Finding these patterns in data collected using passive acoustic monitoring, however, is not straightforward. First and foremost, click trains are not always complete because clicks are emitted in a narrow beam and porpoises are not always oriented towards the recorder. This is problematic because identifying patterns require access to most if not all clicks within a click train. A strict classifier that only identifies onaxis clicks would have high levels of false negatives, while a classifier with relaxed criteria will have high levels of false positives. The PorCC algorithm developed in this thesis (Chapter 4 - PorCC: the harbour porpoise click classifier [15]) includes two classifiers, a strict one for high-quality clicks (HQ), and a relaxed one for low-quality clicks (LQ). This means that click trains may include other sources of sounds, such as high-frequency noise or multipath signals, incorrectly classified as LQ. Therefore, to identify underlying patterns it is necessary to first remove undesired sounds.

A simple method developed to separate overlapping click trains of bottlenose dolphin (*Tursiops truncatus*) provides a good approach to this problem [233]. Lepper et al. (2005) used information about inter-click interval (ICI) and amplitude variations within click trains to group clicks from the same click train (*i.e.*, animal). The algorithm begins at the first click and identifies the following ones based on predefined threshold of inter-click interval and amplitude variations. These thresholds are defined as the standard deviation of the distribution of the logarithm of the ICI and amplitude differences of consecutive clicks [233]. Therefore, in this case it is assumed that the thresholds are fixed. Additionally, for these thresholds to work, the detector/classifier system used must have identified all clicks in the click train, and it also should be cleaned of noise incorrectly classified as echolocation clicks. Following the same logic, we can assume that there is a click train of interest and we can use the same approach to separate it from noise misclassified as clicks, using adaptable thresholds instead of pre-defined ones, as harbour porpoises can begin clicking with ICIs of over 50 ms and end at ICIs below 1 ms [*e.g.*, 8, 9, 10, 11, 12, 13, 108].

This Chapter presents a series of consecutive algorithms based on the ideas devel-

oped by Lepper et al. (2005) adapted to remove unwanted sound sources within click trains, exposing underlying patterns, and identifying behaviours.

5.2 Material and Methods

Two datasets were used to develop and test the algorithms in this Chapter. One dataset was constructed with acoustic data collected in two locations in Danish waters, between the Baltic Sea and the Kattegat Strait, and the other with acoustic data collected in the Firth of Clyde, Scotland, and off Northern Ireland.

5.2.1 Data collection and extraction

The acoustic data consisted of individual impulsive sounds (*i.e.*, potential harbour porpoise clicks) extracted from the *.pgdf* files generated by the Click Detector Module in PAMGuard, and classified using the Porpoise Click Classifier PorCC (Chapter 4 - PorCC: the harbour porpoise click classifier) with the default values. The default threshold value for high-quality (HQ) harbour porpoise clicks was 0.9999 and 0.55 for low-quality (LQ) ones. Data from Denmark were collected between August and November of 2015 using SoundTraps (Ocean Instruments, New Zealand), and the data from the Firth of Clyde and off Ireland were collected between 2016 and 2019 using a towed hydrophone array. The recordings (*i.e.*, *.wav* files) were used for verification purposes when available (*i.e.*, data from Scotland).

In this Chapter, the parameters used were peak-to-peak amplitude (dB re: 1μ Pa) and inter-click interval (ICI) in ms. See Chapter 3 - Data description for a detailed description of how the data were collected and extracted, and parameters estimated. Clicks were grouped in series of clicks separated from each other by no more than 1
second. This rendered a total of 139,338 potential click trains (n = 2,249,017 clicks). Only click trains with at least 16 clicks were selected for further work (n = 21,185). The algorithms in this Chapter were written in different versions of MATLAB (Mathworks, MA), from 2016b onwards. Three algorithms were developed for this Chapter, each with a different objective, which are meant to be used sequentially. The first aims to determine whether the clicks grouped in an acoustic event correspond to a click train produced by a harbour porpoise (or another NBHF species). The second aims to separate click trains from undesired sources of sounds. And the third aims to identify general behaviours from the click train pattern.

5.2.2 Click train type

In this first step, the aim is to identify click trains that were produced by harbour porpoises, both of high and low-quality, labelled narrow-band high-frequency (NBHF) and low-quality (LQ) NBHF, respectively. 'NBHF' click trains are those where patterns can be clearly recognised visually and thus have highly likely been produced by harbour porpoises or other NBHF species. In 'LQ-NBHF' click trains patterns sometimes can be seen but they are not always clear due to high levels of high-frequency background noise or multipath signals. Two more types of click trains were defined: 'Sonar' and 'Non-NBHF', the latter having no recognisable pattern for the human observer (Figure 5.1). Henceforth, this algorithm would be referred to as 'Type algorithm'.

The classification of click train types is based mainly on the percentage of variation in the repetition rate (*i.e.*, number of clicks per second - CPS) between consecutive clicks, specifically the median of the variation. This metric was selected as it



Figure 5.1: Click train types. Example of each click train type. NBHF = narrow-band high-frequency. LQ = low quality. The y axis is truncated. Red = high-quality clicks. Blue = low-quality clicks.

was expected to predict best whether there is an underlying pattern. The threshold values were chosen based on data exploration. The algorithm includes several steps and iterations, which also uses information about the length of the click train and how many high-quality (HQ) clicks it has. A schematic of the decision tree is shown in (Figure 5.2).

In environments with high levels of high-frequency background noise silent periods longer than 1 second are rare, and thus a 'click train' would be thousands of clicks long. To avoid this problem, a maximum length of 1,500 clicks was set, in which case the train is divided into smaller click trains with a maximum of 1,500 clicks, and a series of steps are followed to classify them into either of the four categories described above. Click trains that are classified as LQ-NBHF are run again through the 'Type algorithm'. The reason for this step, is that the original click train could have been in fact more than one click train (*e.g.*, it was hidden behind noise) and thus the metrics used to classify click train into types may have changed.



Figure 5.2: Click train types. Decision tree of the algorithm to identify different click train types. $n = \text{length of the click train. } CPS = \text{clicks per second. } CF = \text{centroid frequency. } * A \text{ click train was labelled as 'Sonar' when the mean peak frequency exceeded 140 kHz, the median CPS was between 7 and 8.7, and the mean percentage of change of CF was below 0.5$

The 24th of November 2015 was selected for testing purposes as it contained the highest number of click trains (n = 5,140). Click trains were manually labelled first and subsequently using the algorithm developed in this Section. The results were used to assess the performance of the algorithm against manual labelling.

5.2.3 Extracting patterns

The algorithm to identify patterns (henceforth 'Pattern algorithm') consists of two steps described below:

5.2.3.1 Deleting unwanted sources of sounds

Click trains have different characteristics. Some have high signal-to-noise ratio and few outliers, while others are a mix of HQ and LQ porpoise clicks, and others have high levels of background noise. This means that different approaches are needed to maximise performance. The values used in this section are: clicks per second (CPS), obtained by dividing 1000 by the ICI (in ms); the maximum difference (*i.e.*, jump) in CPS, when sorted in ascending order. It indicates whether there are outliers (*e.g.*, multipath); median percentage of change of CPS; and mean percentage of change of centroid frequency.

At the end of this step there will be a new click train or the click train will go through an additional series of step (next section). The full algorithm is in Appendix C and includes comments in blue.

5.2.3.2 Identifying stable areas and extracting patterns

The second step is to identify areas with low variation in CPS, by estimating the moving average of CPS (n = 5 consecutive clicks). Positions where the variation is under 5% are used as starting points. From these starting points, the algorithm moves backwards and forwards in an iterative process in search of clicks that are consistent in both CPS and amplitude variations. The reference and threshold values are updated in each iteration, as these depend on the click selected in the last iteration.

The full algorithm is in Appendix C and includes comments in blue.

5.2.4 Identifying behaviours

Three main behaviours were defined for the purpose of this study: 'Orientation', 'Foraging', and 'Socialising'. Patterns that did not fit in either of these categories were labelled 'Unknown'. Henceforth, this algorithm would be referred to as 'Behaviour algorithm'.

5.2.4.1 Orientation

Orientation is characterised by having a relatively regular inter-click interval greater than 10 ms. In other words, the porpoise produces under 100 clicks per second, typically between 20 and 60 (Figure 5.3), while it inspects the environment it moves through [e.g., 109, 208].



Figure 5.3: **Orientation**. Spectrogram (Hann window, FFT size = 512) of a typical click train produced by a harbour porpoise while orienting itself or inspecting its environment. Upper panel: amplitude variation in arbitrary units.

Click trains are labelled as 'Orientation', if at least 90% of all clicks have repetition rates below 100 CPS.

5.2.4.2 Foraging

Foraging or feeding events are characterised by a change in clicking rate that follows a specific pattern: a search phase where the porpoise produces fewer than 100 clicks per second, indicating that the animal is not focused on an specific object yet, an approach phase that occurs when a prey is found during which the CPS increases rapidly, and a final stage of high-production rate that can reach up to 640 clicks per second [*e.g.* 9]). The final stage is known as a feeding buzz [123].



Figure 5.4: **Foraging**. Spectrogram (Hann window, FFT size = 512) of a click train produced by a harbour porpoise during a foraging/feeding event. The final phase is known as a feeding buzz, reaching in this case up to 200 clicks per second. Upper panel: amplitude variation in arbitrary units.

The algorithm labels a click train as 'Foraging' when it finds an increase in consecutive clicks from below 100 to over 100 CPS. In other words, the algorithm finds the first click with a repetition rate of over 100 CPS, and estimates the mean CPS for the previous and following 5 clicks.

5.2.4.3 Socialising

Communication or social calls are patterned in many ways, which are different from feeding buzzes and click trains used for orientation purposes. However, there might be overlap between them, for example, the contact call described in Clausen et al. (2010) [12] is similar to the foraging event described in Deruiter et al. (2009) [9]. Social calls lack the low-repetition period that characterise feeding buzzes. An example of a spectrogram of social call is shown in Figure 5.5

The patterns of communication or social calls include V-shaped (Figure 7.15), decreasing CPS, and short and long calls with stable CPS values throughout (see Chapter 6 - Acoustic behaviour). The algorithm labels a click train as 'Socialising'



Figure 5.5: **Burst pulse**. Spectrogram (Hann window, FFT size = 1,024) of a social call produced by a harbour porpoise. The repetition rate ranges between 550 and 650 clicks per second. Upper panel: amplitude variation in arbitrary units.

when all clicks have CPS over 100, or when CPS decreases regularly from over 100. It is expected that this algorithm is not able to identify all communication calls, as knowledge of the different patterns is lacking and there is overlap with foraging buzzes. Additionally, the arbitrary selection of 1 second as a separation time between click trains also means some click trains identified might include more than one echolocation event, hindering the performance of the algorithm.

5.2.4.4 Unknown

Click trains that did not have a pattern found by the previous algorithms are labelled as 'Unknown' (Figure 5.6).



Figure 5.6: **Unknown**. Example of a click train produced by harbour porpoises where the underlying behaviour is not clear. Red = high-quality clicks. Blue = low-quality clicks.

5.3 Results and discussion

A total of 20,516 click trains were found on the 24th of November 2015 (n = 390,528 clicks), of which 5,140 had at least 16 clicks. Of these 1,048 were NBHF, 1,607 were LQ-NBHF, and 3,025 were Non-NBHF. No 'Sonar' click trains were found. The algorithm correctly labelled 93.8% of click trains (n = 4,819). The algorithm was used to label all click trains in both datasets.

5.3.1 Click train type

A total of 139,338 click trains (n = 2,249,017 clicks) were found in both datasets, of which 2,436 were NBHF (see Table 5.2 - NBHF click trains include 8 click trains classified as 'Sonar'. Recordings of sonar were only found in Scottish data (Figure 5.7). It is likely, however, that sonar in Danish waters have different acoustic characteristics (*e.g.*, peak frequency) or repetition rates and therefore the algorithms



would need to be area specific.

Figure 5.7: **Sonar**. Example of a spectrogram (Hann window, FFT size = 1024) of a sonar in Scottish waters, recorded on the 13th of October 2016. Upper panel: amplitude variation in arbitrary units.

The chosen minimum separation time (*i.e.*, 1 second) between consecutive echolocation events, as well as the minimum and maximum click train lengths are arbitrary and thus results should not be interpreted as having biological importance.

Additionally, this means that the number of click trains found here is likely an underestimation. Moreover, an extra 118,153 potential click trains with fewer than 16 clicks were found (Table 5.2). The decision to not consider click trains with under 16 clicks may also impact the ability to identify behaviours (*e.g.*, resting or sleeping), which needs to be taken into account in future work.

Table 5.1: Click trains (CT) per day in data collected in Danish waters using static devices.

			Denmark			
Date	CT	Clicks	NBHF	LQNBHF	NonNBHF	n < 16
12 Aug 2015	7,694	92,763	222	442	530	6,500
13 Aug 2015	6,994	$132,\!046$	113	125	1,778	4,978
24 Aug 2015	6,406	306,929	171	1,520	236	$4,\!479$
25 Aug 2015	$4,\!152$	$579,\!197$	186	1,577	156	2,233
$27 { m Aug} 2015$	$11,\!484$	$259,\!598$	291	1,917	670	$8,\!606$
$26~{\rm Sep}~2015$	12,779	$146,\!856$	44	835	1,563	10,337
$27~{\rm Sep}~2015$	572	$5,\!695$	1	11	84	476
08 Nov 2015	5,772	$17,\!298$	27	30	90	$5,\!625$
09 Nov 2015	$15,\!528$	62,472	56	54	523	$14,\!895$
10 Nov 2015	$14,\!053$	$50,\!860$	195	45	226	$13,\!589$
11 Nov 2015	4,951	$13,\!488$	48	9	32	4,862
$24~\mathrm{Nov}~2015$	20,516	$390,\!528$	1,048	$1,\!607$	3,025	$15,\!376$
$25~\mathrm{Nov}~2015$	$5,\!183$	$95,\!999$	150	312	954	3,767
Total	$116,\!684$	$2,\!153,\!729$	$2,\!552$	$7,\!942$	9,867	95,723

 Table 5.2: Click trains (CT) per day in data collected in Scottish waters using towed

 devices. *NBHF include 8 'Sonar'

			Scotland			
Date	CT	Clicks	NBHF	LQNBHF	NonNBHF	n < 16
28 Aug 2016	167	926	6	2	3	156
29 Aug 2016	389	945	4	1	2	382
30 Aug 2016	800	$1,\!637$	3	0	1	796
31 Aug 2016	101	145	0	0	0	101
$18 { m Sep} \ 2016$	1,072	4,172	36	4	4	1,028
$19 { m Sep} { m 2016}$	435	2,019	18^{*}	0	4	413
$20~{\rm Sep}~2016$	$1,\!153$	4,364	25^{*}	5	1	1,122
01 Oct 2016	455	1,825	15^{*}	3	1	436
02 Oct 2016	944	2,016	8	0	1	935
04 Oct 2016	$1,\!657$	$7,\!423$	47	6	11	1,593
05 Oct 2016	$1,\!679$	4,349	21	3	5	$1,\!650$
06 Oct 2016	1,553	$3,\!649$	16	0	5	1,532
12 Oct 2016	$1,\!433$	$3,\!151$	10	1	2	$1,\!420$
13 Oct 2016	$1,\!607$	$4,\!371$	25	0	5	1,577
23 Oct 2016	1,023	5,504	33	5	11	974
24 Oct 2016	879	$3,\!567$	19	3	7	850
02 Nov 2016	719	3,227	34	1	6	678
03 Nov 2016	$1,\!622$	4,708	24	4	11	1,583
07 Nov 2016	726	$3,\!875$	22	11	8	685
$04 { m Sep} \ 2017$	447	2,228	16	0	2	429
29 Oct 2017	293	1,342	8	2	2	281
01 Nov 2017	596	1,503	4	1	3	588
19 Aug 2019	$3,\!541$	15,797	68	36	84	$3,\!353$
Total	$23,\!291$	$82,\!793$	462	88	179	$22,\!562$

5.3.2 Patterns

Only click trains previously identified as NBHF were used in this section (n = 3,010), of which 2,548 were from Denmark and 462 from Scotland. A large number of clicks were removed from click trains exposing the existing patterns (Figure 5.8).



Figure 5.8: **Patterns**. Original and cleaned versions of a NBHF click train, recorded in Danish waters on the 27th of August 2015. Red = high-quality clicks. Blue = low-quality clicks.

After this step, the number of clicks trains with at least 16 clicks was reduced to 2,656, of which 2,240 were from Denmark and 416 from Scotland. The algorithm failed in some cases, removing clicks that were part of the click train and keeping clicks that were supposed to be removed. The performance of the algorithm was tested by applying the 'Behaviour algorithm' (from next section), before and after the 'Pattern algorithm'. The results are presented in the next section.

5.3.3 Behaviour

A total of 2,656 click trains were used in this Section, of which 416 were from Scotland and 2,240 from Denmark. These had at least 16 clicks after applying the

'Pattern algorithm'. The three main behavioural states defined above were identified in the data when manually and automatically labelled (Figure 5.9).



Figure 5.9: **Behaviours**. Number of click trains of each behaviour found in this study. See text for a description of each. Note: the Orientation bar is truncated.

The algorithm correctly labelled 66.3% (n = 1,759) and 92.3% (n = 2,452) of

click trains before and after the 'Pattern algorithm' was applied, respectively.

Table 5.3: Confusion matrix of classification of behaviours in data collected in Danish and Scottish waters using static and towed devices, respectively.

Behaviour	Orientation	Foraging	Socialising	Unknown	Total
Orientation	2,113	3	0	2	2,118
Foraging	0	66	6	7	79
Socialising	1	71	199	13	284
Unknown	3	66	32	74	175

This means that applying the algorithm to extract patterns improved the ability to classify behaviours by 39.6%.

5.3.3.1 Orientation

A total of 2,118 click trains were labelled as 'Orientation', with repetition rates below 100 clicks per second throughout the click train. This represents 79.7% of all click trains. These had a mean length of 41.5 (SD = \pm 42.8) clicks, ranging from 16 to 637. The algorithm correctly classified 99.9% of click trains (n = 2,113)



Figure 5.10: **Orientation**. Typical click train produced by a harbour porpoise while moving through its environment. Red = high-quality clicks. Blue = low-quality clicks

Within these click trains, two patterns were found: landmarks and potentially resting behaviour. Landmarks are characterised by increasing repetition rates from 20 to 60 CPS, which indicates that the porpoise is using fixed objects in the environment to orient themselves [105]. Porpoise acoustic behaviour during rest or sleeping is not well understood, but evidence suggest they remain silent or click at low repetition rates [107]. See Chapter 6 - Acoustic behaviour.

5.3.3.2 Foraging

A total of 79 (3%) click trains were clear 'Foraging' events, showing a gradual increase of repetition rates, reaching up to ~600 CPS. These values are consistent with what has been found in porpoises in captivity and in the wild, using both static recorders as well as acoustic tags [e.g., 8, 9, 13]. Foraging events had a mean of 137.7 clicks (SD = ± 149.6), ranging from 18 to 970. The algorithm correctly classified 83.5% (n = 66) of click trains.



Figure 5.11: **Feeding**. Example of a click train produced by a harbour porpoise during a foraging/feeding event. Red = high-quality clicks. Blue = low-quality clicks

An example of a typical foraging event is shown in Figure 5.11. Note that the amplitude of individual clicks decreases as the repetition rate increases. Although in some cases this may indicate that the animal is moving away from the acoustic recorder, it is consistent with what have been observed in tag data [e.g., 108].

5.3.3.3 Socialising

A total of 284 click trains (10.7%) contained patterns that were consistent with communication calls. Different patters were found, some of which have been described in the literature before and some that have not. Social calls are described in detail in Chapter 6 - Acoustic behaviour. Some of these click trains may represent the end of foraging events, which were not detected in their entirety, either because the animal was vocalising in a direction away from the hydrophone or because the classifier did not identify the other clicks. This would explain why there are no preceding periods with high inter-click intervals, however, for many of these, there is a clear leap from low repetition rates to high, without the intermediate values as



seen in Figure 5.11.

Figure 5.12: **Socialising**. Example of a social call produced by a click train produced by a harbour porpoise. Red = high-quality clicks. Blue = low-quality clicks

Additionally, regular clicks are usually of higher amplitude and so if the final buzz is detected, it is expected that the previous clicks would be detected too, especially due to the short duration of these click trains. In the literature, a threshold of 10 ms is used to classify a given click train as a foraging event [*e.g.* 9, 247, 248]. These results suggest that using this threshold in all cases might over estimate foraging events, as most of the social calls found here and in other studies have repetition rates over 100 CPS [10, 11, 13].

The overall performance of the algorithm was 70.1% (n = 199). The algorithm performed poorly distinguishing between foraging and social calls with 25% (n = 71) of social calls classified as 'Foraging' and 4.6% (n = 13) as 'Unknown'. Similarly, 7.6% of 'Foraging' were classified as 'Socialising' and 8.9% (n = 7) as 'Unknown'. This result was expected due to the overlap in repetition rates between these behaviours.

5.3.3.4 Unknown

Click trains that did not fit in any of the patterns were labelled as 'Unknown', which comprised 6.6% (n = 175) of all click trains. Some of these may represent partial click trains, due to the porpoise moving away from the recorder or due to mistakes made by the 'Pattern algorithm'. In other cases, the click train may be complete but the pattern has not been reported before in the literature (see Chapter 6 -Acoustic behaviour).



Figure 5.13: **Unknown**. Example of a click train produced by harbour porpoises where the underlying behaviour is not clear. Red = high-quality clicks. Blue = low-quality clicks.

Moreover, some patterns may be the result of an unsuccessful foraging event. The pattern in Figure 5.13, for example, could be formed by a porpoise who missed its prey in the first attempt and continued to chase it in the water column using echolocation.

5.4 Limitations

The algorithms developed and tested in this Chapter showed high levels of accuracy. However, because the values chosen here for minimum separation time and maximum length were arbitrary, it means that some click trains would be missed. For example, the 'Type algorithm' assumes that when a click train is longer than 1,500 clicks, this is due to high background noise levels and thus LQ clicks are discarded. This decision could result in removing NBHF click trains (Figure 5.14).



Figure 5.14: *Limitations*. Example of a long 'click train' (n = 6,341) before the 'Pattern algorithm' is applied. Red = high-quality clicks. Blue = low-quality clicks.

This outcome is a direct consequence of the decision-making pathway of the algorithms and means it can be improved. It also means that for monitoring purposes, click trains can be lost and potentially the absence of porpoises is a false negative. To explore this matter, 'positive porpoise minutes' (PPM) were estimated before and after the 'Pattern algorithm' was applied in all days in the Danish data, which was collected with static recording devices (Figure 5.15 and Figure 5.16).



Figure 5.15: **Positive porpoise minutes**. Values obtained before the 'Pattern algorithm' was applied.

The results show that the time periods with PPM remain similar before and after the 'Pattern algorithm' was applied and that it although removes $\sim 12.1\%$ of click trains (n = 308), representing $\sim 9.9\%$ of PPM, the overall pattern does not change (Figure 5.15 and Figure 5.16).



Figure 5.16: **Positive porpoise minutes**. Values obtained after the 'Pattern algorithm' was applied.

5.5 Summary

In order to use passive acoustic monitoring devices (*e.g.*, SoundTraps - Ocean Instruments, New Zealand) to study the behaviour of harbour porpoises in the wild, a series of analytical tools are necessary. Several algorithms were developed and tested in this Chapter, which are meant to use sequentially, to identify click trains produced by harbour porpoises, extract patterns, and identify underlying behaviours. The performance ranged from 69.2% to 92.7% accuracy.

The method developed here to extract underlying patterns (*i.e.*, the 'Pattern algorithm') was based on a algorithm developed by Lepper et al. (2005) who used pre-defined thresholds of inter-click interval and amplitude variations to separate overlapping click trains [233]. The method was not directly applicable for harbour porpoises and instead an adaptable threshold was implemented. Although the 'Pattern algorithm' did not provide fully cleaned click trains, it improved the per-

formance of the 'Behaviour algorithm' by 39.6%.

The 'Behaviour algorithm' performed relatively well, and several patterns were found both in click trains classified as 'Orientation' as well as in 'Socialising'. These were consistent with social calls, found here for the first time in data collected using static acoustic devices that record continuously.

The number of clicks per click trains was varied, but because harbour porpoises click almost continuously, the length likely provides little information about their natural behaviour. Moreover, length depends on how a click train is defined, as well as the spatial orientation of the animal, which may move its head away from the recording device or be too far to be detected.

The results from this Chapter show these algorithms can potentially be used for monitoring programmes, such as the Static Acoustic Monitoring of the Baltic Sea Harbour Porpoise, known as the SAMBAH Project.

Chapter 6

Acoustic behaviour

6.1 Introduction

Even though large groups of a few dozens to hundreds of harbour porpoises have been observed, porpoises are usually seen alone or in groups of two or three individuals [*e.g.*, Nairn, unpublished data and 2, 3, 17, 79, 114]. Our understanding of their social behaviour and communication is limited to visual observations in captivity and data obtained from wild animals carrying acoustic tags [9, 10, 11, 12, 13]. The few existing descriptive studies were conducted in captive settings due to the unique opportunities it provides for behavioural observations [9, 10, 11, 12].

These studies showed harbour porpoises emit clicks at different repetition rates, and that at least some of the patterns in repetition rates can be linked to specific behaviours. However, the name and characteristics of the calls is not always consistent between studies. For example, 'distress calls' were reported in several studies, but the characteristics of the calls differed between them. Busnel and Dziedzic (1966) described these as calls with repetition rates of between 130 and 250 clicks per second (CPS) and duration between 0.4 and 1.55 s [10], while Amundin (1991) described distress calls based on the HF component as calls with repetition rates of between 100 and 500 CPS, but with repetition rates between 300 and 400 CPS based on the LF component only [11]. Other social calls include 'contact call' between mother and calf, 'aggression', 'courtship', and 'dominance' calls.

Social calls have rarely been reported in the wild. Koschinski et al (2008) found a series of calls emitted by a calf that was accidentally caught in a fishing net in Canadian waters, which were recorded with a T-POD. In this case, the authors knew in advance where to look for those calls [8]. The only study to date which specifically looked at repetition rates of wild harbour porpoises clicks is Sorensen et al. (2018). A total of six porpoises, accidentally caught in pound nets in Danish waters, were mounted with acoustic and behavioural tags and released into the wild [13]. This study confirmed what other studies found: harbour porpoises emit calls which are qualitatively different from feeding buzzes. The distribution of repetition rates of social calls found by Sorensen et al. was bimodal with peak around 125 CPS and 1,200 CPS [13].

Sorensen et al. (2018) also provided unique insights into the social lives of harbour porpoises. The authors found they spend much of their time (between 9.9% and 58.8% of all one-minute intervals) at hearing distance from other porpoises, as the longer period without calls from other porpoises was 180 minutes. The production rate of social calls for all individuals was between 0.2 and 1 call per minute [13], remarkable evidence that porpoises are social animals.

The aim of this Chapter is to describe vocalisation patterns found in data from wild harbour porpoises collected using passive acoustic monitoring systems. The data used in this Chapter were the click trains identified as 'Orientation' (n = 2,118)and 'Socialising' (n = 284") from Chapter 5 - Patterns of click trains. High-quality and low-quality harbour porpoise clicks are shown in red and blue, respectively, throughout this Chapter.

6.2 Orientation

Click trains labelled 'Orientation' are characterised by having repetition rates below 100 CPS. These indicate porpoises are inspecting the environment they move through. Within these click trains, additional patterns were identified, which are described below.

6.2.1 Landmarks

In a series of experiments conducted in captivity, Verfuss et al. (2005) found that harbour porpoises use objects in their environment to orient themselves, even during day time and despite having been in the enclosure for several years. During these experiments, porpoises produce click trains with repetition rate that increased from ~ 20 to ~ 60 CPS, at which point it dropped. The authors called this pattern 'landmarks' [105].

This pattern was found in 0.9% of click trains (n = 103) labelled as 'Orientation'. In most cases the repetition rate increased from ~ 20 to ~ 60 CPS, although in some cases it reached 100 CPS before it dropped.

CHAPTER 6: ACOUSTIC BEHAVIOUR



Figure 6.1: *Landmark*. Example of a 'landmark' recorded in Danish waters using a static passive acoustic device.

6.2.2 Sleeping or resting?

Little is known about the sleeping habits of harbour porpoises in their natural environment. Wright et al. (2017) analysed acoustic and behavioural data from six tagged porpoises in Danish waters and found what seems like sleeping behaviour. These periods were characterised by shallow, parabolic-shaped dives of short duration (under 1 min). Further, 43% of parabolic dives were silent (*i.e.*, fewer than 10 clicks per dive) and during these dives porpoises were less likely to roll than during other dive types. Additionally, when clicking, they produced a maximum of ~ 23 CPS [107].

At least 16 click trains (0.07%) classified as 'Orientation' had very low clicking rates (~5 CPS), which could indicate resting or sleeping periods (Figure 6.2). It is possible that the low clicking rate observed in some of the data is the result of erroneous click classification by the PorCC algorithm or removal of clicks by



Figure 6.2: *Sleep?*. Click train produced by a harbour porpoise potentially representing resting or sleeping behaviour.

any of the algorithms used in this study. Low repetition rates can also be sonar misidentified as NBHF click trains (see Figure 5.7). However, at least in cases when verification was possible, these were real harbour porpoise clicks and the only ones detected in the recordings (Figure 6.2).

6.2.3 Other

Unlike the two patterns described above, some patterns with repetition rates below 100 CPS were not described before in the literature. Two examples are shown in



Figure 6.3 and Figure 6.4

Figure 6.3: Unknown. Click train produced by a harbour porpoise.



Figure 6.4: V-shaped?. Click train produced by a harbour porpoise.

It is difficult to hypothesise what these patterns may represent. The pattern in Figure 6.3, for example, may be simple inspection of objects at sea, something floating the animal echolocates to. The pattern in Figure 6.4, on the other hand, is similar to a series of calls detected both in Scottish and Danish waters (see subsection 6.3.1).

6.3 Social calls

This Section presents a description of the types of potential social calls found in this study. Some of these were similar to calls described in the literature, both from the repetition rate of the low and the high-frequency components of the porpoise clicks [10, 11, 12].

6.3.1 V-shaped

V-shaped calls are, as the name indicates, calls that look like a V. They begin at about 300 CPS, go down to fewer than 150 CPS and back to 300. These call types were detected in Scottish and Danish waters. In Scotland, it was detected during an interaction with a solitary short-beaked common dolphin (see Chapter 7 - Interspecies communication), and it was detected three times in a short period of time (two are shown in Figure 6.5).



Figure 6.5: V-shape. Example of a potential social call produced by harbour porpoises in Scottish waters.

CHAPTER 6: ACOUSTIC BEHAVIOUR

In Danish waters, the V-shaped call was detected at the end of a five-second period that included what are potentially also social calls (Figure 6.6).



Figure 6.6: V-shape. Example of a potential social call produced by harbour porpoises in Danish waters.

It is encouraging that the patterns in Figure 6.5 and Figure 6.6 are so similar, as they have been recorded in two populations hundreds of km apart. This pattern has not been described in the literature before.

6.3.2 Downsweep

There are at least three types of click trains that have a downsweep pattern. A first type begins at repetition rates around 600-700 CPS and decreases to 400 CPS. A second type begins at repetition rates around 300-400 CPS and decrease to 100 CPS. Lastly, a third type begins at 200-250 CPS and decrease steadily to 50 CPS or less (Figure 6.7).

It is possible these are the same type of call and that the last two types are simply incomplete click trains. However, are least in some of them, the repetition



Figure 6.7: **Downsweep**. Examples of types of communication call produced by harbour porpoises with a downward pattern, with different repetition rates.

rate increases slightly at the beginning, which means they either begin at 600-700 CPS or at 200-250 CPS.

6.3.3 Upsweep

These calls are characterised by an increasing repetition rate (Figure 6.8). The patterns in Type 1 resemble 'aggressive' calls described by Clausen et al. 2010 [12], while the patterns in Type 2 resemble 'distress' calls of short duration reported by Amundin (1991) [11].

Repetition rates during feeding events overlap with the click trains represented



Figure 6.8: **Upsweep**. Examples of types of communication call produced by harbour porpoises with a upward pattern in repetition rates.

here, limiting the ability to distinguish between them. Moreover, Clausen et al. (2010) described 'contact' and 'grooming' calls which are very similar to feeding events [12].

6.3.4 Short

This was the most common type of call in the datasets. They are characterised by a stable repetition rate between 100 and 300 CPS. These resemble calls described by Amundin (1991) and Busnel (1966) as 'distress calls' [10, 11].



Figure 6.9: Short calls. Examples of short calls produced by harbour porpoises.

6.3.5 Long calls

These calls had low, relatively constant repetition rates, between 100 and 250 CPS, and duration around 1 second (Figure 6.10). Amundin described these as long distress calls



Figure 6.10: Long calls. Examples of long calls produced by harbour porpoises.

Busnel and Dziedzic (1966) described 'courtship calls' as long calls with repetition rates around 500 CPS, from the low-frequency component of the porpoise click. One call that resemble this description is shown in Figure 6.11. This call was deleted by the 'Pattern algorithm' as it was comprised exclusively by low-quality clicks (see Chapter 5 - Patterns of click trains).

Identifying these calls would provide crucial information about the natural be-



Figure 6.11: Courtship?. Social call produced by harbour porpoises.

haviour of harbour porpoises and would help identify important areas for them that require protection.

6.3.6 Phrases

Amundin (1991) used the word 'phrase' to describe calls that comprised at least three calls produced in a short period of time. These include the same or different call types [11]. At least three phrases were found in this study. The first type was described in the 'V-shaped' section, found in data collected in Scotland. The second type is shown in (Figure 6.12), which comprised six nearly identical calls repeated in just below 8 seconds, also found in Scottish data.

The third comprised short calls with repetition rates around 120 CPS (Figure 6.13).

CHAPTER 6: ACOUSTIC BEHAVIOUR



Figure 6.12: **Phrase**. Example of a 'phrase', a series social calls produced by harbour porpoises [11]. Top: amplitude. Middle: Clicks per second. Bottom: time domain and spectrogram. Recordings made on the 13th of October 2016 in the Firth of Clyde (Scotland).

Other potential phrases were found in the data, comprising different call types. Detecting phrases in data from passive acoustic devices would depend on many factors, including selecting a suitable minimum time separation between consecutive click trains. If the separation is too short, then click trains that are not related may be pushed together, while if the separation is too long, calls that make up a phrase may be separated.



Figure 6.13: **Phrase**. Example of a 'phrase', a series social calls produced by harbour porpoises [11].

6.3.7 Phrase?

There were other calls that did not belong to any of the types described before. It is likely that these click trains were incomplete due to the animal moving away from the hydrophone or because they represent another type that is not common or were not recorded often enough.



Figure 6.14: *Phrase?*. Example of a click train produced by harbour porpoises.

Others, however, showed clear patterns but at very low repetition rates, below 100 CPS, such as the one shown in Figure 6.14. Note this pattern was recorded over a period of six seconds. A similar pattern was found where repetition rates reached 400 CPS in data from Denmark.

6.4 Overlaps

Although rare in the datasets, there were several click trains in which it was possible to see there were more than one harbour porpoise vocalising at the same time. One of the clearest cases is shown in Figure 6.15, where at least two (and possibly three) animals are emitting sounds simultaneously.



Figure 6.15: **Overlap**. Example of multiple harbour porpoises vocalising simultaneously.

These were rare, but the algorithms developed here would have a difficult time dealing with cases like these, despite being correctly classified as NBHF click trains.
6.5 Summary and implications

This chapter presented a detailed description of different patterns of repetition rates of harbour porpoise clicks. Many of these are consistent with patterns described in the existing literature, which seem to correspond to specific behaviours. For example, patterns consistent with using fixed objects in the environment to orient themselves (*i.e.*, landmarks) [105] and social calls consistent with 'distress' calls [11].

Unfortunately, the database from Denmark did not include the recordings and therefore it was not possible to verify click trains, especially incomplete ones and those with patterns that did not fit in any of the known categories. This Chapter shows the algorithms developed in this thesis can be used to study the behaviour of harbour porpoises in their natural environment using acoustic recorders, including static as well as towed ones. Although visual observations are necessary to verify the patterns described here correspond to social calls, at least some of the patterns have been described in the literature before. This is the first time social calls are found in data collected in the wild using continuous recordings. Although it is likely that some foraging events were misidentified as social calls (and vice versa) there were more social calls than foraging events detected in both datasets (34 vs 29 in Scotland and 251 vs 50 in Denmark). The data is not directly comparable to the tag data from Sorensen et al. (2018) as they were collected and analysed using different methodologies. However, no social calls were found with repetition rates over 800 CPS. This may be due to how impulsive detectors work, which could result in over or under estimation of inter-click interval values (and therefore CPS), as well as misidentification of clicks as noise. Algorithms exist to identify buzzes which could

CHAPTER 6: ACOUSTIC BEHAVIOUR

be adapted for behavioural studies.

This study constitutes a proof of concept and it shows that simple algorithms could be used for behavioural studies. Moreover, new analytical tools, such as machine or deep learning, could be easily implemented to identify patterns in a most effective manner once patterns have been described and verified.

Chapter 7

Interspecies communication

7.1 Introduction

Kylie (Figure 7.1) is a solitary female short-beaked common dolphin (*Delphinus delphis*) who has inhabited the Firth of Clyde, in the west of Scotland, since the early 2000s. In 2017, Ryan reported that Kylie was seen with a harbour porpoise in two different locations within the Firth of Clyde, four years apart [142]. While observations of both species engaging in interspecific interactions are commonly observed in areas where they overlap with other species [140, 174, 249], Kylie's association with harbour porpoises is unique and rare on several levels. Firstly, there are only 7 reported cases of common dolphins leading a solitary life on a permanent basis [175, 237]. Secondly, common dolphins in the Firth of Clyde are rarely seen (Nairn, unpublished data), while the harbour porpoise is a resident species to the inner Firth of Clyde, where several hotspots have been identified [90].

Thirdly, their social behaviour is also unalike. Common dolphins live and travel



Figure 7.1: Kylie. Solitary short-beaked common dolphin in the Firth of Clyde, West Scotland.

in large groups that regularly exceed 50 individuals but can reach thousands of animals [165], while harbour porpoises are usually seen in groups of two or three animals, or as solitary individuals [79], rarely congregating in large groups [*e.g.*, Nairn, unpublished data 1, 3, 114]. Lastly, these two species are also acoustically quite different. Unlike harbour porpoises, common dolphins produce a wide range of sonic and ultrasonic sounds that they use for foraging and communication purposes (Figure 7.2). These include whistles (*i.e.*, frequency modulated, long duration, tonal calls), buzzes, barks, yelps, squeals, and other burst-pulsed calls (*i.e.*, a series of rapidly produced clicks that occur both in echolocation and communication) [165, 166].

This Chapter presents the results of observations of Kylie accompanied by a harbour porpoise since 2009 to date, as well as the acoustic behaviour of Kylie and a harbour porpoise when travelling together, from recordings made in three different



Figure 7.2: Whistles and clicks. Example of whistles (horizontal lines) and clicks (vertical lines) produced by short-beaked common dolphins in the Celtic Deep - extracted from Scullion (2004) [250].

days in 2016 and 2017. These recordings provided an exceptional and unprecedented opportunity to study their acoustic behaviour as a case study of interspecific vocal communication in wild cetaceans.

7.2 Methods

7.2.1 Data collection

Acoustic data were collected during systematic and opportunistic boat-based surveys in the Firth of Clyde (Figure 3.1), from June 2016 to July 2018. These surveys were conducted from the 'Saorsa', a 40-foot-long sailing vessel, and recordings were made using a towed hydrophone array (See chapter 3 for more details). Additionally, and using a SoundTrap ST300 (Ocean Instruments, New Zealand), Nairn (unpublished data) was able to record Kylie's sounds for a period of 7 days in the summer of 2019. The recorder (sampling frequency = 576 kHz) was positioned under one of the navigational buoys in the Hunterston/Fairlie channel where she spends much of her time. Nairn discovered that Kylie produces a wide range of sounds, including whistles and burst pulsed of low frequency range (< 20 kHz) much of the time. These data was not used for this project, but the recordings were key to interpreting the implications of the results of this study.

7.2.2 Data extraction and analysis

Recordings were run through the Click Detector Module built-in in PAMGuard [221], which detects impulsive sounds and saves them as individual clips. All potential dolphin and porpoise clicks were selected based on direction of arrival, observed inter-click intervals (ICI), the characteristic of the waveform, power spectrum, and Wigner plot, all of which are also available in PAMGuard's Click Detector display. They were selected manually and extracted using custom-built algorithms written in MATLAB 2017a (Mathworks, MA), together with date and time, and direction of arrival estimated using trigonometric methods within PAMGuard's algorithms [224]. For all extracted clicks additional variables were estimated, including amplitude, duration (estimated as the 80% of the energy of the clip), centroid frequency (kHz), root mean square bandwidth (BW_{RMS} , kHz) [238], and ICI, estimated as the time difference between the beginning (*i.e.*, sample number in the recording) of the click and the previous one (ms) (See Chapter 3 for a detailed description of each of the variables).

Once extracted, all clicks were individually checked and assigned to either the porpoises or Kylie (*i.e.*, non-harbour porpoise clicks). The discrimination was based on the unique characteristics of narrow-band high-frequency (NBHF) clicks produced by harbour porpoises (Figure 7.3). The direction of arrival of the click, amplitude, the inter-click interval, and the spectrogram of the original recording were used to avoid assigning echoes or other sounds to Kylie. The decision to focus on individual clicks instead of click trains, was made because there were cases where click trains slightly overlapped (*e.g.*, Figure 7.20). Moreover, although there are available classifiers for porpoises and dolphins, the discrimination was made manually as the performance of existing dolphin click classifiers is either low or unknown [170]. Moreover, using a classifier would overlook changes in Kylie's clicks, as classifiers are trained to identify specific temporal and spectral characteristics of the clicks [172].



Figure 7.3: Porpoise click. Typical harbour porpoise click from a sample saved by PAMGuard Click Detector Module (bottom), and its power spectrum (right) and Wigner plot (top).

In order to validate the click discrimination method described above, a total of eight volunteers were trained to identify NBHF clicks. The volunteers were given written instructions (in English or Spanish) for training and testing purposes. In or-

der to avoid bias, volunteers were not given information about what the study was about. For training purposes, the volunteers were provided with 50 high-quality harbour porpoise clicks. These were selected from data collected in Danish waters during the summer of 2015 using a SoundTrap 300 (Ocean Instruments, New Zealand), and 250 clicks (20%) from the three days of recordings of Kylie and a porpoise, which were selected randomly using a random number generator. The signals were provided as black and white images consisting of the waveform, frequency spectrum, and Wigner plot, without axes. The reliability of agreement between Cosentino's and the volunteers' discrimination was estimated using Fleiss' Kappa [251] with a MATLAB function developed by Cardillo [252]. Spectrograms of the recordings were inspected using Audacity version 2.3.0, for verification purposes and in search of whistles and other tonal and pulsed calls Kylie may have produced.

Once echoes and other unwanted sounds were removed from the database, clicks were plotted in time against their direction of arrival (*i.e.*, bearing), colour coded by species, in order to identify periods during which the dolphin and the porpoise were travelling next to each other (Figure 7.16). This was indicated by the simultaneous changes in bearing of consecutive clicks, regardless of which animal produced them [238]. In other words, if clicks were coming from the same direction at almost the same time it was assumed that the animals were travelling close to each other. If, on the other hand, the direction was different, it was assumed that the animals were not travelling in close proximity, although remained in the same area. Clicks produced when the animals were close together (n = 1,248) were separated for further analysis.

7.2.3 High frequency clicks

High-frequency clicks were defined, for the purpose of this study, as clicks with peak and centroid frequencies above 100 kHz, as this is the lower limit of energy content in harbour porpoise clicks [7]. The density distributions of high-frequency clicks produced by the dolphin were estimated using a bootstrap method, with 5,000 iterations for four scenarios:

1) when Kylie was seen alone (n = 419) - these are periods where no porpoise was seen around Kylie. It was not possible, however, to confirm that no porpoises were within hearing distance from Kylie.

2) periods when she was seen with a porpoise (n = 2,023) - these periods include both when travelling next to a porpoise (3) as well as when the porpoise was visible in the vicinity (4)

3) when seen with a porpoise but not travelling next to it - these are periods when the porpoise and Kylie are interacting, yet not next to each other (n = 775), and 4) when travelling in close proximity to the porpoise (n = 1,248) - this include only periods where the animals were assumed to be close enough so received signals did not vary in direction of arrival.

Subsequently, a Tukey's honestly significant difference (HSD) test was performed to test whether the proportion of high frequency clicks was significantly different between scenarios.

7.2.4 Click trains

A click train can be loosely defined as a 'series of clicks separated by gradually or cyclically changing inter-click intervals suggesting a unit during an echolocation

event or a communication signal' [8]. Two types of click trains were defined to study the vocalisation patterns of Kylie and the porpoise when travelling together. The first type, was defined based on the ICI only, where an ICI of one second or more separated two consecutive click trains, regardless of whether it contained clicks produced by both or one animal. The second type was defined by species and ICI, in other words, whenever a click was produced by the other species, or if there was a gap of one second or more, a new click train would begin, even if it contained only 1 click. Subsequently, and in order to investigate whether the order in which Kylie and the porpoise vocalised when travelling next to each other (n = 1,248) was random, the non-parametric Wald–Wolfowitz Runs Test was performed on both types of click trains. The runs test reveals whether the observed pattern is the result of random vocalisation or not.

Additionally, in order to estimate the transition probabilities between porpoise and Kylie clicks, and because Kylie produced two types of clicks (low- and high-frequency clicks), three states were defined: low-frequency clicks (LF), high-frequency clicks (HF), and harbour porpoise clicks (P). These states were used to estimate the transition probabilities in two scenarios: travelling close to each other and together in the same area but not travelling next to each other. The transition probabilities between states, including self-transitions, were compared using the Markov chain method [253].

7.3 Results and discussion

Harbour porpoises were detected throughout the Firth of Clyde on all survey days. A detailed analysis of the distribution and density of harbour porpoises in the Firth

of Clyde using these data is presented in Brown (2018) [90]. No common dolphins were observed, except for Kylie, who, in all cases, was seen swimming around the same navigational buoy located off Cumbrae in the Fairlie/Hunterston Channel Figure 3.1. Kylie was seen alone on the 4th of October, and 7th and 23rd of November 2016 as well as the 1st of November 2017 and the 19th of July 2018, when she was also acoustically detected. She was seen with a harbour porpoise on the 2nd and 3rd of November 2016 (day 1 and 2), and the 4th of September 2017 (day 3). These encounters lasted less than 5 min on day 1, just over 6 min on day 2, and 12 min on day 3, as the vessel discontinued a systematic survey and abandoned its transect line so as to better observe the animals. A fourth encounter with Kylie and a porpoise took place on 7 June 2018, although no acoustic recordings were made. Photographs of the porpoise from the observations in 2017 and 2018 were compared to those in Ryan et al. (2017) [142] and with photographs from 2004, 2009, and 2015 (Figure 7.4).

How or why Kylie became a solitary individual is, and will remain, a mystery. Female short-beaked common dolphins live up to 30 years of age and become sexually mature at 6-8 years of age [23]. Eye witnesses reported that Kylie had a calf when she was first seen in the area, which means she was already an adult when she became solitary. However, she may have been with a porpoise, which are much smaller than common dolphins [23] and could have been misidentified as a calf. Regardless, she is now an adult and has spent almost 20 years in the area on her own. It is likely she has not interacted with conspecifics during this time while she has been exposed to harbour porpoise clicks regularly, as the inner Firth of Clyde is an area of high harbour porpoise density year round [90].

Kylie spends much of her time around navigational buoys, especially three buoys

located in the Fairlie/Hunterston channel (Figure 3.1), where locals see her regularly and year-round. She has never been seen interacting with other common dolphins. In 2017, Ryan [142] reported that Kylie was seen with the same individual porpoise in two different locations within the Firth of Clyde, 4 years apart. One of the sightings (June 2011) took place around a navigational buoy in the Fairlie/Hunterston channel and the other one (August 2015) in Loch Fyne. The authors were lucky to obtain good quality images from both encounters and thus were able to individually identify the harbour porpoise [142]. In this study, and using opportunistic data collected since 2004 by several parties, including the Clyde Porpoise CIC and their volunteers, as well as by third parties, it was possible to compare photographs with those reported by Ryan [142]. The data showed that not only Kylie interacts with harbour porpoise more often than previously thought, but also that she interacts with different individual porpoises, although with one at a time (Figure 7.4).

When Kylie is alone, her behaviour is almost stereotyped. She surfaces for a few seconds (to breathe) approximately once per minute, heading in the direction away from the buoy, at a distance of two to three metres from it (Figure 7.5). This behaviour repeats for hours on end until she moves to other areas (*e.g.*, to another buoy or to feed) or to approach passing vessels.

Kylie is regularly visited by locals and tourists in recreational vessels, especially during the summer season. Many visitors try to tease her to follow them or engage in some aerial display (e.g., breaching), by passing close to her at a high speed. Vessels types include sailing boats, zodiacs, and jet skis (Figure 7.6). She occasionally engages and follows the vessels for a few hundred metres, but then quickly returns to the buoy. She has also been seen approaching passing vessels on her own, including



Figure 7.4: Travelling together. Solitary short-beaked common dolphin, known as Kylie, travelling together with a harbour porpoise (Top: photo by D. Nairn, 2018. Bottom left: photo by P. Nichols, 2017. Bottom right: photo by G. Patterson, 2009). All pictures were taken off Cumbrae, in the Fairlie/Hunterston Channel, in the Firth of Clyde, West Scotland.

recreational boats as well as the research vessel used in this study (Figure 7.6).



Figure 7.5: Kylie. Solitary short-beaked common dolphin next to navigational buoy off Cumbrae, in the Fairlie/Hunterston Channel, in the Firth of Clyde, West Scotland.



Figure 7.6: Recreational vessels. Kylie is regularly visited by recreational vessels

7.3.1 Kylie's sounds

Three of the volunteers who participated in this project to validate the methodology to assign clicks to either the porpoise or the dolphin (see Methods above) either over or under-assigned signals to the NBHF class. Of the 250 signals used for testing purposes, a total of 31 signals were assigned as harbour porpoise clicks (12.5% of the dataset) by MC, while these volunteers found 114, 117, and 8 respectively. The null hypothesis of the Fleiss' Kappa test (*i.e.*, the level of agreement is accidental) was rejected both when including (p = 0, k = 0.28, level of agreement 'Fair') and excluding (p = 0, k = 0.6, level of agreement 'Substantial') the results of the abovementioned volunteers. It was therefore assumed that the method used to assign clicks to either the porpoise or the dolphin was adequate.

A total of 2,023 clicks were recorded during the encounters with the dolphin-porpoise pair. Of these, 1,499 were produced by Kylie and 524 by the porpoise, and were either single clicks or part of click trains. Additionally, 419 clicks were recorded when the dolphin was seen alone in November 2017 (n = 72) and July 2018 (n = 347). These clicks were mostly of two types: low-frequency (< 50 kHz) and high-frequency clicks (> 100 kHz), and although clicks with intermediate centroid frequency were detected, these were the minority of clicks, and were produced as a transition state between low and high-frequency clicks (Figure 7.7). The centroid frequency ranged from 23.4 kHz to 187 kHz, with low and high-frequency clicks being narrow-band and clicks with intermediate centroid frequency being broadband (Figure 7.8).

Despite the fact that Kylie is known to produce a wide range of sounds (Nairn, unpublished data), she was recorded producing only two types: a burst pulsed call (n = 1) and echolocation clicks (n = 1,786). No other tonal sounds that are typically produced by common dolphins while socialising, such as whistles, were recorded. Given Kylie is capable of producing typical common dolphin sounds, it is possible that whistles and other tonal sounds were not recorded due to the short duration of the encounters. However, this apparent preference for producing clicks must be interpreted in the context in which they were recorded.

The proportion of high-frequency clicks produced by Kylie was a significantly



Figure 7.7: Centroid Frequency. Histogram of the centroid frequency of clicks produced by the porpoise (orange) and by Kylie (light blue) when travelling together.

higher when she was accompanied by a porpoise than when she was seen alone. When she was on her own, the proportion was 0.21 (SD = \pm 0.02), which increased to 0.32 (SD = \pm 0.01) when seen around a porpoise. When, on the other hand, the porpoise was around but not next to Kylie, she produced a proportion of highfrequency clicks of only 0.03 (SD = \pm 0.01), which significantly increased to 0.44 (SD = \pm 0.02) when travelling next to it. This is shown graphically in Figure 7.9 and numerically in Table 7.1.

The null hypothesis (H_0) of the Tukey's honestly significant difference (HSD) test was that these proportions were the same in all scenarios. The null hypothesis was rejected and all comparisons between proportions of high frequency clicks were significantly different ($\alpha = 0.05$).



Figure 7.8: Bandwidths. Variation of the bandwidth of clicks produced by Kylie and the porpoise in different scenarios. The bandwidth was plotted against the centroid frequency. Note that low and high frequency clicks are narrow-band (low y values), while mid-frequency clicks are broadband (high y values).

The fact that Kylie produces only clicks and a higher proportion of high-frequency ones when travelling with porpoises, especially when in close proximity to them, suggests that she is actively using clicks instead of other communication sounds. The maximum hearing sensitivity for harbour porpoises is around 125 kHz, however, their hearing abilities extend to both higher (>150 kHz) and lower frequencies (<10 kHz), with a decreased sensitivity at 63 kHz [254]. Curiously, common dolphins' highest sensitivity is around 64 kHz, decreasing sharply in frequencies beyond 100 kHz [255]. It is expected, therefore, that at short distances they are capable of hearing each other, even without the dolphin having to produce high-frequency clicks, suggesting there is another reason why she prefers them.

The variability of clicks produced by Kylie was high, both in the temporal and



Figure 7.9: High frequency clicks. Histograms and density distribution of the proportion of high frequency clicks produced by a short-beaked common dolphin in four scenarios. 1) Dolphin alone (n = 419). 2) Dolphin nearby a harbour porpoise (n = 2,024). 3) Dolphin and porpoise in the same area, but not travelling next to each other (n = 775). 4) Dolphin and porpoise travelling next to each other (n = 1,248).

spectral domains. The typical low-frequency click had a centroid frequency at 37 kHz, with most of the energy concentrated around it (Figure 7.11). Some of the high-frequency clicks were oligocyclic, similar to low-frequency clicks and high-frequency clicks reported for other dolphin species, such as the Dusky dolphin (*Lagenorhynchus albirostris*) [33]. However, some of Kylie's high-frequency clicks seem to have most of the energy between 100 and 150 kHz than the broadband clicks produced by other dolphin species (Figure 7.10).

Moreover, some of Kylie's high-frequency clicks were polycyclic and similar, in

Table 7.1: Tukey's HSD test. Result table of a Tukey's HSD multiple comparisons test among proportions (n = 4) performed in the proportion of high frequency clicks emitted by a short-beaked common dolphin while travelling alone or with a harbour porpoise in the Firth of Clyde, Scotland. SE = Standard error. q = value obtained in the test, which is compared to a critical q value (qc) from the studentised range distribution. If q is larger than qc from the distribution, the two means are significantly different. S = Significant ($\alpha = 0.05$). Scenarios: 1 = Travelling alone, 2 = Accompanied by a porpoise but not travelling next to it, 3 = Travelling very close to a porpoise, 4 = Overall when travelling near a porpoise.

Comparison		Difference	SE	q	$\mathbf{q}\mathbf{c}$	Decision
	4 - 3	7.11	0.814	8.74	3.633	S
	4 - 2	23.94	1.098	21.80	3.633	\mathbf{S}
	4 - 1	6.73	1.119	6.01	3.633	\mathbf{S}
	3 - 2	31.05	1.149	27.03	3.633	\mathbf{S}
	3 - 1	13.84	1.169	11.84	3.633	\mathbf{S}
	2 - 1	17.21	1.382	12.45	3.633	\mathbf{S}



Figure 7.10: High frequency. High-frequency click produced by Kylie when travelling together with a harbour porpoise. Note that the waveform is olygocyclic, with few peaks, compared to the waveform of harbour porpoise clicks



Figure 7.11: Low frequency. Typical click train of low-frequency clicks produced by Kylie when travelling together with a harbour porpoise. Top: variation of amplitude levels (dB re: 1μ Pa) in time. Bottom: power spectrum of consecutive clicks.

both time and frequency domains, to those produced by harbour porpoises (Figure 7.12). Note, however, that while the spectral characteristics are very similar, it is still possible to distinguish between Kylie's and the porpoise clicks, as Kylie's clicks have energy below 100 kHz (visible in the power spectrum and the Wigner plot) and the envelope of the signal in the time domain is different to the Gaussian shape of the porpoise click.

Some clicks with high-frequency peaks are an intermediate between low and highfrequencies (Figure 7.13), with mixed characteristics in the power spectrum, but different enough to distinguish from porpoise clicks. Two examples are shown in Figure 7.14, both of which are broadband but with most of the energy between 100 and 150 kHz. Figure 7.13 shows a spectrogram of all clicks produced by Kylie in this study, sorted by centroid frequency. It is clear that Kylie shows great ability to



Figure 7.12: Matching. Left: Typical low-frequency click produced by Kylie when alone. Middle: typical click produced by harbour porpoises. Note the lack of energy below 100 kHz. Right: High-frequency click produced by Kylie when travelling close to a porpoise. Note that while most of the energy is concentrated between 100 and 150 kHz, there is significant energy below 100 kHz. Additionally, the waveform is unlike a porpoise click as it begins with an abrupt onset.

change the spectral characteristic of the clicks. This ability was previously observed in long-beaked common dolphins [172].

Production learning as a result of interactions with another species is not uncommon for cetaceans kept in closed enclosures [e.g., 185, 188, 256]; however, they are rare in the wild [190]. This is the first study to present evidence for vocal production learning in short-beaked common dolphins. The results of this study show Kylie alters the spectral characteristics of her echolocation clicks by increasing its centroid frequency to match those of the porpoise. She was able to do this within a few seconds (Figure 7.17). Moreover, she produced a higher proportion of high fre-



Figure 7.13: Concatenated spectrogram of Kylie's clicks. Clicks were sorted by centroid frequency which are highlighted by the black line - as per [172].

quency clicks when travelling close to a harbour porpoise than in any other scenario. Furthermore, these results support the hypothesis of interspecific vocal communication, and constitutes the first report of this kind for cetaceans in their natural environment.

Janik and Slater [180, 181] state that production learning requires modifications of the frequencies of the signal to produce a signal that was not previously in the animal's or species' repertoire. Such modifications should result from exposure to acoustic inputs, either artificial sounds or by interacting with individuals of the same or another species [181]. It is known that long-beaked common dolphins have the ability to produce clicks with a wide range of centroid frequencies [172], and this study supports those findings. Nonetheless, the data indicates that this is not a case of contextual learning (*i.e.*, producing an existing signal in another context), but a case of production learning in the form of vocal convergence, following Tyack



Figure 7.14: *High frequency.* Broadband high-frequency clicks produced by Kylie when travelling together with a harbour porpoise.

(2008) [191]. Tyack suggests that vocal convergence in adult individuals are, in fact, the result of production learning [191]. All echolocation clicks with energy over 100 kHz described for dolphins are oligocyclic, of short-duration, and broadband [33, 170, 241], while at least some of the high-frequency clicks produced by Kylie were polycyclic, resembling NBHF clicks (Figure 7.12).

These findings suggest Kylie is actively modifying the characteristics of her clicks to match those of the harbour porpoise. The transition probabilities showed that Kylie was significantly more likely to keep producing high-frequency clicks when travelling next to the porpoise, while the porpoise was more likely to vocalise when not close to Kylie. Moreover, these results support the results of the runs test, indicating that the animals did not interrupt each other.

7.3.2 The porpoises

The spectral characteristics of the harbour porpoise clicks were consistent with what is known for this species, with centroid frequencies centred at 139.7 kHz (SD = ± 9.4 kHz) [6, 7, 114]. These values are also within the values found for the porpoises in the Firth of Clyde recorded throughout 2016 and 2017, which had a mean centroid frequency of 136.4 kHz (SD = ± 0.6 kHz), ranging from 107 kHz to 159.9 kHz (Table 7.2). The centroid frequency values for the porpoises that interacted with Kylie are slightly higher than the overall population in the Firth of Clyde, but still within range. A recent unpublished work (PhD Thesis - [96]) found that young porpoises produce clicks with a higher peak and centroid frequencies than adults during the first 9 months. While the results come from one calf only, born in captivity, if this is representative, it is possible that Kylie is in fact visited by young individuals. Photographs of Kylie with porpoises also support this hypothesis (Figure 7.4).

The porpoise that accompanied Kylie on the first day emitted a total of 3 buzzes or calls and all of them were recorded while it was moving in a different direction than Kylie. These calls were between 100 and 180 ms long and consisted of several dozen clicks each (mean = 70). All calls followed the same pattern resembling a

Table 7.2: Porpoise clicks. Centroid frequency values for porpoise clicks per day and the overall, and the population

	Mean $(\pm SD)$	Min	Max
Day 1 $(n = 379)$	$136.5 (\pm 8.1)$	118.9	172.7
Day 2 $(n = 119)$	$147.1 \ (\pm 7.2)$	120.5	161.2
Day 3 $(n = 29)$	$151.9 (\pm 5.8)$	136.7	162.9
All 3 days $(n = 527)$	$139.7 (\pm 9.4)$	118.97	172.7
All porpoises $(n = 25764)$	$136.4 (\pm 8.1)$	107.0	159.96

V: the calls begin with a repetition rate of between 250 and 270 clicks per second (CPS), going below 150 CPS and back to 150-170 CPS again Figure 7.15.



Figure 7.15: Communication calls? clicks per second produced by the porpoise while spending time with Kylie, recorded on the 2nd of November 2016.

Potential communication calls in harbour porpoises have been recorded in at least three studies carried out on captive and free-ranging animals [10, 11, 12, 13]. The first study to describe potential communication calls was conducted in the 1960s. Not much attention was given to these, as only 3 studies focused on communication in harbour porpoises since then (see Chapter 6 for more information). The stereotyped pattern of these buzzes suggest these may be communication calls, which may function to maintain cohesion or contact with other individuals [189].

7.3.3 Consecutive clicks

Visual inspection of all clicks organised in chronological order against direction of arrival (estimated using the time of arrival differences [224]) revealed Kylie and the porpoise travelled close to each other, at least part of the time, which is evident by the consistency in the direction of where the clicks are coming from during a long period (Figure 7.16). Additionally, Kylie can rapidly change the spectral characteristics of the clicks, moving from low to high frequencies within a few ms (Figure 7.16). This variation of the spectral characteristics of Kylie's clicks was

previously reported for long-beaked common dolphins [172]. This, and the fact that low- and high-frequency clicks were narrowband, while mid-frequency clicks are broadband, could explain the inconsistent descriptions in the literature [39, 171, 257].



Figure 7.16: Consecutive clicks. Clicks produced by Kylie and a harbour porpoise while travelling in close proximity (evidenced by the consistency of the direction where these clicks come from (i.e., y axis)) plotted in chronological order (n = 1,248). The position of the vessel is at 0°. Porpoise clicks are represented as a black rhombus and dolphin clicks are dots colour coded by centroid frequency (in kHz). Recordings were made on 2nd (top figure) and 3rd (middle figure) November 2016, and 4th September 2017 (bottom figure) in the Firth of Clyde, Scotland.

Echolocation signals are highly directional and so the waveform and spectral content suffer from distortion at the edges of the beam. This spectral filtering due to orientation is not significant for harbour porpoises [53], but it is for delphinids. It is therefore possible that the variability observed could, at least in part, be the result of changes in the orientation of Kylie with respect to the hydrophones. However,

off-axis clicks are biased towards low frequencies [232] and the changes observed in this study are the opposite, with the increased proportion of high-frequency clicks. Figure 7.17 shows an example of a click train produced by Kylie when she changes from low-frequency to high-frequency clicks within seconds, moving through broadband or bimodal clicks.



Figure 7.17: Waterfall. Click train produced by Kylie over a period of 4 s (above). Recordings were made on the 2nd of November 2016 while seen travelling with a harbour porpoise.

The transition probabilities showed that Kylie was significantly more likely to produce consecutive high-frequency clicks when travelling with the porpoise than when the porpoise was far from her. Self-transitions for Kylie were higher than transitions into other states, which means that once she started vocalising she was more likely to continue to vocalise (Figure 7.18). On the other hand, the porpoise was less likely to continue vocalising when it was close to Kylie than when far from

her, suggesting the porpoise tried to keep in contact with Kylie when far, and that it did not interrupt Kylie when next to her. These results are consistent with the results of the Wald–Wolfowitz runs test for individual clicks and click trains (see below).



Figure 7.18: Markov Chains. Transition probabilities between low-frequency, highfrequency, and porpoise clicks in two different scenarios: while travelling close to each other (left) and while being in the same area but not close to each other (right).

7.3.4 Overlap

During periods when Kylie and a porpoise were travelling next to each other, a total of 166 click trains were recorded, based on ICI and species, 135 of which were produced by Kylie. Based on ICI only, 49 click trains were detected (Table 7.3). Of these, 27 were produced by Kylie, 5 by the porpoise, and 17 had clicks produced by both individuals. In most cases when there was overlap, it was the porpoise producing few clicks as Kylie vocalised. The null hypothesis for the Wald–Wolfowitz

runs test performed on consecutive clicks when the dolphin and the porpoise were travelling next to each other (n = 1,248) was rejected (p = $< 1^{-50}$, $\alpha = 0.05$), indicating that clicks were not produced at random. This means that once the porpoise (or Kylie) started vocalising, they were more likely to continue vocalising and, therefore, there was no significant overlap in their clicks (*i.e.*, they do not interrupt each other). These results are consistent with the results of the Markov Chains.

The potential turn-taking behaviour indicated by these results is apparent in Figure 7.19, especially for day 1 (top figure) and day 3 (bottom figure).

Turn taking has not been studied thoroughly in non-human animals except, maybe, in non-human primates, although it seems to be present in many species of birds and mammals, including cetaceans [258]. The phenomenon is not always described as turn-taking, however, and the terminology is not consistent between studies, making it difficult to identify [259]. A recent study based on a 34-sec recording of two captive bottlenose dolphins reported a clear case of turn-taking. The author also claimed that the dolphins form sentences, much as humans do [260]. Similarly, Schulz reported that different sperm whale (*Physeter macrocephalus*) groups show a similar pattern when it comes to coda production. Codas are a series of clicks emit-

10001 0 0000 (2	1 b) and or species					
		Median	Mean $(\pm SD)$	Min	Max	
ICI	Both $(n = 17)$	36.5	$54.3 (\pm 52.6)$	3	189	
	Kylie (n = 27)	6	$11.1 (\pm 11.1)$	1	37	
	Porpoise $(n = 5)$	4	$5.7 (\pm 3.9)$	2	13	

1

3

 $8.1 (\pm 20.7)$

 $5.4 (\pm 5.7)$

1

2

163

22

Kylie (n = 135)

Porpoise (n = 27)

ICI and Species

Table 7.3: Click trains. Mean and median lengths of click trains based on inter-click intervals (>1 s) and/or species





Figure 7.19: Turn taking. Click trains of both the harbour porpoise (top lines) and Kylie (bottom lines) for 2nd of November 2016 (top figure), 3rd of November 2016 (middle figure), and 4th of September 2017 (bottom figure), colour-coded by centroid frequency (kHz).

ted with a particular pattern and are thought to be used for communication. Schulz found that most of the time codas from different animals did not overlap (only in 15-16% of cases) and that in 22% of cases they were followed by codas produced by another individual. In addition, the authors found evidence of matching of the coda types by either the responder (*i.e.*, the whale that produces the codas later) and the initiator (*i.e.*, the first whale to produce a coda) [258]. More studies focused on this behaviour are needed before any conclusions can be drawn, but the available data suggest it might be more common in the animal kingdom than previously thought [259].

Turn taking in spontaneous conversations in humans follows a series of rules described by Sacks et al. (1974) [261]. For two speakers these rules are (1) Speaker-

change recurs, or at least occurs (2) Overwhelmingly, one party talks at a time (i.e., when one speaker starts vocalising, it is more likely to continue vocalising) (3) Occurrences of more than one speaker at a time are common, but brief <math>(i.e., they usually do not interrupt each other) (4) Transitions (from one turn to a next) with no gap and no overlap are common. (5) Turn order is not fixed (6) Turn size is not fixed (7) Length of conversation is not specified in advance (8) What parties say is not specified in advance (9) Relative distribution of turns is not specified in advance (10) Number of parties can vary (11) Talk can be continuous or discontinuous (12) Repair mechanisms exist for dealing with turn-taking errors and violations; *e.g.*, if two parties find themselves talking at the same time, one of them will stop prematurely, thus repairing the trouble [261]. The vocalisation pattern observed during the periods where Kylie and a porpoise were travelling next to each other follows these rules almost in its entirety.

Whether communication signals carry information is still a matter of debate. Some authors argue that the senders emit signals to manipulate or elicit a behaviour change in the receiver [262], while others argue that the purpose of signalling is to transmit information regardless of what the receiver does with it (see [117]). It is possible that some of Kylie's high-frequency clicks were misidentified as porpoise clicks and vice versa, however, given the stereotypical characteristic of NBHF clicks, it is expected that the error rate is low. The overall results, therefore, indicate Kylie is actively producing more high-frequency clicks when in close proximity to the harbour porpoise and that there is little overlap in their sounds. Moreover, this is consistent with the hypothesis of vocal communication between Kylie and the porpoises. Furthermore, the evidence suggests vocal production learning abilities,



Figure 7.20: Turn taking. Example of overlapping clicks produced by a harbour porpoise (vertical lines restricted to frequencies over 100 kHz) and Kylie, the solitary short-beaked common dolphin (remaining vertical lines). Top = waveform. Bottom = spectrogram. Recording made on the 2 of November 2016 at 1021 in the inner Firth of Clyde.

not previously reported for short-beaked common dolphins.

7.4 Summary

The results of this study provide evidence of vocal communication between Kylie and harbour porpoises. Kylie only produced clicks despite being able to produce other sounds, and emitted a higher proportion of high-frequency clicks while in close proximity to a porpoise. The results also indicate neither the porpoise or Kylie's clicks were produced at random and that there was little overlap in their sounds, suggesting turn-taking behaviour. Moreover, because the majority of clicks were clearly produced by either (*e.g.*, Figure 7.14 and Figure 7.10), even if mistakes were made in assigning clicks to either of the animals, the overall results suggest Kylie is capable of production learning in the form of vocal convergence.

These interactions may be similar to when humans and their dogs bark at each other. However, the results must be interpreted in the context in which they took place. Whistles are one of the commonest sounds detected during common dolphin aggregations [165]. With frequencies below 20 kHz, whistles are in the lower limit of the harbour porpoise hearing range [254], but Kylie produced exclusively clicks with peak frequencies of at least 23 kHz, some of which had similar temporal and spectral characteristics similar to harbour porpoise clicks (Figure 7.12).

In summary, these results support the hypothesis of vocal convergence and interspecies vocal communication, and constitute the first report of this kind for cetaceans in their natural environment.

Chapter 8

D-PorCCA - from acoustics to behaviour

In this Chapter the algorithms developed in the previous chapters are brought together into a standalone application, D-PorCCA (available here). D stands for 'detector', PorCC for 'porpoise click classifier' and A for 'application'.

8.1 Introduction

Recent technological advancement, in both storage and processing power capabilities, have made it possible to use passive acoustic monitoring (PAM) devices that record continuously to monitor different cetacean species for long periods of time. This, however, leaves bioacousticians with the challenge of having the time and resources to analyse the large amounts of data they collected. Data analysis is time-consuming and this usually results in leaving data unanalysed. But, it is now

CHAPTER 8: D-PORCCA - FROM ACOUSTICS TO BEHAVIOUR

possible to develop analytical tools that are capable of providing researchers with pre-selected acoustic events with minimal input to overcome this problem.

This Chapter describes the main functionalities of D-PorCCA. The application includes the algorithms developed in the previous Chapters, which together aim to study the behaviour of harbour porpoises in their natural environment from the repetition patterns of their acoustic signals.

8.2 Description

D-PorCCA is a standalone application with a user-friendly interface. It was developed using App Designer, a MATLAB (Mathworks, MA) development environment that allows to create desktop or web apps that can be used without having a MAT-LAB account. The executable file automatically downloads 'MATLAB Compiler' for installation, after which the user can open the application like any other desktop application. App Designer was introduced with the 2016a version of MATLAB. The environment is easy to use and has most functionalities that exist in the older 'GUIDE' environment. With App Designer adding components and functions is straightforward, facilitating improvements and updates work.

D-PorCCA has two main tabs: the Main Display where individual click trains can be explored and the Metrics Display, to visualise summary data.

8.2.1 Main display

The application opens on the main page (Figure 8.1). Here, the user can see simultaneously three aspects of the click trains: variations in amplitude, repetition rates or inter-click intervals, and centroid frequency or direction of arrival of each click, if

CHAPTER 8: D-PORCCA - FROM ACOUSTICS TO BEHAVIOUR

available.



Figure 8.1: Main page. D-PorCCA main page.

Additionally, a series of parameters are available, including date and time, length and type of click train, as well as behaviour. The user can also make changes and add notes. These functions are described in detail below:

- Date and time: date and time of the beginning of the click train.
- **Day/Night**: available if the user provides latitude and longitude (in decimal degrees) of the devices, whether they are moored to the seabed or of the central area, in the case of surveys conducted using towed hydrophones. If the survey area is relatively small, it is expected to have minimal or no impact
on the results. The coordinates are used to estimate sunrise and sunset times of each day using the function 'sunrise' developed by Beauducel (2019) [263]. Subsequently, this information is used to determine whether the click train was recorded at day or night time.

- Length: total number of clicks in the click train.
- Click train type: there are four categories of click train types: 'NBHF', 'LQ-NBHF', 'Sonar', or 'Non-NBHF'. More information on the decision-making pathway for these categories is provided in Chapter 5 - Patterns of click trains.
- Behaviour: there are four behavioural categories: 'Orientation', 'Foraging', 'Socialising', and 'Unknown', and are available for click trains classified as 'NBHF'. More information on how click trains are assigned to either of these categories is provided in Chapter 5 Patterns of click trains. Further information and examples of each category are also available in Chapter 6 Acoustic behaviour.
- **Calf**: D-PorCCA does not currently include a classifier for young harbour porpoises, however, there is potential to develop such classifier in future work and add it to the application.
- Notes: add (or see existing) notes for the click train.
- Save updates: save changes.
- **Display**: select to display only 'NBHF' click trains or all types.
- Individual click trains 3D: prompt a 3D display (Figure 8.2). The user can

inspect each click individually for verification purposes as well as to visualise specific aspects of it that cannot be appreciated in other ways (e.g., energy content). In this display it is also possible to delete clicks individually.

- **Spectrogram**: prompt spectrogram of the click train if the recordings (*.wav* files) are available.
- Selection: select individual click trains to extract for further analysis.
- Validation: change the type and behaviour of the click train manually.



Figure 8.2: *Individual click trains*. Pop up window showing a waterfall image of a click train.

Visualising click trains in 3D is useful for verification purposes as the energy content is evident in the power spectrum. Additionally, each click can be inspected, including the waveform, power spectrum, and spectrogram. The user can delete clicks manually from this window if so desires.

8.2.1.1 Click trains

In D-PorCCA, a click train is defined as a series of consecutive clicks separated by less than a given time period (in ms), which the user can select (default = 1,000 ms). The user can also select the minimum (default = 16 clicks) and maximum (default = 1,500 clicks) number of clicks. Clicks displayed were classified by the PorCC algorithm [15] as either high-quality (HQ) or low-quality (LQ) clicks, which are seen in red and blue, respectively. Clips classified as 'Noise' are not displayed.

8.2.1.2 Behaviour

Four behavioural categories are used in D-PorCCA, which are described below:

- 'Orientation': characterised by repetition rates below 100 clicks per second (CPS). This indicates the porpoise is inspecting the environment it moves through.
- 'Foraging': Foraging events are characterised by three phases: search, approach, and feeding buzz [123]. In the search phase the porpoise produces fewer than 100 CPS; in the approach phase the repetition rate increases rapidly, and a final stage of high-production rate that can reach up to 640 CPS [*e.g.* 9]).
- 'Socialising': social calls are characterised by not having a period with slow repetition rates as feeding events do.
- 'Unknown': Click trains without a recognisable pattern.

A description of the algorithms developed to identify behavioural states from click trains is given in Chapter 5 - Patterns of click trains.

8.2.2 Metrics display

Within the Metrics display, there are two sections. On the left side, the user can see a summary table with information about click trains detected in the study period. On the right side, the user can choose between a series of metrics to be displayed (Figure 8.3). The metrics estimated are described below.

8.2.2.1 Metrics

A series of metrics are available as a summary data the user can inspect and download. These metrics were selected because they are already used in the literature or because they provide an easy way to visualise the data.

	METRICO	,									
LL DAYS					DISPLAY SE	TINGS					
Select main folder E:\data\PGDF_SoundTrap DATA Browse UPLOAD DATA					Select a date		Metrics to display O Positive porpoise minute O All O Type of CT Image: Behaviour Image: Night/Day			◯ All ● Night/Day	UPDATE
Date	NBHF	LQNBHF	NonNBHF	Noise	DISPLAY						
12-Aug-2015	0	3	1051	*	500 c						_
12-Aug-2015	103	109	1659		000						
13-Aug-2015	27	22	515		450 -						
13-Aug-2015	70	110	864								
24-Aug-2015	15	62	743		400 -						
24-Aug-2015	83	256	1205		250						
25-Aug-2015	0	0	0		350						
25-Aug-2015	165	375	1570		300 -						
27-Aug-2015	120	193	1489		Der						
27-Aug-2015	148	180	1280		토 250 -						
26-Sep-2015	7	29	1609		z						
26-Sep-2015	5	18	3028		200 -						
27-Sep-2015	0	0	0		150 -						
27-Sep-2015	1	2	208								
08-Nov-2015	3	4	15		100 -						
08-Nov-2015	12	19	68				_		_		
09-Nov-2015	0	0	0		50 -						
09-Nov-2015	26	46	306		0						
10 Nov 2015	2	1	20	•	5	Foraging	g Orier	ntation	Socialising	Unki	nown

Figure 8.3: Metrics. Metrics tab in D-PorCCA.

- 1. **Positive Porpoise Minute**: for each minute of the day, a positive porpoise minute is one where at least a NBHF click train is detected. This metric is available only if the data was collected using a static device. The plot is divided into two parts, one with the total number of positive minutes per day, and the other one plotting each positive minute in the 24 hours of the day (see Figure 5.16).
- 2. **Types of click trains**: four categories of click trains are shown: 'NBHF', 'LQ-NBHF', 'Non-NBHF', and 'Sonar'.
- Behaviour: four categories of behaviour are shown: 'Orientation', 'Foraging', 'Socialising', and 'Unknown'.

For each of these metrics, the user can visualise either the total, or the numbers divided into day vs night if available (Figure 8.3). Additionally, the user can select to plot data from the entire survey period or from a specific date. These plots are made from the summary table shown on the left side of the Metrics panel. Lastly, the user can save the plots as *.jpeg* or other format directly from the figure.

8.2.3 Menus

D-PorCCA includes a series of Menus to make the user's experience easier.

8.2.3.1 Detector

The built-in detector extracts potential clicks from the *.wav* file. The signal is filtered with a 6th order Butterworth band-pass, with the low cut-off frequency set at 2 kHz and the high cut-off frequency at 95% of half of the sampling frequency. These filters are applied to reduce the background noise at low frequencies. In the

next step, and to find potential clicks, an adaptive threshold is set based on the first 10,000 samples (*i.e.*, maximum value under 0.1). All peaks over that threshold, separated by at least 200 samples (*i.e.*, 400 μ s for a 500 kHz sampling frequency) are considered potential clicks, and are extracted by adding 120 sample to each side of the peak of the signal. Subsequently, all necessary variables and parameters are estimated, and files constructed to function within D-PorCCA.

8.2.3.2 PorCC

The user can run the ProCC classifier [15] by accessing individual clicks directly from the .pgdf files generated by PAMGuard's Click Detector Module.

Input values

The user must input a series of parameters for PorCC to function, which are also requested in the Detector menu. These are:

- Number of hydrophones or channels used in the recording process.
- Sampling frequency (kHz) Fs: The sampling frequency is the number of samples per second in a sound recording.
- Hydrophone sensitivity (dB re $1V/1\mu$ Pa): is the amplitude (dB re 1μ Pa) of the sound necessary to generate 1 Volt. This value is provided by the manufacturer and is used to estimate the amplitude of the signals.
- Gain (dB): total gain of the recording system. This value is used to estimate the amplitude of the signals.
- Peak to Peak (Volts): refers to the clipping value of the recording system. This information is provided by the manufacturer and is use to estimate the

amplitude of the given signals.

• **Probability thresholds**: PorCC estimates the probability of a given signal to be a high-quality (HQ) or low-quality (LQ) harbour porpoise (or narrow-band high-frequency -NBHF- click). The probability threshold value in the PorCC Menu is the threshold of selected by the user. The default (and recommended) values for HQ is 0.999999 and for LQ is 0.6.

8.2.3.3 Click trains

In this menu, the user can run the algorithms to organise the data (e.g., tables created by the Detector or the PorCC algorithm) into click trains, applying the 'Pattern algorithm' and 'Behaviour algorithm' developed in Chapter 5 - Patterns of click trains.

Once these algorithms are applied, the user can open existing projects.

8.2.3.4 Download

The user is provided with a list of summary data they can download in .csv format:

- Click trains: Information about each click train is stored in a table, which contains the following parameters: Identification number, date and time, where it is stored (*i.e.*, *.pgdf* or *.wav* file), length (number of clicks), day or night time, type, behaviour, selected (0/1), and notes.
- Summary table: Summary of parameters for each day (*e.g.*, total number of NBHF click trains).

8.2.3.5 Help

In this section, the user can find information about D-PorCCA, and they can search for specific topics, including input values, how to set up the application, and what each of the behaviours represent.

8.3 Other applications

There are at least 16 species of whales, dolphins, and porpoises that produce narrowband high-frequency (NBHF) clicks. Due to the similarities in the characteristics of these clicks, tools developed to study porpoises were successfully used to study other NBHF species [e.g., 212]. This suggests that D-PorCCA can potentially be used to study other NBHF species. This Chapter presents the results of using D-PorCCA on data from Heaviside's dolphin (*Cephalorhynchus heavisidii*).

8.3.1 Heaviside's dolphins

Heaviside's dolphins are small, coastal dolphins (Figure 8.4) endemic to the west coast of southern Africa. They mainly occur within the Benguela ecosystem, which comprises the shelf waters of the west coast of South Africa, Namibia, and southern Angola. Heaviside's dolphins have small home ranges and at least some individuals show site fidelity [264]. Moreover, there is evidence of a temporal and spacial segregation of behaviours, resting and socialising close to the coast during the day, and foraging far from shore at night [265].

The first study which focused on the sounds of Heaviside's dolphins was carried out in the 1970s [61]. The authors used equipment that was not able to record highfrequency sounds and therefore only low-frequency sounds were reported. Watkins et al did not record whistles or other tonal sounds typical of other delphinids [*e.g.*, 266], only pulsed sounds (*i.e.*, clicks) with energy below 5 kHz. Depending on repetition rates, three sound types were described: 'clicks at slow rates, short bursts of pulses, and series of pulses at a rapid enough rate to produce a tonal "cry" sound' [61].



Figure 8.4: Heaviside's dolphins. Picture by Stephani Flynn from the Namibian Dolphin Project

In 2011, Morisaka et al. used technologies able to record and describe the high-frequency components of the Heaviside's dolphin clicks for the first time. These are narrow-band high-frequency (NBHF) clicks with temporal and spectral characteristics similar to those produced by other non-whistling species, like the harbour porpoise [6]. They are polycyclic, with several peaks in the time domain, and have their energy concentrated between 100 and 150 kHz. Unlike other NBHF clicks, the distribution of the centroid frequency is bimodal, with peaks at 122 kHz and at 130 kHz. Individual clicks have one or two peaks at 122 kHz and 130 kHz, respectively, in the power spectrum. The reason for the double peak in the power spectrum is unknown [51].

More recently, Martin et al. (2018, 2019) found that Heaviside's dolphins also produce clicks with broader frequency bandwidths that have energy well below 100 kHz, which they produced both as click trains at slow rates and as burst pulses, which are thought to be used for communication purposes [50]. As shown in Figure 8.5, the repetition rates of these buzzes is variable and might be indicative of their behaviour, which has been observed for other NBHF species, such as the harbour porpoise and the Hector's dolphin (*Cephalorhynchus hectorii*) [11, 13, 43].



Figure 8.5: Examples of Heaviside's dolphin burst-pulse signals. (a) Patterned at the beginning, (b) patterned at the end, (c) patterned in the middle, (d) patterned at both the beginning and the end, (e) continuous segmented pattern and (f) series of four short burst-pulses. For each signal, (i) the interclick intervals (ICIs) throughout the signal and (ii) the spectrogram of the signal (512-pt, FFT, Hamming window, 50% overlap; MATLAB version 2017b) are shown. NBHF click trains from other individuals can be seen overlapping in all spectrograms except in (c) and (d). Note only the latter half of the continuously segmented patterned signal shown was suitable for click detection. From Martin et al. 2019 [62]

The aim of this Section was to test the algorithms in D-PorCCA for acoustic data from Heaviside's dolphins.

8.3.2 Methods

The acoustic data used in this Section were collected using a SoundTrap ST300 HF off Walvis Bay (Namibia) during June 2019. The data were collected by The Namibian Dolphin Project as part of an ongoing project aiming to compare the performance of two existing static passive acoustic monitoring devices. The SoundTrap was deployed 3 m below the water surface from a drifting boat in the vicinity of Heaviside's dolphins. No other cetacean species were seen in the area. A total of seven *.wav* files were used, corresponding to approximately 37 min of recordings. Three files were from the 10th, three from the 11th, and one from the 26th of June 2019.

A custom-built impulsive-sound detector (see Section 8.2.3.1) was used to detect potential dolphin clicks within the *.wav* files. Each of these potential clicks were classified by PorCC ([15] - also see Chapter 4 - PorCC: the harbour porpoise click classifier) into one of three categories: high-quality clicks (HQ), low-quality clicks (LQ), and noise. The probability thresholds were the default values, that is for HQ clicks was set at 0.999999 and for LQ at 0.6. For LQ and HQ clicks, a series of parameters were estimated (see Section 3.3) and saved for further analysis in D-PorCCA (see Chapter 8 D-PorCCA - from acoustics to behaviour).

8.3.3 Results and discussion

A total of 1,936 Heaviside's dolphin clicks were found. Of these, PorCC classified 1,272 as LQ and 664 as HQ. Using 500 ms as the maximum separation time between two consecutive clicks, these clicks were found in click trains (n = 24) of between 11 and 333 clicks, including 2 buzzes (n = 493 clicks). Some of the clicks were likely noise or echoes, but because the data were collected in a quiet environment, it was not necessary to do any further cleaning of the click trains. Most clicks were typical NBHF clicks, with centroid frequencies around 130 and 132 kHz, but some had peaks below 110 kHz, as those reported by Martin et al [50, 62]. The data used here did show a bimodal distribution of peak frequencies but not the one reported before [51], as the peak at 122 kHz is not visible (8.6). This discrepancy could be due to the small sample size used here.



Figure 8.6: Centroid frequency. Histogram of the centroid frequencies of all the Heaviside's clicks found in this study (n = 1,940).

Almost a third of the clicks had peak frequencies between 100 and 110 kHz, which is clearly visible in Figure 8.6. Because PorCC ignores clicks with peak or centroid frequency below 100 kHz, some clicks may have been missed, given the abrupt cut at frequencies close to 100 kHz seen in Figure 8.6. These clicks made up the buzzes and are the broadband clicks previously described by Martin et al [50, 62], which they hypothesised were used for communication purposes. No attempts were made to match the pattern of the buzzes with the visual observations made concurrently with the recordings.



Figure 8.7: *Heaviside's dolphins. Broadband buzz produced by a Heaviside's dolphins off Namibia - spectrogram of the buzz from the original recording.*

The broadband nature of these clicks is visible in Figure 8.7, showing they do have energy below 100 kHz, unlike harbour porpoise clicks [7], yet PorCC classified them as NBHF clicks using the default threshold values. This is counter-intuitive, however PorCC classification does not use -10 dB bandwidth, which is the parameter that better describes the broadband feature of these clicks. This is evident in the power spectrum of the buzz (Figure 8.8).

The repetition rate of the buzz was over 500 clicks per second, which can be ap-



Figure 8.8: *Heaviside's dolphins*. Broadband buzz produced by a Heaviside's dolphins off Namibia - spectrogram of the buzz from the original recording.

preciated in the main display of D-PorCCA (Figure 8.9), where also two overlapping click trains are visible: the broadband buzz, and few clicks with higher amplitude and higher centroid frequency. The latter are not visible in the spectrogram (Figure 8.7).

Typical NBHF click trains had higher peak and centroid frequency and were produced at low repetition rates, ranging between 10 and 60 clicks per second. Low repetition rates indicate these clicks are typical echolocation clicks used by dolphins to orient themselves in the environment.

These are promising results. Although the recordings were made in a quiet environment, the potential application for other NBHF is clear. More research is needed to test whether PorCC and other algorithms in D-PorCCA can identify the broadband clicks in an area with different levels and types of background noise.



Figure 8.9: Heaviside's dolphins. Broadband buzz produced by a Heaviside's dolphins off Namibia - buzz as seen in D-PorCCA.

8.3.4 Conclusions

A small sample of recordings of Heaviside's dolphins was used to test the potential of using D-PorCCA to study other NBHF species. The results suggest it is indeed possible, even using the default values. Moreover, PorCC, the harbour porpoise click classifier developed in this thesis [15], was able to identify both the typical NBHF and the broadband clicks produced by Heaviside's dolphins [50, 62]. Modifications may be needed to detect the whole range of Heaviside's dolphins clicks.

Overall, the results indicate all algorithms within D-PorCCA, including those to detect behavioural patterns (*i.e.*, travelling/orientation, foraging/feeding, and so-

cialising/communication) can also be used for Heaviside's dolphins.

Chapter 9

Conclusions

This Chapter presents an overview of the thesis, highlighting the main results and discussing the implications in context. Lastly, future lines of research are recommended.

9.1 Overview of the Thesis

The main goal of this thesis was to develop a standalone application, D-PorCCA, to study the behaviour of harbour porpoises in their natural environment. Harbour porpoises vocalise almost continuously producing only stereotyped clicks, and available data prior to this thesis indicated that the behaviour could be inferred from the patterns in repetition rates. The behaviours could be divided into three main categories: travelling or orientation, foraging and feeding, and socialising and communication.

Data on social calls came almost exclusively from experiments conducted in captive

settings. These studies found several call types, including 'distress', 'aggression', 'contact', and 'courtship' [10, 11, 12]. In the wild, one study found a series of sounds that resemble social calls previously described in the literature [8, 11]. The authors recorded opportunistically a young harbour porpoise trapped in a fishing net using a static acoustic device that logs data about acoustic events (*i.e.*, T-POD - Chelonia Ltd., UK). In this case the authors knew where to look in the data, which is almost never possible when using passive acoustic monitoring (PAM) devices.

The most recent study focused on repetition rates in wild harbour porpoise is Sorensen et al. (2018). In this case, the data came from six porpoises that were accidentally caught in fishing nets in Denmark. The animals were mounted with behavioural and acoustic tags, and released into the wild. Data confirmed porpoises produce social calls that are qualitatively different from other sounds.

This means that PAM could be used for behavioural studies. However, the required analytical tools for such studies were not yet available. This is the first study focused on vocalisation patterns of harbour porpoise in the wild using PAM devices that record continuously.

A series of tools have been developed in this thesis, which can identify clicks produced by harbour porpoises and other species with similar sounds, and use the underlying patterns of repetition rates to identify behaviours. These algorithms were tested in two datasets from different study areas (Denmark and Scotland) and collected with different devices (SoundTrap - Ocean Instruments, New Zealand, and towed hydrophone arrays). All algorithms showed high levels of accuracy, between 69.2% and 98.5%. These results suggest D-PorCCA and the algorithms within it have potential for behavioural studies as well as for monitoring purposes using noninvasive methods.

Additionally, in this thesis, the interspecific interactions of wild harbour porpoises with a solitary short-beaked common dolphin was studied, as a study case of potential interspecific communication. Using two hydrophones it was possible to use the direction of arrival of individual clicks, as well as the temporal and spectral characteristics to study the acoustic behaviour of the dolphin and the porpoises.

9.2 Summary of Findings

The main results of this thesis are highlighted below.

9.2.1 Behaviours

Using the algorithms developed in this thesis, the three main behavioural categories were observed, based on the patterns of repetition rates. Moreover, subpatterns within those categories were also found, suggesting there is potential for more detailed studies.

9.2.1.1 Orientation

Within 'Orientation', two subpatterns were found. First, patterns called 'landmarks', which indicate that the porpoises are using fixed objects in their environment to orient themselves. The second were a series of click trains with very low repetition rates, from 3 to 10 clicks per second. These may represent sleeping or resting behaviour.

9.2.1.2 Feeding in water column?

Harbour porpoises feed on pelagic (*e.g.*, herring) and demersal species (*e.g.*, sand eel). It is expected that the foraging technique used to catch free-swimming herrings is different than the one used to catch sand eels. Such differences are also likely reflected in the acoustic behaviour of the porpoise. In both cases a search phase and a final feeding buzz are expected, however, the approach phase could take different forms.



Figure 9.1: Foraging?. Example of a click train produced by harbour porpoises.

Some repetition rates did not have a clear pattern nor were similar to any of the patterns from the rest of the data. Although no visual data was available, some of these patterns could potentially represent unsuccessful foraging events where the porpoise chased a fish in the water column. One possible example of this, is shown in Figure 9.1.

9.2.1.3 Social calls

For the first time, social calls of wild harbour porpoises were detected and described using passive acoustic monitoring devices that record continuously. These include calls previously reported (e.g., consistent with distress calls [e.g., 11]) as well as new ones (e.g., calls with V-shape pattern). The most striking discovery is the identification of 'phrases', which are comprised by a series of calls (the same type or not) produced in a short period of time. Phrases were first described by Amundin in 1991 [11] and were never reported again.

9.2.2 Vocal imitation

The acoustic data collected during interactions between harbour porpoises and a solitary short-beaked common dolphin in the west of Scotland yielded noteworthy results. First, the data suggest the dolphin changes the characteristics of her echolocation clicks to match, or imitate those of the porpoise. Second, at least one of the porpoises emitted a 'phrase' comprising three V-shaped calls during these interactions. Lastly, the data suggest turn-taking behaviour, as the animals vocalised with little or no overlap.

9.3 Implications of the Research

The implications of this research are two-fold. First and foremost, this thesis provide new analytical tools that would make behavioural studies possible in a non-invasive way. Secondly, using D-PorCCA to analyse existing and new data can reveal important areas for harbour porpoises that need protection.

9.3.0.1 Behavioural studies

The results of this study confirmed harbour porpoises produce sounds with different patterns in repetition rates, many of which are consistent with social calls. More-

over, it suggest the patterns may be similar between populations as similar calls were found in both Scottish and Danish data. Most of these calls had repetition rates over 100 clicks per second (e.g., with inter-click intervals below 10 ms). Further, some calls overlapped with patterns of feeding buzzes.

Our understanding of harbour porpoise feeding behaviour may need to be reconsidered due to the results of this study. First, many studies use the 10 ms inter-click interval as a threshold to detect a feeding event, which may lead to overestimation of feeding activity, given the characteristics of social calls. Second, foraging events have been described using the same characteristics in repetition rates, namely a search phase, an approach phase, and a final feeding buzz. Although this pattern has been reported widely and been used to detect feeding events from passive acoustic monitoring data, some of the patterns found in this study may challenge that idea.

Lastly, harbour porpoises belong to a group commonly known as 'non-whistling' cetaceans. At least 16 species from this group, including whales, porpoises, and dolphins, produce very similar sounds. These species also have in common their seemingly lack of social life. The tools developed here can potentially be used to study the acoustic behaviour of those species.

9.3.1 Conservation

Knowledge about how porpoises use their environment is crucial to identifying key areas for them and to develop effective conservation measures, such as Special Areas of Conservation required under the EU Habitats Directive [14]. For example, identifying patterns related to courtship or mating can be used to identify breeding

areas and thus avoid overlap with human activities.

In addition, harbour porpoises are highly sensitive to noise and these tools can help us understand how noise-producing human activities affect their behaviour and if it interferes with life functions (*e.g.*, feeding), and thus provide countries with tools to collect robust data to fulfil their legal obligations.

9.4 Recommendations for Future Research

Throughout this Thesis, a number of practical or theoretical limitations were encountered. These are discussed here to provide insight into how these limitations could be overcome. In doing so, the potential for future research is elucidated.

9.4.1 Compare with existing tools

The T-PODs and C-PODs (Chelonia Ltd., UK) are the most used passive acoustic monitoring devices to study harbour porpoises in the wild. Given these devices do not record the waveform of the signals and that the algorithms are not publicly available, validation is subjective and not always possible, especially with regard to missed events. D-PorCCA has similar functionalities and thus the performance should be tested against each other. This is specially important in areas with high background noise.

9.4.2 Visual data

Many of the patterns found in this study are consistent with communication or social calls reported in the literature. However, many patterns had not been described before, and therefore visual surveys carried out concurrently with acoustic recordings are necessary to understand these patterns. Although data could be obtained from experiments in captivity, behavioural data is best from the wild.

9.4.3 Separation of click trains

If data is collected with at least two hydrophones it is possible to estimate the direction of arrival of each signal, facilitating separation of click trains. However, data is usually collected using one hydrophone and thus algorithms able to separate overlapping click trains based on one sensor would be needed. The algorithms developed in this thesis were not able to deal with this problem.

9.4.4 Calves

Calves and young harbour porpoises produce clicks with different temporal and spectral characteristics than those of adults. The existing data suggest that this information could be used to identify the presence of young individuals during acoustic events. An algorithm able to distinguish between adults and young individuals could be easily incorporated into D-PorCCA and supplemented with visual data.

9.4.5 Other species

There are at least 16 species of whales, dolphins, and porpoises that produce narrowband high-frequency clicks similar to those of the harbour porpoises. It was shown in this thesis that D-PorCCA can potentially be used to study those species, such as the pygmy and dwarf sperm whales, of which we know next to nothing.

Bibliography

- P Hammond, Giovanni Bearzi, Arne Bjørge, Karin A. Forney, Leszek Karczmarski, T. Kasuya, William F. Perrin, M.D. Scott, J.Y. Wang, Randall S. Wells, and Ben Wilson. *Phocoena phocoena*. The IUCN Red List of Threatened Species. 8235, 2008.
- [2] Philip S. Hammond, Harald Benke, Per Berggren, Anne Collet, Mads Peter Heide-Jørgensen, Sara Heimlich-Boran, Mardik F. Leopold, and Nils Øien. The distribution and abundance of harbour porpoises and other small cetaceans in the North Sea and adjacent waters. Technical report, Sea Mammal Research Unit, Gatty Marine Laboratory, University of St Andrews, Fife, UK., 1995.
- [3] Anna Hall. Foraging behaviour and reproductive season habitat selection of northeast Pacific porpoises. PhD thesis, The University of British Columbia, 2011.
- [4] Jonas Teilmann, Casper T. Christiansen, Sanne Kjellerup, Rune Dietz, and Gösta Nachman. Geographic, seasonal, and diurnal surface behavior of harbor porpoises. *Marine Mammal Science*, 29(2):60–76, 2013.
- [5] Peter Teglberg Madsen, D. A. Carder, Kristian Beedholm, and Sam H. Ridg-

way. Porpoise clicks from a sperm whale nose—convergent evolution of 130 kHz pulses in toothed whale sonars? *Bioacoustics*, 15(2):195–206, 2005.

- [6] Bertel Møhl and S. Andersen. Echolocation: high-frequency component in the click of the harbour porpoise (Phocoena phocoena L.). Journal of the Acoustical Society of America, 54(5):1368–1379, 1973.
- [7] Michael Hansen, Magnus Wahlberg, and Peter Teglberg Madsen. Lowfrequency components in harbor porpoise (*Phocoena phocoena*) clicks: communication signal, by-products, or artifacts? *Journal of the Acoustical Society* of America, 124(6):4059, 2008.
- [8] Sven Koschinski, Ansgar Diederichs, and Mats Amundin. Click train patterns of free-ranging harbour porpoises acquired using T-PODs may be useful as indicators of their behaviour. *Journal of Cetacean Research and Management*, 10(2):147–155, 2008.
- [9] Stacy Lynn Deruiter, Alexander Bahr, Marie-Anne Blanchet, Sabina Fobian Hansen, Jakob Højer Kristensen, Peter Teglberg Madsen, Peter L. Tyack, and Magnus Wahlberg. Acoustic behaviour of echolocating porpoises during prey capture. The Journal of experimental biology, 212:3100–3107, 2009.
- [10] René-Guy Busnel and Albin Dziedzic. Acoustic signals of the pilot whale Glicephala melaena and of the porpoises *Delphinus delphis* and *Phocoena phocoena*. In Whales, Dolphins, and Porpoises, pages 608–646. 1966.
- [11] Mats Amundin. Sound production in odontocetes with emphasis on the harbour porpoise, Phocoena phocoena. PhD thesis, 1991.

- [12] Karin Tubbert Clausen, Magnus Wahlberg, Kristian Beedholm, Stacy Lynn Deruiter, and Peter Teglberg Madsen. Click communication in harbour porpoises *Phocoena phocoena*. *Bioacoustics*, 20:1–28, 2010.
- [13] Pernille Meyer Sørensen, Danuta Maria Wisniewska, Frants H. Jensen, Mark P. Johnson, Jonas Teilmann, and Peter Teglberg Madsen. Click communication in wild harbour porpoises (*Phocoena phocoena*). Scientific reports, page 8:9702, 2018.
- [14] Council Directive. Council Directive 92/43/EECof 21 May 1992 on the conservation of natural habitats and of wild fauna and flora, 1992.
- [15] Mel Cosentino, Francesco Guarato, Jakob Tougaard, David Nairn, Joseph C. Jackson, and James F. C. Windmill. Porpoise click classifier (PorCC): A high-accuracy classifier to study harbour porpoises (*Phocoena phocoena*) in the wild. *The Journal of the Acoustical Society of America*, 145(6):3427–3434, 2019.
- [16] Jay Barlow and Barbara L Taylor. Estimates of sperm whale abundance in the Northeastern temperate Pacific from a combined acoustic and visual survey. *Marine Mammal Science*, 21(3):429–445, 2005.
- [17] Sarah J. Dolman, Nicola K. Hodgins, Colin D. MacLeod, Graham John Pierce, and Caroline R. Weir. Harbour porpoises (*Phocoena Phocoena*) and minke whales (*Balaenoptera Acutorostrata*) observed during land-based surveys in The Minch, north-west Scotland. Journal of the Marine Biological Association of the United Kingdom, 94(6):1185–1194, 2014.

- [18] J. G.M. Thewissen, Lisa Noelle Cooper, Mark T. Clementz, Sunil Bajpai, and B. N. Tiwari. Whales originated from aquatic *artiodactyls* in the Eocene epoch of India. *Nature*, 450(7173):1190–1194, 2007.
- [19] Jonathan H. Geisler and Jessica M. Theodor. Hippopotamus and whale phylogeny. *Nature*, 458(7236):E1–E1, 2009.
- [20] Michael R. McGowen, Georgia Tsagkogeorga, Sandra Álvarez-carretero, Mario dos Reis, Monika Struebig, Robert Deaville, Paul D. Jepson, Simon Jarman, Andrea Polanowski, Phillip A. Morin, and Stephen J. Rossiter. Phylogenomic Resolution of the Cetacean Tree of Life Using Target Sequence Capture. Systematic Biology, page syz068, 2019.
- [21] A. Rus Hoelzel, Ada Natoli, Marilyn E. Dahlheim, Carlos Olavarria, Robin William Baird, and Nancy A. Black. Low worldwide genetic diversity in the killer whale (Orcinus orca): Implications for demographic history. Proceedings of the Royal Society B: Biological Sciences, 269(1499):1467–1473, 2002.
- [22] Annalisa Berta, James Sumich, and K Kovak. Marine mammals: Evolutionary biology. Academic press., San Diego, CA, second edition, 2008.
- [23] William F. Perrin, Bernd Würsig, and J. G. M. Thewissen, editors. Encyclopedia of Marine Mammals. Academic Press, second edition, 2009.
- [24] David K. Mellinger and Christopher W. Clark. Blue whale (Balaenoptera Musculus) sounds from the North Atlantic. The Journal of the Acoustical Society of America, 114(2):1108, 2003.

- [25] Lee A. Miller and Magnus Wahlberg. Echolocation by the harbor porpoise: Life in coastal waters. *Frontiers in Physiology*, 4(52):1–6, 2013.
- [26] John W. Richardson, R. Greene, Jr. Charles, Charles I. Malme, and Denis H. Thomson. *Marine mammals and noise*. 1995.
- [27] Walter Zimmer. Passive acoustic monitoring of cetaceans. Cambridge, University Press, 2011.
- [28] Laela S. Sayigh, Peter L. Tyack, Randall S. Wells, and Michael D. Scott. Signature whistles of free-ranging bottlenose dolphins, *Tursiops truncatus*: Stability and mother-offfspring comparisons. *Behavioral Ecology and Sociobiology*, 26:247–260, 1990.
- [29] Whitlow W. L. Au. Echolocation. In William F. Perrin, Ben Würsig, and J. G. M. Thewissen, editors, *Encyclopedia of Marine Mammals*, page 1352. Elsevier, London, second edi edition, 2009.
- [30] Sofie M. Van Parijs and Peter J. Corkeron. Vocalizations and behaviour of Pacific humpback dolphins Sousa Chinensis. Ethology, 107(8):701–716, 2001.
- [31] D. R. Griffin. Listening in the dark. Cornell University Press, Ithaca, NY, 1958.
- [32] Annemarie Surlykke, Paul E. Nachtigall, Richard R. Fay, and Arthur N. Popper, editors. *Biosonar*, volume 51. 2014.
- [33] Whitlow W. L. Au. Echolocation signals of wild dolphins. Acoustical Physics, 50(4):454–462, 2004.

- [34] Mafalda de Freitas, Frants H. Jensen, Julian A. Tyne, Lars Bejder, and Peter Teglberg Madsen. Echolocation parameters of Australian humpback dolphins (*Sousa sahulensis*) and Indo-Pacific bottlenose dolphins (*Tursiops Aduncus*) in the wild. *The Journal of the Acoustical Society of America*, 137(6):3033–3041, 2015.
- [35] Mafalda de Freitas, Joshua N Smith, Frants H Jensen, Kristian Beedholm, and Peter Teglberg Madsen. Echolocation click source parameters of Australian snubfin dolphins (Orcaella heinsohni). The Journal of the Acoustical Society of America, 143(4):2564–2569, 2018.
- [36] James J. Finneran, Brian K. Branstetter, Dorian S. Houser, Patrick W. Moore, Jason Mulsow, Cameron Martin, and Shaun Perisho. High-resolution measurement of a bottlenose dolphin's (*Tursiops Truncatus*) biosonar transmission beam pattern in the horizontal plane. *The Journal of the Acoustical Society* of America, 136(4):2025–2038, 2014.
- [37] Walter M. X. Zimmer, Mark P. Johnson, Peter Teglberg Madsen, and Peter L. Tyack. Echolocation clicks of free-ranging Cuvier's beaked whales (*Ziphius Cavirostris*). The Journal of the Acoustical Society of America, 117(6):3919– 3927, 2005.
- [38] Bertel Møhl, Magnus Wahlberg, Peter Teglberg Madsen, Lee A. Miller, and A Surlykke. Sperm whale clicks: directionality and source level revisited. The Journal of the Acoustical Society of America, 107(1):638–648, 2000.
- [39] W. E. Evans. Echolocation by marine delphinids and one species of fresh-water

dolphin. The Journal of the Acoustical Society of America, 54(1):191–199,1973.

- [40] Annamaria Izzi DeAngelis, Joy E. Stanistreet, Simone Baumann-Pickering, and Danielle M. Cholewiak. A description of echolocation clicks recorded in the presence of True's beaked whale (*Mesoplodon mirus*). The Journal of the Acoustical Society of America, 144(5):2691–2700, 2018.
- [41] Line Anker Kyhn, Jakob Tougaard, Kristian Beedholm, Frants H. Jensen, Erin Ashe, Rob Williams, and Peter Teglberg Madsen. Clicking in a killer whale habitat: narrow-band, high-frequency biosonar clicks of harbour porpoise (*Phocoena phocoena*) and Dall's Porpoise (*Phocoenoides dalli*). *PLoS ONE*, 8(5), 2013.
- [42] Maria Vanesa Reyes Reyes, Alexander Marino, Natalia A. Dellabianca, Marta Hevia, Mónica Torres, Andrea Raya Rey, and Mariana L. Melcón. Clicks of wild Burmeister's porpoises (*Phocoena spinipinnis*) in Tierra del Fuego, Argentina. *Marine Mammal Science*, (Published online Feb):1–12, 2018.
- [43] Stephen M. Dawson. Clicks and Communication: The Behavioural and Social Contexts of Hector's Dolphin Vocalizations. *Ethology*, 88(4):265–276, 1991.
- [44] Thomas Götz, Ricardo N. Antunes, and Sonja Heinrich. Echolocation clicks of free-ranging Chilean dolphins (*Cephalorhynchus eutropia*). *The Journal of the Acoustical Society of America*, 128(2):563–566, 2010.
- [45] Line Anker Kyhn. Passive acoustic monitoring of toothed whales, with implications for mitigation, management and biology. PhD thesis, Aarhus University, 2010.

- [46] Line Anker Kyhn, Jakob Tougaard, Frants Havmand Jensen, Magnus Wahlberg, G. Stone, A. Yoshinaga, Kristian Beedholm, and Peter Teglberg Madsen. Feeding at a high pitch: Source parameters of narrow band, high-frequency clicks from echolocating off-shore hourglass dolphins and coastal Hector's dolphins. *The Journal of the Acoustical Society of America*, 125(3):1783–1791, 2009.
- [47] Maria Vanesa Reyes Reyes, Vanesa P. Tossenberger, Miguel A. Iñiguez, John A. Hildebrand, and Mariana L. Melcón. Communication sounds of Commerson's dolphins (*Cephalorhynchus Commersonii*) and contextual use of vocalizations. *Marine Mammal Science*, 32(4):1219–1233, 2016.
- [48] Karlina Merkens, David Mann, Vincent M. Janik, Diane Claridge, Marie Hill, and Erin Oleson. Clicks of dwarf sperm whales (*Kogia sima*). Marine Mammal Science, 34(4):963–978, 2018.
- [49] Mariana L. Melcón, Mauricio Failla, and Miguel A. Iñíguez. Echolocation behavior of franciscana dolphins (*Pontoporia blainvillei*) in the wild. *The Journal of the Acoustical Society of America*, 131(6):EL448–EL453, 2012.
- [50] Morgan J. Martin, Tess Gridley, Simon H. Elwen, and Frants H. Jensen. Heaviside's dolphins (*Cephalorhynchus Heavisidii* relax acoustic crypsis to increase communication range. 2018.
- [51] Tadamichi Morisaka, Leszek Karczmarski, Tomonari Akamatsu, Mai Sakai, Steve Dawson, and Meredith Thornton. Echolocation signals of Heaviside's dolphins (*Cephalorhynchus heavisidii*). The Journal of the Acoustical Society of America, 129(1):449–457, 2011.

- [52] Peter Teglberg Madsen, Danuta Maria Wiśniewska, and Kristian Beedholm. Single source sound production and dynamic beam formation in echolocating harbour porpoises (*Phocoena phocoena*). Journal of Experimental Biology, 213(18):3105–3110, 2010.
- [53] Whitlow W. L. Au, Ronald A. Kastelein, Tineke Rippe, and Nicole M. Schooneman. Transmission beam pattern and echolocation signals of a harbour porpoise (*Phocoena phocoena*). Journal of the Acoustical Society of America, 106(6):3699–3705, 1999.
- [54] Jens C. Koblitz, Magnus Wahlberg, Peter Stilz, Peter Teglberg Madsen, Kristian Beedholm, and Hans-Ulrich Schnitzler. Asymmetry and dynamics of a narrow sonar beam in an echolocating harbor porpoise. The Journal of the Acoustical Society of America, 131(3):2315–2324, 2012.
- [55] Anders Galatius, Morten Tange Olsen, Mette Elstrup Steeman, Rachel A Racicot, Catherine D Bradshaw, Line A Kyhn, and L E E A Miller. Raising your voice: evolution of narrow-band high- frequency signals in toothed whales (Odontoceti). Biological Journal of the Linnean Society, 126:231–224, 2019.
- [56] Thomas A. Clay, Jeffrey C. Mangel, Joanna Alfaro-Shigueto, David J. Hodgson, and Brendan J. Godley. Distribution and Habitat Use of a Cryptic Small Cetacean, the Burmeister's Porpoise, Monitored From a Small-Scale Fishery Platform. *Frontiers in Marine Science*, 5(July), 2018.
- [57] G K Silber. Acoustic signals of the vaquita (Phocoena sinus). Aquatic Mammals, 17(3):130–133, 1991.

- [58] Songhai Li, Ding Wang, Kexiong Wang, Tomonari Akamatsu, Zhiqiang Ma, and Jiabo Han. Echolocation click sounds from wild inshore finless porpoise (*Neophocaena phocaenoides sunameri*) with comparisons to the sonar of riverine N. p. asiaeorientalis. The Journal of the Acoustical Society of America, 121(6):3938, 2007.
- [59] Songhai Li, Kexiong Wang, Ding Wang, and Tomonari Akamatsu. Echolocation signals of the free-ranging Yangtze finless porpoise (Neophocaena phocaenoides asiaeorientialis). The Journal of the Acoustical Society of America, 117(5):3288–3296, 2005.
- [60] Songhai Li, Kexiong Wang, Ding Wang, Shouyue Dong, and Tomonari Akamatsu. Simultaneous production of low- and high-frequency sounds by neonatal finless porpoises. *The Journal of the Acoustical Society of America*, 124(2):716–718, 2008.
- [61] William A Watkins, William E. Schevill, and Peter B. Best. Underwater Sounds of *Cephalorhynchus heavisidii* (Mammalia:Cetacea). *Journal of Mammalogy*, 58(3):316–320, 1977.
- [62] Morgan J. Martin, Simon H. Elwen, Reshma Kassanjee, and Tess Gridley. To buzz or burst-pulse? The functional role of Heaviside's dolphin, *Cephalorhynchus heavisidii*, rapidly pulsed signals. *Animal Behaviour*, 150:273–284, 2019.
- [63] Andrew J. Read. Porpoises, Overview. In Encyclopedia of Marine Mammals, pages 770–772. Elsevier Inc., third edit edition, 2018.

- [64] Maria Vanesa Reyes Reyes, Miguel A. Iñíguez Bessega, and Sarah J. Dolman. Review of legislation applied to seismic surveys to mitigate effects on marine mammals in Latin America. In Proceedings of Meetings on Acoustics. Fourth International Conference on the Effects of Noise on Aquatic Life, volume 27, page 032002, 2016.
- [65] Line Anker Kyhn, Frants Havmand Jensen, Kristian Beedholm, Jakob Tougaard, M Hansen, and Peter Teglberg Madsen. Echolocation in sympatric Peale's dolphins (*Lagenorhynchus australis*) and Commerson's dolphins (*Cephalorhynchus commersonii*) producing narrow-band high-frequency clicks. Journal of Experimental Biology, 213(11):1940–1949, 2010.
- [66] Ken Marten. Ultrasonic analysis of pygmy sperm whale (Kogia breviceps) and Hubbs' beaked whale (Mesoplodon carlhubbsi) clicks. Aquatic Mammals, 26(1):45–48, 2000.
- [67] Marta Jussara Cremer, Annelise Colin Holz, Pablo Bordino, Randall S. Wells, and Paulo César Simões-Lopes. Social sounds produced by franciscana dolphins, Pontoporia blainvillei (*Cetartiodactyla, Pontoporiidae*). The Journal of the Acoustical Society of America, 141(3):2047–2054, 2017.
- [68] T. Morisaka and R. C. Connor. Predation by killer whales (Orcinus orca) and the evolution of whistle loss and narrow-band high frequency clicks in odontocetes. Journal of Evolutionary Biology, 20(4):1439–1458, 2007.
- [69] Thomas A. Jefferson, P. J. Stacey, and Robin William Baird. A review of killer whale interactions with other marine mammals: predation to co-existence. *Mammal Review*, 21:151–180, 1991.
- [70] Charlotte Dunn and Diane Claridge. Killer whale (Orcinus orca) occurrence and predation in the Bahamas. Journal of the Marine Biological Association of the United Kingdom, 94(6):1305–1309, 2014.
- [71] A. Mel Cosentino. First record of Norwegian killer whales attacking and feeding on a harbour porpoise. *Marine Biodiversity Records*, 8:1–5, 2015.
- [72] Brian K. Branstetter, Judy St. Leger, Doug Acton, John Stewart, Dorian Houser, James J. Finneran, and Keith Jenkins. Killer whale (Orcinus orca) behavioral audiograms. The Journal of the Acoustical Society of America, 141(4):2387–2398, 2017.
- [73] Giovanni Bianucci and Walter Landini. Killer sperm whale: a new basal physeteroid (Mammalia, Cetacea) from the Late Miocene of Italy. Zoological Journal of the Linnean Society, 148(1):103–131, 2006.
- [74] Christina Lockyer. Harbour porpoises (Phocoena phocoena) in the North Atlantic: Biological parameters. NAMMCO Scientific Publications, 5:71–89, 2003.
- [75] Jennifer A. Learmonth, Sinéad Murphy, Patricia L. Luque, Robert J. Reid,
 I. Anthony P. Patterson, Andrew C. Brownlow, Harry M. Ross, Jason P. Barley, M. Begoña Santos, and Graham John Pierce. Life history of harbor porpoises (*Phocoena Phocoena*) in Scottish (UK) waters. *Marine Mammal Science*, 30(4):1427–1455, 2014.
- [76] Philip S. Hammond, Giovanni Bearzi, Arne Bjørge, K.A. Forney, L. Karczmarski, T. Kasuya, William F. Perrin, M.D. Scott, J.Y. Wang, Ran-

dall S. Wells, and Ben Wilson. *Phocoena phocoena* Baltic Sea subpopulation. The IUCN Red List of Threatened Species 2008: e.T17031A98831650. http://dx.doi.org/10.2305/IUCN.UK.2008.RLTS.T17031A6739565.en. Downloaded on 22 October 2018. Technical report, 2008.

- [77] Juan Jesús Bellido, Juan José Castillo, Miguel Ángel Farfá, Juan Jesús Martín, José Luis Mons, and Raimundo Real. Ejemplar enfermo de marsopa *Phocoena phocoena* (Linneaus, 1758) varado en las costas de Málaga. Galemys, 18(1-2):37–39, 2006.
- [78] Sam Blakeman, R Delgado, M Valdivia, Ruth Esteban, U Pena, Philippe Verborgh, Raul Vilela, and Juan Manuel Salazar-Sierra. Costa del Sol: A harbour porpoise (*Phocoena phocoena*) habitat in the Mediterranean Sea. 2019.
- [79] Philip S. Hammond, Per Berggren, Harald Benke, David L. Borchers, Anne Collet, Mads Peter Heide-Jørgensen, Sonja Heimlich, A.R. Hiby, Mardik F. Leopold, and Nils Øien. Abundance of harbour porpoise and other cetaceans in the North Sea and adjacent waters. *Journal of Applied Ecology*, 39:361–376, 2002.
- [80] SCANS II. Small Cetaceans in the European Atlantic and North Sea. Final Report to the European Commission on Project LIFE04NAT/GB/000245. *Population (English Edition)*, 44(0):0–54, 2008.
- [81] SCANS III. Estimates of cetacean abundance in European Atlantic waters in summer 2016 from the SCANS-III aerial and shipboard surveys. Technical report, SCANSS, 2017.

- [82] Ida Carlén, Len Thomas, Julia Carlström, Mats Amundin, Jonas Teilmann, Nick Tregenza, Jakob Tougaard, Jens C. Koblitz, Signe Sveegaard, Daniel Wennerberg, Olli Loisa, Michael Dähne, Katharina Brundiers, Monika Kosecka, Line Anker Kyhn, Cinthia Tiberi Ljungqvist, Iwona Pawliczka, Radomil Koza, Bartlomiej Arciszewski, Anders Galatius, Martin Jabbusch, Jussi Laaksonlaita, Jussi Niemi, Sami Lyytinen, Anja Gallus, Harald Benke, Penina Blankett, Krzysztof E. Skóra, and Alejandro Acevedo-Gutiérrez. Basin-scale distribution of harbour porpoises in the Baltic Sea provides basis for effective conservation actions. *Biological Conservation*, 226(May 2017):42– 53, 2018.
- [83] Meike Linnenschmidt, Jonas Teilmann, Tomonari Akamatsu, Rune Dietz, and Lee A. Miller. Biosonar, dive, and foraging activity of satellite tracked harbor porpoises (*Phocoena phocoena*). Marine Mammal Science, 29(2):E77–E97, 2013.
- [84] W. J. Strietman. Photo-identification of harbour porpoises in the Oosterschelde estuary, the Netherlands., 2012.
- [85] Cindy R. Elliser, Katrina H. MacIver, and Michelle Green. Group characteristics, site fidelity, and photo-identification of harbor porpoises, *Phocoena phocoena*, in Burrows Pass, Fidalgo Island, Washington. Marine Mammal Science, 34(2):365–384, 2018.
- [86] Michael A. Bigg, Graeme M. Ellis, and Kenneth C. Balcomb. The photographic identification of individual cetaceans. Whalewatcher 20(2):10-12. 1986., page 3, 1986.

- [87] Jonas Teilmann and Signe Sveegaard. Chapter 21 Porpoises the World Over: Diversity in Behavior and Ecology. In *Ethology and Behavioral Ecology of Marine Mammals*, pages 449–464. 2019.
- [88] Andrew J. Read and Andrew J. Westgate. Monitoring the movements of harbour porpoises (*Phocoena phocoena*) with satellite telemetry. *Marine Biology*, 130:315–322, 1997.
- [89] Cormac G. Booth, Clare B. Embling, Jonathan C. D. Gordon, Susannah V. Calderan, and Philip S. Hammond. Habitat preferences and distribution of the harbour porpoise *Phocoena phocoena* west of Scotland. *Marine Ecology Progress Series*, 478:273–285, 2013.
- [90] William Brown. Fine scale distribution and habitat use of harbour porpoises (Phocoena phocoena) in the Firth of Clyde. PhD thesis, University of St Andrews, 2018.
- [91] Signe Sveegaard. Spatial and temporal distribution of harbour porpoises in relation to their prey. PhD thesis, Aarhus University, 2011.
- [92] Joshua M. Lawrence, Eric Armstrong, Jonathan C. D. Gordon, Susan Maersk Lusseau, and Paul Fernandes. Passive and active, predator and prey: using acoustics to study interactions between cetaceans and forage fish. *ICES Journal of Marine Science*, 73(8):2075–2084, 2016.
- [93] Sven Koschinski. Current knowledge on harbour porpoises (*Phocoena pho-coena*) in the Baltic Sea. Ophelia, 55(3):167–197, 2001.

- [94] Nynne H Nielsen, Jonas Teilmann, Signe Sveegaard, Rikke G Hansen, Mikkelholger S Sinding, Rune Dietz, and Mads Peter Heide-jørgensen. Oceanic movements, site fidelity and deep diving in harbour porpoises from Greenland show limited similarities to animals from the North Sea. *Marine Ecology Progress Series*, 597:259–272, 2018.
- [95] F. H. Fisher and V. P. Simmons. Sound absorption in sea water. The Journal of the Acoustical Society of America, 62(3):558–564, 1977.
- [96] Lara Delgado García. Acoustic development and behaviour of Odontocete calves. PhD thesis, University of Southern Denmark, 2016.
- [97] Stephen M. Dawson, R. Ewan Fordyce, Sam H. Ridgway, Thomas E. Brough, and Elisabeth Slooten. Observations of a New Zealand dolphin (*Cephalorhynchus hectori*) breathing via its mouth. *Marine Mammal Science*, 33(1):350–355, 2017.
- [98] Lonneke L. IJsseldijk, Mardik F. Leopold, Elisa L.Bravo Rebolledo, Rob Deaville, Jan Haelters, Jooske IJzer, Paul D. Jepson, and Andrea Gröne. Fatal asphyxiation in two long-finned pilot whales (*Globicephala Melas*) caused by common Soles (Solea solea). *PLoS ONE*, 10(11):1–12, 2015.
- [99] Danuta Maria Wiśniewska, John M Ratcliffe, Kristian Beedholm, Christian B Christensen, Mark P. Johnson, Jens C. Koblitz, Magnus Wahlberg, and Peter Teglberg Madsen. Range-dependent flexibility in the acoustic field of view of echolocating porpoises (*Phocoena phocoena*). *eLife*, 4:e05651, 2015.
- [100] Stefan Huggenberger, Michael A. Rauschmann, Thomas J. Vogl, and Helmut H.A. Oelschläger. Functional morphology of the nasal complex in the

harbor porpoise (*Phocoena phocoena* L.). Anatomical Record, 292(6):902–920, 2009.

- [101] Kenneth J. Dormer. Mechanism of sound production and air recycling in delphinids: Cineradiographic evidence. The Journal of the Acoustical Society of America, 65:229–239, 1979.
- [102] Peter Teglberg Madsen, Marc O. Lammers, Danuta Maria Wisniewska, and Kristian Beedholm. Nasal sound production in echolocating delphinids (*Tur-siops truncatus* and *Pseudorca crassidens*) is dynamic, but unilateral: clicking on the right and whistling on the left side. *Journal of Experimental Biol-ogy*, 216:4091–4102, 2013.
- [103] William Keener, Marc A. Webber, Isidore D. Szczepaniak, Tim M. Markowitz, and Dara N. Orbach. The Sex Life of Harbor Porpoises (*Phocoena phocoena*): Lateralized and Aerial Behavior. *Aquatic Mammals*, 44(6):620–632, 2018.
- [104] Gísli A. Víkingsson, Droplaug Ólafsdóttir, and Jóhann Sigurjónsson. Geographical, and seasonal variation in the diet of harbour porpoises (*Phocoena phocoena*) in Icelandic coastal waters. NAMMCO Scientific Publications, 5:243–270, 2003.
- [105] Ursula K. Verfuss, Lee A. Miller, and Hans-Ulrich Schnitzler. Spatial orientation in echolocating harbour porpoises (*Phocoena Phocoena*). Journal of Experimental Biology, 208:3385–3394, 2005.
- [106] Jakob Tougaard, Jacob Carstensen, Jonas Teilmann, Henrik Skov, and Per Rasmussen. Pile driving zone of responsiveness extends beyond 20 km for

harbor porpoises (*Phocoena phocoena* (L.)). The Journal of the Acoustical Society of America, 126(1):11–14, 2009.

- [107] Andrew J. Wright, Tomonari Akamatsu, Jonas Teilmann, Kim Nørgaard Mouritsen, Signe Sveegaard, and Rune Dietz. Silent porpoise: potential sleeping behaviour identified in wild harbour porpoise. *Animal Behaviour*, 133:211– 222, 2017.
- [108] Danuta Maria Wiśniewska, Mark P. Johnson, Jonas Teilmann, Laia Rojano-Doñate, Jeanne Shearer, Signe Sveegaard, Lee A. Miller, Ursula Siebert, and Peter Teglberg Madsen. Ultra-high foraging rates of harbor porpoises make them vulnerable to anthropogenic disturbance. *Current Biology*, 26:1441–1446, 2016.
- [109] Tomonari Akamatsu, Jonas Teilmann, Lee A. Miller, Jakob Tougaard, Rune Dietz, Ding Wang, Kexiong Wang, Ursula Siebert, and Yasuhiko Naito. Comparison of echolocation behaviour between coastal and riverine porpoises. *Deep Sea Research Part II: Topical Studies in Oceanography*, 54:290–297, 2007.
- [110] Whitlow W.L. Au. The sonar of dolphins. Springer-Verlag, New York, 1993.
- [111] Michael Ladegaard and Peter Teglberg Madsen. Context-dependent biosonar adjustments during active target approaches in echolocating harbour porpoises. *The Journal of Experimental Biology*, page jeb.206169, 2019.
- [112] M. Begoña Santos, Graham John Pierce, Jennifer A. Learmonth, Robert J. Reid, Harry M. Ross, I. Anthony P. Patterson, D. G. Reid, and D. Beare. Variability in the diet of harbor porpoises (*Phocoena phocoena*) in Scottish waters 1992–2003. *Marine Mammal Science*, 20(1):1–27, 2004.

- [113] Stine D. Ross, Heidi Andreasen, and Niels G. Andersen. An important step towards accurate estimation of diet composition and consumption rates for the harbor porpoise (*Phocoena phocoena*). Marine Mammal Science, 32(4):1491– 1500, 2016.
- [114] Line Anker Kyhn, Jakob Tougaard, Len Thomas, Linda Rosager Duve, Joanna Stenback, Mats Amundin, Geneviève Desportes, and Jonas Teilmann. From echolocation clicks to animal density—Acoustic sampling of harbor porpoises with static dataloggers. *The Journal of the Acoustical Society of America*, 131(1):550, 2012.
- [115] Sacha Viquerat, Helena Herr, Anita Gilles, Verena Peschko, Ursula Siebert, Signe Sveegaard, and Jonas Teilmann. Abundance of harbour porpoises (*Phocoena Phocoena*) in the western Baltic, Belt Seas and Kattegat. *Marine Biology*, 161:745–754, 2014.
- [116] A.-C. Cucknell, O. Boisseau, Russell Leaper, R. Mclanaghan, and A. Moscrop. Harbour porpoise (*Phocoena Phocoena*) presence, abundance and distribution over the Dogger Bank, North Sea, in winter. *Journal of the Marine Biological Association of the United Kingdom*, 97(7):1455–1465, 2016.
- [117] Ulrich E. Stegmann. 00. Introduction: A primer on information and influence in animal communication. 2013.
- [118] Christophe A.H. Bousquet, David J.T. Sumpter, and Marta B. Manser. Moving calls: A vocal mechanism underlying quorum decisions in cohesive groups. *Proceedings of the Royal Society B: Biological Sciences*, 278(1711):1482–1488, 2011.

- [119] Gabriella E.C. Gall and Marta B. Manser. Group cohesion in foraging meerkats: Follow the moving 'vocal hot spot'. Royal Society Open Science, 4(4), 2017.
- [120] K. Schonrogge, F. Barbero, L. P. Casacci, J. Settele, and J. A. Thomas. Acoustic communication within ant societies and its mimicry by mutualistic and socially parasitic myrmecophiles. *Animal Behaviour*, 134:249–256, 2017.
- [121] Martin Heil and Richard Karban. Explaining evolution of plant communication by airborne signals. Trends in Ecology and Evolution, 25(3):137–144, 2010.
- [122] Zhuo Chen and John J Wiens. The origins of acoustic communication in vertebrates. *Nature Communications*, 11(369):1–8, 2020.
- [123] Whitlow W. L. Au and Mardi Hastings, editors. Principles of marine bioacoustics. Springer Science+Business Media, LLC, New York, 2008.
- [124] Thomas Buus Sørensen and Carl Christian Kinze. Reproduction and reproductive seasonality in danish harbour porpoises, *Phocoena phocoena*. Ophelia, 39(3):159–176, 1994.
- [125] Tina Kesselring, Sacha Viquerat, Ralph Brehm, and Ursula Siebert. Coming of age: - Do female harbour porpoises (*Phocoena phocoena*) from the North Sea and Baltic Sea have sufficient time to reproduce in a human influenced environment? *PloS one*, 12(10):e0186951, 2017.
- [126] A. D. Goodson and C. R. Sturtivant. Sonar characteristics of the harbour

porpoise (*Phocoena phocoena*): source levels and spectrum. *ICES Journal of* Marine Science, 53:465–472, 1996.

- [127] Nick C. Tregenza, Simon D. Berrow, Philip S. Hammond, and Russell Leaper. Harbour porpoise (*Phocoena Phocoena* L.) by-catch in set gillnets in the Celtic Sea. *ICES Journal of Marine Science*, 54:896–904, 1997.
- [128] Jacob Nabe-Nielsen, Richard M. Sibly, Jakob Tougaard, Jonas Teilmann, and Signe Sveegaard. Effects of noise and by-catch on a Danish harbour porpoise population. *Ecological Modelling*, 272:242–251, 2014.
- [129] Arne Bjørge, Mette Skern-Mauritzen, and Marjorie C. Rossman. Estimated bycatch of harbour porpoise (*Phocoena phocoena*) in two coastal gillnet fisheries in Norway, 2006-2008. Mitigation and implications for conservation. *Biological Conservation*, 161:164–173, 2013.
- [130] Andrew J. Wright, Marie Maar, Christian Mohn, Jacob Nabe-Nielsen, Ursula Siebert, Lasse Fast Jensen, Hans J. Baagøe, and Jonas Teilmann. Possible causes of a harbour porpoise mass stranding in Danish waters in 2005. *PLoS ONE*, 8(2):e55553, 2013.
- [131] Michael Dähne, Sven Adler, Kathrin Krügel, Ursula Siebert, Anita Gilles, Klaus Lucke, Verena Peschko, Janne Sundermeyer, Michael Dähne, Anita Gilles, Klaus Lucke, Verena Peschko, Sven Adler, Kathrin Krügel, Janne Sundermeyer, and Ursula Siebert. Effects of pile-driving on harbour porpoises (*Phocoena phocoena*) at the first offshore wind farm in Germany. *Environmental Research Letters*, 8(2):025002, 2013.

- [132] Alejandro Acevedo-Gutiérrez, Ann DiBerardinis, Shawn Larkin, Katrina Larkin, and Paul Forestell. Social interactions between tucuxis and bottlenose dolphins in Gandoca-Manzanillo, Costa Rica. Latin American Journal of Aquatic Mammals, 4(1):49–54, 2005.
- [133] Mariano A. Coscarella and Enrique A. Crespo. Feeding aggregation and aggressive interaction between bottlenose (*Tursiops truncatus*) and Commerson's dolphins (*Cephalorhynchus commersonii*) in Patagonia, Argentina. *Journal of Ethology*, 28:183–187, 2010.
- [134] E. Jourdain and D. Vongraven. Humpback whale (Megaptera novaeangliae) and killer whale (Orcinus orca) feeding aggregations for foraging on herring (Clupea harengus) in Northern Norway. Mammalian Biology - Zeitschrift für Säugetierkunde, 86:27–32, 2017.
- [135] Jochen R. Zaeschmar, Sarah L. Dwyer, and Karen A. Stockin. Rare observations of false killer whales (*Pseudorca crassidens*) cooperatively feeding with common bottlenose dolphins (*Tursiops truncatus*) in the Hauraki Gulf, New Zealand. *Marine Mammal Science*, 29(3):555–562, 2013.
- [136] Jean Baptiste Thiebot and Henri Weimerskirch. Contrasted associations between seabirds and marine mammals across four biomes of the southern Indian Ocean. Journal of Ornithology, 154:441–453, 2013.
- [137] Alexandros Frantzis and Denise L. Herzing. Mixed-species associations of striped dolphins (*Stenella coeruleoalba*), short-beaked common dolphins (*Del-phinus delphis*), and Risso's dolphins (*Grampus griseus*) in the Gulf of Corinth (Greece, Mediterranean Sea). Aquatic Mammals, 28(2):188–197, 2002.

- [138] Anoukchika D. Ilangakoon. Cetacean diversity and mixed-species associations off Southern Sri Lanka. In 7th International Symposium on SEASTAR2000 and Asian Bio-logging Science (The 11th SEASTAR2000 workshop), pages 23–28, 2012.
- [139] Jochen R. Zaeschmar, Ingrid N. Visser, Dagmar Fertl, Sarah L. Dwyer, Anna M. Meissner, Joanne Halliday, Jo Berghan, David Donnelly, and Karen A. Stockin. Occurrence of false killer whales (*Pseudorca crassidens*) and their association with common bottlenose dolphins (*Tursiops truncatus*) off northeastern New Zealand. *Marine Mammal Science*, 30(2):594–608, 2014.
- [140] Arda M Tonay, Ayhan Dede, and Ayaka Amaha Oztürk. An unusual interaction between bottlenose dolphins (*Tursiops truncatus*) and a harbour porpoise (*Phocoena phocoena*). Journal of Black Sea / Mediterranean Environment, 23(3):222–228, 2017.
- [141] Mark H Deakos, Brian K. Branstetter, Lori Mazzuca, Dagmar Fertl, and Joseph R Mobley. Two unusual interactions between a bottlenose dolphin (*Tursiops truncatus*) and a humpback whale (*Megaptera novaeangliae*) in Hawaiian waters. Aquatic Mammals, 36(2):121–128, 2010.
- [142] Conor Ryan, Greig Macleod, Cameron Dinsdale, and Stuart Cook. Longterm association between a solitary common dolphin (*Delphinus delphis*) and a harbour porpoise (*Phocoena phocoena*). Aquatic Mammals, 43(1):113–115, 2017.
- [143] Harry M. Ross and Ben Wilson. Violent interactions between bottlenose dol-

phins and harbour porpoises. *Proceedings: Biological Sciences*, 263(1368):283–286, 1996.

- [144] Rocío Espada, Liliana Olaya-Ponzone, Luisa Haasova, Estefanía Martín, and José C. García-Gómez. Hybridization in the wild between *Tursiops truncatus* (Montagu 1821) and Delphinus delphis (Linnaeus 1758). *PLoS ONE*, 14(4):1– 16, 2019.
- [145] A. Cañadas, R. Sagarminaga, and S. García-Tiscar. Cetacean distribution related with depth and slope in the Mediterranean waters off southern Spain. *Deep-Sea Research Part I: Oceanographic Research Papers*, 49(11):2053–2073, 2002.
- [146] Karin A. Forney and Paul R. Wade. Worldwide Distribution and Abundance of Killer Whales. In Whales, Whaling, and Ocean Ecosystems, pages 145–162. 2007.
- [147] John K. B. Ford, Graeme M. Ellis, Lance G. Barrett-Lennard, Alexandra B. Morton, and Kenneth C. Balcomb. Dietary specialization in two sympatric populations of killer whales (*Orcinus orca*) in coastal British Columbia and adjacent waters. *Canadian Journal of Zoology*, 76:1456–1471, 1998.
- [148] Andrew D. Foote, Phillip A. Morin, John W. Durban, Eske Willerslev, Ludovic Orlando, and M. Thomas P. Gilbert. Out of the pacific and back again: Insights into the matrilineal history of pacific killer whale ecotypes. *PLoS ONE*, 6(9), 2011.
- [149] Ruth Esteban, Philippe Verborgh, Pauline Gauffier, Joan Giménez, Christophe Guinet, and Renaud de Stephanis. Dynamics of killer whale,

bluefin tuna and human fisheries in the Strait of Gibraltar. *Biological Con*servation, 194:31–38, 2016.

- [150] Philip S. Hammond, Kelly Macleod, Per Berggren, David L. Borchers, Louise Burt, Ana Cañadas, Geneviève Desportes, Greg P. Donovan, Anita Gilles, Douglas Gillespie, Jonathan C. D. Gordon, Lex Hiby, Iwona Kuklik, Russell Leaper, Kristina Lehnert, Mardik F. Leopold, Phil Lovell, Nils Øien, Charles G M Paxton, Vincent Ridoux, Emer Rogan, Filipa Samarra, Meike Scheidat, Marina Sequeira, Ursula Siebert, Henrik Skov, René J. Swift, Mark L. Tasker, Jonas Teilmann, Olivier Van Canneyt, and José Antonio Vázquez. Cetacean abundance and distribution in European Atlantic shelf waters to inform conservation and management. *Biological Conservation*, 164:107–122, 2013.
- [151] Carl Christian Kinze. Spækhugger Orcinus orca. In Hans J. Baagøe and T. Jensen, editors, Dansk Pattedyratlas, page 392. Copenhagen: Gyldendal, 2007.
- [152] Thibaut Bouveroux, Jeremy J. Kiszka, Michael R. Heithaus, Thierry Jauniaux, and Sylvain Pezeril. Direct evidence for gray seal (*Halichoerus grypus*) predation and scavenging on harbor porpoises (*Phocoena phocoena*). Marine Mammal Science, 30(4):1542–1548, 2014.
- [153] Thierry Jauniaux, Mutien-Marie Garigliany, Pauline Loos, Jean-Luc Bourgain, Thibaut Bouveroux, Freddy Coignoul, Jan Haelters, Jacky Karpouzopoulos, Sylvain Pezeril, and Daniel Desmecht. Bite injuries of grey seals (*Halichoerus grypus*) on harbour porpoises (*Phocoena phocoena*</i>). *PLoS* ONE, 9(12):e108993, 2014.

- [154] Thomas Stringell, Dave Hill, Dafydd Rees, Ffion Rees, Padrig Rees, Greg Morgan, Lisa Morgan, and Ceri Morris. Predation of harbour porpoises (*Phocoena phocoena*) by grey seals (*Halichoerus grypus*) in wales. Aquatic Mammals, 41(2):188–191, 2015.
- [155] Jan Haelters, Eligius Everaarts, Paulien Bunskoek, Lineke Begeman, John W J Hinrichs, and Lonneke L. IJsseldijk. A suspected scavenging event by red foxes (*Vulpes vulpes*) on a live, stranded harbour porpoise (*Phocoena phocoena*). Aquatic Mammals, 42(2):227–232, 2016.
- [156] Glenn A. Bowen. Halichoerus grypus. The IUCN Red List of Threatened Species 2016. e.T9660A45226042. 8235, 2016.
- [157] Mark P. Cotter, Daniela Maldini, and Thomas A. Jefferson. "Porpicide" in California: Killing of harbor porpoises (*Phocoena phocoena*) by coastal bottlenose dolphins (*Tursiops truncatus*). Marine Mammal Science, 28(1):E1– E15, 2012.
- [158] Andrew C. Brownlow, Nicholas J. Davison, and Mariel T. I. ten Doeschate. Annual Report 2015 - 1 January to 31 December 2015, for Marine Scotland, Scottish Government. Technical report, Inverness, 2015.
- [159] Robin William Baird. An interaction between Pacific white-sided dolphins and a neonatal harbor porpoise. *Mammalia*, 62(1):129–134, 1998.
- [160] Sylvain Larrat, Lena N. Measures, and Stéphane Lair. Rake marks on a harbor porpoise (*Phocoena phocoena*) calf suggestive of a fatal interaction with an atlantic white-sided dolphin (*Lagenorhynchus acutus*). Aquatic Mammals, 38(1):86–91, 2012.

- [161] Thomas A. Jefferson. Dall's Porpoise. In Encyclopedia of Marine Mammals, pages 239–242. 2018.
- [162] Pamela M. Willis, Bernard J. Crespi, Lawrence M. Dill, Robin W. Baird, and M. Bradley Hanson. Natural hybridization between Dall's porpoises (*Phocoenoides dalli*) and harbour porpoises (*Phocoena phocoena*). Canadian Journal of Zoology, 82(5):828–834, 2004.
- [163] Kevin P. Robinson, Sonja M. Eisfeld, Marina Costa, and Mark P. Simmonds. Short-beaked common dolphin (*Delphinus Delphis*) occurrence in the Moray Firth, north-east Scotland. *Marine Biodiversity Records*, 3(May 2014), 2010.
- [164] Caroline R. Weir, Colin D. MacLeod, and Susannah V. Calderan. Fine-scale habitat selection by white-beaked and common dolphins in the Minch (Scotland, UK): Evidence for interspecific competition or coexistence? *Journal* of the Marine Biological Association of the United Kingdom, 89(5):951–960, 2009.
- [165] E. Elizabeth Henderson, John A. Hildebrand, M. H. Smith, and Erin A. Falcone. The behavioral context of common dolphin (*Delphinus sp.*) vocalizations. *Marine Mammal Science*, 28(3):439–460, 2012.
- [166] Ina Christiane Ansmann, John C. Goold, Peter G. H. Evans, Mark Peter Simmonds, and Simon G. Keith. Variation in the whistle characteristics of short-beaked common dolphins, *Delphinus delphis*, at two locations around the British Isles. *Journal of the Marine Biological Association of the UK*, 87:19–26, 2007.

- [167] B. S. Gurevich. Echolocation discrimination of geometric between figures in the dolphin, *Delphinus delphis. oskovskogo Universiteta*, *Biologiya*, *Pochovedeniye*, 3:109–112, 1969.
- [168] James F Fish and Charles W. Turl. Acoustic source levels of four species of small whales. Technical Report December, NUC, San Diego, CA, 1976.
- [169] Whitlow W. L. Au and Marc O. Lammers, editors. Introduction: listening in the ocean. Springer Science+Business Media, LLC, 2016.
- [170] Melissa S. Soldevilla, E. Elizabeth Henderson, Gregory S. Campbell, Sean M. Wiggins, John A. Hildebrand, and Marie A. Roch. Classification of Risso's and Pacific white-sided dolphins using spectral properties of echolocation clicks. The Journal of the Acoustical Society of America, 124(1):609–624, 2008.
- [171] K. J. Palmer, Kate L. Brookes, and Luke Rendell. Categorizing click trains to increase taxonomic precision in echolocation click loggers. *The Journal of the Acoustical Society of America*, 142(2):863–877, 2017.
- [172] Marie A. Roch, Holger Klinck, Simone Baumann-Pickering, David K. Mellinger, Simon Qui, Melissa S. Soldevilla, and John A. Hildebrand. Classification of echolocation clicks from odontocetes in the Southern California Bight. The Journal of the Acoustical Society of America, 129(1):467–475, 2011.
- [173] Lissa Goodwin and Margaux Dodds. Long rangers: A report on solitary dolphins and whales, including recommendations for their protection. Technical report, Marine Connection, 2008.

- [174] Tilen Genov, Giovanni Bearzi, Silvia Bonizzoni, and Milena Tempesta. Longdistance movement of a lone short-beaked common dolphin *Delphinus delphis* in the central Mediterranean Sea. *Marine Biodiversity Records*, 5:e9, 2012.
- [175] Laetitia Nunny and Mark P. Simmonds. A Global Reassessment of Solitary-Sociable Dolphins. Frontiers in Veterinary Science, 5(January):1–16, 2019.
- [176] Christina Lockyer. The history and behaviour of a solitary wild, but sociable, bottlenose dolphin (*Tursiops truncatus*) on the west coast of England and Wales. *Journal of Natural History*, 12(5):513–528, 1978.
- [177] Monika Wilke, Mike Bossley, and Wade Doak. Managing Human Interactions with Solitary Dolphins. Aquatic Mammals, 31(4):427–433, 2005.
- [178] Bruno Díaz López, Julia Andrea Bernal Shirai, Alberto Bilbao Prieto, and Paula Méndez Fernández. Diving activity of a solitary wild free ranging bottlenose dolphin (*Tursiops truncatus*). Journal of the Marine Biological Association of the United Kingdom, 88(6):1153–1157, 2008.
- [179] C M Heyes. Social learning in animals: categories and mechanisms. Biological Reviews, 69:207–231, 1994.
- [180] Vincent M. Janik and Peter J.B. Slater. Vocal Learning in Mammals, volume 26. Elsevier Masson SAS, 1997.
- [181] Vincent M. Janik and Peter J. B. Slater. The different roles of social learning in vocal communication. *Animal Behaviour*, 60:1–11, 2000.
- [182] D G Richards, J P Wolz, and L M Herman. Vocal mimicry of computer-

generated sounds and vocal labeling of objects by a bottlenosed dolphin, *Tursiops truncatus*. Journal of Comparative Psychology, 98(1):10–28, 1984.

- [183] José Z. Abramson, M^a Victoria Hernández-Lloreda, Lino García, Fernando Colmenares, Francisco Aboitiz, and Josep Call. Imitation of novel conspecific and human speech sounds in the killer whale (*Orcinus orca*). Proc. R. Soc. B, 285(1871):20172171, 2018.
- [184] Amanda L. Stansbury and Vincent M. Janik. Formant Modification through Vocal Production Learning in Gray Seals. *Current Biology*, 29(13):2244– 2249.e4, 2019.
- [185] Sam Ridgway, Donald Carder, Michelle Jeffries, and Mark Todd. Spontaneous human speech mimicry by a cetacean. *Current Biology*, 22(20):R860–R861, 2012.
- [186] Katherine Ralls, Patricia Fiorelli, and Sherie Gish. Vocalizations and vocal mimicry in captive harbor seals, *Phoca Vitulina*. *Canadian Journal of Zoology*, 62(5):1050–1056, 1985.
- [187] Whitney B. Musser, Ann E. Bowles, Dawn M. Grebner, and Jessica L. Crance. Differences in acoustic features of vocalizations produced by killer whales crosssocialized with bottlenose dolphins. *The Journal of the Acoustical Society of America*, 136(4):1990–2002, 2014.
- [188] Elena M. Panova and Alexandr V. Agafonov. A beluga whale socialized with bottlenose dolphins imitates their whistles. *Animal Cognition*, 20:1153–1160, 2017.

- [189] Vincent M. Janik. Cetacean vocal learning and communication. Current Opinion in Neurobiology, 28:60–65, 2014.
- [190] Andrew D. Foote, Rachael M. Rachel M. Griffin, David Howitt, Lisa Larsson, Patrick J. O. Miller, and A. Rus Hoelzel. Killer whales are capable of vocal learning. *Biology Letters*, 2(4):509–512, 2006.
- [191] Peter L. Tyack. Convergence of calls as animals form social bonds, active compensation for noisy communication channels, and the evolution of vocal learning in mammals. *Journal of Comparative Psychology*, 122(3):319–331, 2008.
- [192] Michael A. Bigg, Ian MacAskie, and Graeme M. Ellis. Photo-identification of Individual Killer Whales, 1983.
- [193] W. A. Watkins and W. E. Schevill. Sound source location by arrival-times on a non-rigid three-dimentional hydrophone array. *Deep Sea Research*, 19:691– 706, 1972.
- [194] Renata S. Sousa-Lima, Thomas F. Norris, Julie N. Oswald, and Deborah P. Fernandes. A review and inventory of fixed autonomous recorders for passive acoustic monitoring of marine mammals. *Aquatic Mammals*, 39(1):23–53, 2013.
- [195] Tina M. Yack, Jay Barlow, John Calambokidis, Brandon L. Southall, and Shannon Coates. Passive acoustic monitoring using a towed hydrophone array results in identification of a previously unknown beaked whale habitat. The Journal of the Acoustical Society of America, 134(3):2589–2595, 2013.

- [196] Bjarke Klit Nielsen and Bertel Møhl. Hull-mounted hydrophones for passive acoustic detection and tracking of sperm whales (*Physeter macrocephalus*). *Applied Acoustics*, 67(11-12):1175–1186, 2006.
- [197] Russell Leaper, Douglas Gillespie, and Vassili Papastavrou. Results of passive acoustic surveys for odontocetes in the Southern Ocean. Journal of Cetacean Research and Management, 2(3):187–196, 2000.
- [198] René J. Swift, Gordon D. Hastie, Tim R. Barton, Christopher W. Clark, Mark L. Tasker, and Paul M Thompson. Passive acoustic monitoring of whale distribution in the Faroe-Shetland Channel. Technical report, Atlantic Frontier Environmental Network, 2004.
- [199] Marine Conservation Research. Final report of a survey for harbour porpoises conducted from R/V Song of the whale in French and British waters of the English Channel, May to June 2011. Technical report, IFAW, Essex, 2011.
- [200] Philip S. Hammond, Douglas Gillespie, Phil Lovell, Filipa Samarra, René J. Swift, Kelly Macleod, Mark L. Tasker, Per Berggren, David L. Borchers, Louise Burt, Charles G M Paxton, Ana Cañadas, Geneviève Desportes, Greg P. Donovan, Anita Gilles, Kristina Lehnert, Ursula Siebert, Jonathan C. D. Gordon, Russell Leaper, Mardik F. Leopold, Meike Scheidat, Nils Øien, Vincent Ridoux, Emer Rogan, Henrik Skov, Jonas Teilmann, Olivier Van Canneyt, and José Antonio Vázquez. Large scale surveys for cetaceans: Line transect assumptions, reliability of abundance estimates and improving survey efficiency A response to MacLeod. *Biological Conservation*, 170:338–339, 2014.

- [201] Hanna K. Nuuttila, Len Thomas, Jan G. Hiddink, Rhiannon Meier, John R. Turner, James D. Bennell, Nick J. C. Tregenza, and Peter G. H. Evans. Acoustic detection probability of bottlenose dolphins, *Tursiops truncatus*, with static acoustic dataloggers in Cardigan Bay, Wales. *The Journal of the Acoustical Society of America*, 134(3):2596–609, 2013.
- [202] Julia A. K. Carlström. Diel variation in echolocation behavior of wild harbor porpoises. Marine Mammal Science, 21(1):1–12, 2005.
- [203] Jacob Carstensen, Oluf Damsgaard Henriksen, and Jonas Teilmann. Impacts of offshore wind farm construction on harbour porpoises: Acoustic monitoring of echolocation activity using porpoise detectors (T-PODs). Marine Ecology Progress Series, 321:295–308, 2006.
- [204] Chloë Malinka. Evaluation of a drifting porpoise localising array buoy: A novel configuration for tracking porpoises in tidal rapids using passive acoustics. Technical report, 2009.
- [205] Jamie Macaulay, Jamie Macaulay, Jonathan C. D. Gordon, Douglas Gillespie, Chloë Malinka, Mark P. Johnson, and Simon P. Northridge. Appendix 1 : Constructing the KE Buoy. (JANUARY 2016), 2015.
- [206] Mark P. Johnson and Peter L. Tyack. A digital acoustic recording tag for measuring the response of wild marine mammals to sound. *IEEE Journal of Oceanic Engineering*, 28(1):3–12, 2003.
- [207] Satoko Kimura, Tomonari Akamatsu, Lijun Dong, Kexiong Wang, Ding Wang, Yasutoki Shibata, and Nobuaki Arai. Acoustic capture-recapture method for

towed acoustic surveys of echolocating porpoises. The Journal of the Acoustical Society of America, 135(6):3364–70, 2014.

- [208] Tomonari Akamatsu, Ding Wang, Kexiong Wang, and Yasuhiko Naito. Biosonar behaviour of free-ranging porpoises. Proceedings of the Royal Society B: Biological Sciences, 272:797–801, 2005.
- [209] Paul M. Thompson, Kate L. Brookes, Isla M. Graham, Tim R. Barton, Keith Needham, Gareth Bradbury, and Nathan D. Merchant. Short-term disturbance by a commercial two-dimensional seismic survey does not lead to longterm displacement of harbour porpoises. *Proceeding of the Royal Society B*, 280:20132001, 2013.
- [210] Aaron Thode. Tracking sperm whale (*Physeter Macrocephalus*) dive profiles using a towed passive acoustic array. The Journal of the Acoustical Society of America, 116(1):245–253, 2004.
- [211] Nick J. C. Tregenza. A new tool for cetacean research a fully submersible click detector. In *European Research on Cetaceans*, volume 13, page 473, Valencia, Spain, 1999.
- [212] Andrew J. Wright and Nick Tregenza. CPOD successful in trial for detecting Māui dolphin outside harbours. New Zealand Journal of Marine and Freshwater Research, 53(3):451–459, 2019.
- [213] Ruth H. Leeney, David Carslake, and Simon H. Elwen. Using static acoustic monitoring to describe echolocation behaviour of heaviside's dolphins (*Cephalorhynchus heavisidii*) in Namibia. *Aquatic Mammals*, 37(2):151–160, 2011.

- [214] Lorenzo Rojas-Bracho, Armando Jaramillo-Legorreta, Gustavo Cardenas-Hinojosa, Edwyna Nieto-Garcia, Paloma Ladron de Guevara, Barbara L. Taylor, Jay Barlow, Tim Gerrodette, Annette Henry, Nick J. C. Tregenza, René J. Swift, and Tomonari Akamatsu. Assessing trends in abundance for vaquita using acoustic monitoring: within refuge plan and outside refuge research needs. Workshop report—October 19–23, 2009. Technical memorandum NOAA-TM-NMFS-SWFSC-459. National Oceanic and Atmospheric Administration, L. Technical Report May, 2010.
- [215] Ursula K. Verfuß, Christopher G. Honnef, and Harald Benke. Seasonal and geographical variation of harbour porpoise (*Phocoena phocoena*) habitat use in the German Baltic Sea monitored by passive acoustic methods (PODs). *Progress in Marine Conservation in Europe: NATURA 2000 Sites in German Offshore Waters*, (2007):209–224, 2006.
- [216] Enrico Pirotta, Kate L. Brookes, Isla M. Graham, and Paul M. Thompson. Variation in harbour porpoise activity in response to seismic survey noise. *Biology letters*, 10:20131090, 2014.
- [217] Joanna Sarnocinska, Jakob Tougaard, Mark P. Johnson, Peter Teglberg Madsen, and Magnus Wahlberg. Comparing the performance of C-PODs and SoundTrap/PAMGUARD in detecting the acoustic activity of harbor porpoises (*Phocoena phocoena*). In *Proceedings of Meetings on Acoustics. Fourth International Conference on the Effects of Noise on Aquatic Life*, volume 27, page 070013, 2016.
- [218] Karin Tubbert Clausen, Jakob Tougaard, Jacob Carstensen, Matthieu Dele-

fosse, and Jonas Teilmann. Noise affects porpoise click detections – the magnitude of the effect depends on logger type and detection filter settings. *Bioacoustics*, 4622(May):1–16, 2018.

- [219] Nick J. C. Tregenza, Steve Dawson, William Rayment, and Ursula K. Verfuß. Listening to Echolocation Clicks with PODs. In *Listening in the Ocean*, chapter 7, pages 163–210. 2016.
- [220] Jakob Tougaard. Signal detection theory, detectability and stochastic resonance effects. *Biological Cybernetics*, 87:79–90, 2002.
- [221] Douglas Gillespie, Jonathan C. D. Gordon, Ron Mchugh, David Mclaren, David K. Mellinger, Paul Redmond, Aaron Thode, Phil Trinder, and Xiao Yan Deng. PAMGUARD: Semiautomated, open source software for real-time acoustic detection and localisation of cetaceans, 2008.
- [222] JNCC. JNCC guidelines for minimising the risk of injury and disturbance to marine mammals from seismic surveys, 2010.
- [223] Chloe E Malinka, Douglas Gillespie, Jamie D. J. Macaulay, Ruth Joy, and Carol E Sparling. First in situ passive acoustic monitoring for marine mammals during operation of a tidal turbine in Ramsey Sound, Wales. *Marine Ecology Progress Series*, 590:247–266, 2018.
- [224] Douglas Gillespie and Oliver Chappell. An automatic system for detecting and classifying the vocalisations of harbour porpoises. *Bioacoustics*, 13:37–61, 2002.

- [225] Russell Leaper, Oliver Chappell, and Jonathan C. D. Gordon. The development of practical techniques for surveying sperm whale populations acoustically. 1992.
- [226] Douglas Gillespie. An acoustic survey for sperm whales in the Southern Ocean Sancturay conducted from the RSV Aurora australis. *Report of the International Whaling Commission*, 47:897–907, 1997.
- [227] Shannon Rankin, Frederick I. Archer, Jennifer L. Keating, Julie N. Oswald, Michael Oswald, Alex Curtis, and Jay Barlow. Acoustic classification of dolphins in the California Current using whistles, echolocation clicks, and burst pulses. *Marine Mammal Science*, 33(2):520–540, 2017.
- [228] Julie N. Oswald, Shannon Rankin, Jay Barlow, and Marc O. Lammers. A tool for real-time acoustic species identification of delphinid whistles. *The Journal* of the Acoustical Society of America, 122(1):587–595, 2007.
- [229] Julie N. Oswald, Jay Barlow, and Thomas F Norris. Acoustic identification of nine delphinid species in the eastern tropical Pacific Ocean. Marine Mammal Science, 19(1):20–37, 2003.
- [230] Bianca Romeu, Mauricio Cantor, Carolina Bezamat, Paulo C. Simões-Lopes, and Fábio G. Daura-Jorge. Bottlenose dolphins that forage with artisanal fishermen whistle differently. *Ethology*, pages 1–10, 2017.
- [231] Paulo C. Simões-Lopes, Marta E. Fabián, and João O. Menegheti. Dolphin interactions with the mullet artisanal fishing on Southern Brazil: a qualitative and quantitative approach. *Revista Brasileira de Zoologia*, 15(3):709–726, 1998.

- [232] Whitlow W. L. Au, Brian K. Branstetter, Patrick W. Moore, and James J. Finneran. Dolphin biosonar signals measured at extreme off-axis angles: Insights to sound propagation in the head. *The Journal of the Acoustical Society* of America, 132(2):1199–1206, 2012.
- [233] Paul Lepper, Noel Dümortier, Kathleen Dudzinski, and Sekharjit Datta. Separation of Complex Echolocation Signal ' Trains ' From Multiple Bio-Sonar Sources. In Proceedings of the International Conference "Underwater Acoustic Measurements: Technologies & Results" Heraklion, Crete, Greece, 28th June – 1st July 2005, pages 1–6, 2005.
- [234] Josefin Starkhammar, Johan Nilsson, Mats Amundin, Stan A. Kuczaj, Monica Almqvist, and Hans W. Persson. Separating overlapping click trains originating from multiple individuals in echolocation recordings. *The Journal of the Acoustical Society of America*, 129:458–466, 2011.
- [235] Whitlow W. L. Au, Ronald A. Kastelein, Kelly J. Benoit-Bird, Ted W. Cranford, and Megan F. McKenna. Acoustic radiation from the head of echolocating harbor porpoises (*Phocoena phocoena*). The Journal of Experimental Biology, 209:2726–2733, 2006.
- [236] CODA. Cetacean Offshore Distribution and Abundance in the European Atlantic (CODA). Technical report, CODA, 2009.
- [237] Lissa Goodwin and C. Speedie. Relative abundance, density and distribution of the harbour porpoise (*Phocoena phocoena*) along the west coast of the UK. Journal of the Marine Biological Association of the United Kingdom, 88(6):1221–1228, 2008.

- [238] Peter Teglberg Madsen and Magnus Wahlberg. Recording and quantification of ultrasonic echolocation clicks from free-ranging toothed whales. *Deep-Sea Research Part I: Oceanographic Research Papers*, 54:1421–1444, 2007.
- [239] Signe Sveegaard, Jonas Teilmann, Jakob Tougaard, Rune Dietz, Kim Nørgaard Mouritsen, Geneviève Desportes, and Ursula Siebert. High-density areas for harbor porpoises (*Phocoena phocoena*) identified by satellite tracking. *Marine Mammal Science*, 27(1):230–246, 2011.
- [240] Douglas Gillespie and M Oswald. Binary file structure for PAMGUARD detector output. Version 4. 0 1 Introduction 2 File Format, 2017.
- [241] Peter Teglberg Madsen, I. Kerr, and Roger Payne. Source parameter estimates of echolocation clicks from wild pygmy killer whales (*Feresa attenuata*) (L). *The Journal of the Acoustical Society of America*, 116(4):1909–1912, 2004.
- [242] JNCC. JNCC guidelines for minimising the risk of injury to marine mammals from using explosives, 2010.
- [243] Mohamad Adam Bujang, Nadiah Sa'At, Tg Mohd Ikhwan Tg Abu Bakar Sidik, and Chien Joo Lim. Sample size guidelines for logistic regression from observational studies with large population: Emphasis on the accuracy between statistics and parameters based on real life clinical data. *Malaysian Journal of Medical Sciences*, 25(4):122–130, 2018.
- [244] Gerry Quinn and Michael Keough. Experimental Design and Data Analysis for Biologists. 2002.

- [245] J. P. Egan. Signal detection theory and ROC analysis. Academic Press, New York, 1975.
- [246] Danuta Maria Wisniewska, Mark Johnson, Jonas Teilmann, Ursula Siebert, Anders Galatius, Rune Dietz, and Peter Teglberg Madsen. High rates of vessel noise disrupt foraging in wild harbour porpoises (*Phocoena phocoena*). *Proceedings of the Royal Society B: Biological Sciences*, 285(1872):20172314, 2018.
- [247] Amands K. Holdman, Joseph H. Haexl, Holger Klinck, and Leigh G. Torres. Acoustic monitoring reveals the times and tides of harbor porpoise (*Phocoena phocoena*) distribution off central Oregon, U.S.A. *Marine Mammal Science*, 35(January):164–186, 2018.
- [248] B J P Berges, S. C. V. Geelhoed, Meike Scheidat, and Jakob Tougaard. Quantifying harbour porpoise foraging behaviour in CPOD data: identification, automatic detection and potential application. 2019.
- [249] Maddalena Bearzi. Dolphin sympatric ecology. Marine Biology Research, 1(3):165–175, 2005.
- [250] Andrew Scullion. Short-beaked common dolphin, Delphinus delphis, whistles: whistle density, a reliable form of measuring group size? PhD thesis, University of Wales, 2004.
- [251] Joseph L. Fleiss. Measuring nominal scale agreement among many raters. Psychological Bulletin, 76(5):378–382, 1971.

- [252] G. Cardillo. Fleiss'es kappa: compute the Fleiss'es kappa for multiple raters. http://www.mathworks.com/matlabcentral/fileexchange/15426, 2007.
- [253] D. Lusseau. Effects of tour boats on the behaviour of bottlenose dolphins: Using Markov chains to model anthropogenic impacts. *Conservation Biology*, 17(6):1785–1793, 2003.
- [254] Ronald A. Kastelein, Lean Hoek, Christ A. F. de Jong, and Paul J. Wensveen. The effect of signal duration on the underwater detection thresholds of a harbor porpoise (*Phocoena phocoena*) for single frequency-modulated tonal signals between 0.25 and 160 kHz. *The Journal of the Acoustical Society of America*, 128(5):3211–3222, 2010.
- [255] V V Popov and V O Klishin. EEG study of hearing in the common dolphin, Delphinus delphis, 1998.
- [256] Livio Favaro, Silvana Neves, Stefano Furlati, Daniela Pessani, Vidal Martin, and Vincent M. Janik. Evidence suggests vocal production learning in a crossfostered Risso's dolphin (*Grampus griseus*). Animal Cognition, 19(4):847–853, 2016.
- [257] Melissa S. Soldevilla. Risso's and Pacific white-sided dolphins in the Southern California Bight: using echolocation clicks to study dolphin ecology. PhD thesis, 2008.
- [258] Tyler M. Schulz, Hal Whitehead, Shane Gero, and Luke Rendell. Overlapping and matching of codas in vocal interactions between sperm whales: insights into communication function. *Animal Behaviour*, 76(6):1977–1988, 2008.

- [259] Simone Pika, Ray Wilkinson, Kobin H. Kendrick, and Sonja C. Vernes. Taking turns: bridging the gap between human and animal communication. *Proceed*ings of the Royal Society B: Biological Sciences, 285:20180598, 2018.
- [260] Vyacheslav A. Ryabov. The study of acoustic signals and the supposed spoken language of the dolphins. St. Petersburg Polytechnical University Journal: Physics and Mathematics, 2(3):231–239, 2016.
- [261] Harvey Sacks, Emanuel A Schegloff, and Gail Jefferson. A Simplest Systematics for the Organization of Turn-Taking for Conversation. *Source: Language*, 50(1):696–735, 1974.
- [262] J. R. Krebs and Richard Dawkins. Animal signals: Mind-Reading and Manipulation. In *Behavioural Ecology: An evolutionary approach*, pages 380–404. 1978.
- [263] François Beauducel. SUNRISE: sunrise and sunset times (https://www.mathworks.com/matlabcentral/fileexchange/64692-sunrisesunrise-and-sunset-times), MATLAB Central File Exchange. Retrieved October 17, 2019. 2019.
- [264] Simon Elwen, Michael A. Meÿer, Peter B. Best, P. G H. Kotze, Meredith Thornton, and Stephan Swanson. Range and Movements of Female Heaviside'S Dolphins (Cephalorhynchus Heavisidii), As Determined By Satellite-Linked Telemetry. *Journal of Mammalogy*, 87(5):866–877, 2006.
- [265] Simon H Elwen and K. Gopal. Cephalorhynchus heavisidii, Heaviside's Dolphin. The IUCN Red List of Threatened Species 2018: e.T4161A50352086. Technical report, 2018.

[266] Ina Christiane Ansmann. The whistle repertoire and acoustic behaviour of short-beaked common dolphins, Delphinus delphis, around the British Isles, with applications for acoustic surveying. PhD thesis, University of Wales, 2005.

Appendix A

CrossMark

Porpoise click classifier (PorCC): A high-accuracy classifier to study harbour porpoises (*Phocoena phocoena*) in the wild

Mel Cosentino,^{1,a)} Francesco Guarato,¹ Jakob Tougaard,² David Nairn,³ Joseph C. Jackson,¹ and James F. C. Windmill¹

¹Bioacoustics Group, Centre for Ultrasonic Engineering, Department of Electronic and Electrical Engineering, University of Strathclyde, 99 George Street, G1 1RD Glasgow, United Kingdom ²Department of Bioscience, Aarhus University, Frederiksborgvej 399, 4000 Roskilde, Denmark ³Clyde Porpoise CIC, 1-1 Allanton Park Terrace, KA29 0AW Fairlie, United Kingdom

(Received 10 January 2019; revised 2 May 2019; accepted 20 May 2019; published online 12 June 2019)

Harbour porpoises are well-suited for passive acoustic monitoring (PAM) as they produce highly stereotyped narrow-band high-frequency (NBHF) echolocation clicks. PAM systems must be coupled with a classification algorithm to identify the signals of interest. Here, the authors present a harbour porpoise click classifier (PorCC) developed in MATLAB, which uses the coefficients of two logistic regression models in a decision-making pathway to assign candidate signals to one of three categories: high-quality clicks (HQ), low-quality clicks (LQ), or high-frequency noise. The receiver operating characteristics of PorCC was compared to that of PAMGuard's Porpoise Click Detector/ Classifier Module. PorCC outperformed PAMGuard's classifier achieving higher hit rates (correctly classified clicks) and lower false alarm levels (noise classified as HQ or LQ clicks). Additionally, the detectability index (d') for HQ clicks for PAMGuard was 2.2 (overall d' = 2.0) versus 4.1 for PorCC (overall d' = 3.4). PorCC classification algorithm is a rapid and highly accurate method to classify NBHF clicks, which could be applied for real time monitoring, as well as to study harbour porpoises, and potentially other NBHF species, throughout their distribution range from data collected using towed hydrophones or static recorders. Moreover, PorCC is suitable for studies of acoustic communication of porpoises. © 2019 Acoustical Society of America. https://doi.org/10.1121/1.5110908

[WWA]

Pages: 3427-3434

I. INTRODUCTION

Studying harbour porpoises Phocoena phocoena in their natural environment is a difficult task. They are small and surface for only a few seconds at a time, travelling in groups of three or fewer animals, or as solitary individuals (Hammond et al., 2002). However, they are highly vocal (Linnenschmidt et al., 2013) and are therefore well suited for passive acoustic monitoring (PAM). Harbour porpoises produce highly stereotyped narrow-band high-frequency (NBHF) clicks for echolocation and communication. These clicks have peak and centroid frequencies between 100 and 160 kHz, centred around 130 kHz (Møhl and Andersen, 1973), with no spectral energy below 100 kHz (Hansen et al., 2008). The duration of individual clicks ranges from 50 μ s to 175 μ s and the half-power (-3 dB) bandwidth is around 15 kHz (Kyhn et al., 2010). Clicks are emitted in series, often referred to as "trains." A click train is loosely defined as "any series of clicks separated by gradually or cyclically changing inter-click interval suggesting a unit during an echolocation event or a communication signal" (Koschinski et al., 2008). Other odontocetes that produce NBHF clicks are all the porpoises (Phocoenidae), some dolphins of the Lissodelphininae subfamilily, pygmy, and dwarf

^{a)}Electronic mail: melania.cosentino@strath.ac.uk, ORCID: 0000-0001-7837-8669

J. Acoust. Soc. Am. 145 (6), June 2019

sperm whales (Kogiidae), and the river dolphin *Pontoporia blainvillei* (Galatius *et al.*, 2019).

Different PAM devices are used to study harbour porpoises, including animal-borne devices (Akamatsu et al., 2007), towed hydrophone arrays (e.g., Gillespie et al., 2005; Sveegaard et al., 2011), and static devices (e.g., Carlström, 2005; Carstensen et al., 2006). Static PAM devices can be roughly divided between those that record continuously (e.g., SoundTrap-Ocean Instruments, New Zealand) and click detectors or data loggers that only store information about the transient sounds detected, such as date and time, peak frequency, and amplitude (e.g., C-PODs; Chelonia Ltd., Cornwall, UK). C-PODs, and the earlier version T-POD, are used for a wide variety of studies, including seasonal and geographical changes in distribution (Verfuß et al., 2007) and response to anthropogenic noise (Carstensen et al., 2006; Pirotta et al., 2014). Moreover, they have been used to study porpoise acoustic behaviour (Koschinski et al., 2008), including diurnal variations in echolocation rates and click train patterns (Carlström, 2005). Continuous recordings at high sampling rates, required to record harbour porpoise clicks, generate an enormous amount of data and so the storage capacity limits the length of the data collection period. Moreover, data analysis is time consuming and so often part of the data remains unanalysed. On the other hand, click data loggers do not require high storage capacity and thus are more suitable for long-term studies and, since they are coupled with an automatic real-time classifier,

0001-4966/2019/145(6)/3427/8/\$30.00 © 2019 Acoustical Society of America 3427

the time invested in post-deployment data analysis is reduced significantly. However, it is not possible to carry out *post hoc* verification as the C-POD does not record the sound itself, and the detection and classification algorithms are not publicly available.

On the other hand, there are vast amounts of acoustic continuous recordings made using PAM systems that could be used to fill gaps in our understanding of harbour porpoise behaviour and communication in the wild. To that end, however, a classification system that can accurately and reliably identify harbour porpoise clicks is required. A classification system, in simple terms, assigns a given signal x to one of kpre-defined classes according to a series of parameters or functions. For continuous recordings, one of the most used harbour porpoise detector/classifier systems is PAMGuard's Click Detector and Classifier modules. PAMGuard is a modular, open source software designed to detect and classify marine mammal sounds (Gillespie et al., 2009), and it is used worldwide for a wide range of studies (Cucknell et al., 2016; Lawrence et al., 2016; Pinn et al., 2010). The standard settings of the classifier include a pre-filter (4th order digital Butterworth IIR 10 kHz high pass filter) and a trigger filter (4th order digital Chebyshev IIR 100-150 kHz band pass filter, pass band ripple 2.0). Clicks are classified as produced by porpoises by comparing the test band (110-150 kHz) to control bands (40-90 kHz and 160-190 kHz), with a 6 dB threshold ("general configuration file-porpoise click detection," available at www.pamguard.com). As an open source the software is regularly improved, and although the user can manage the settings, there is no available information about its performance. The precision (i.e., percentage of individual clicks correctly classified as porpoise clicks) reported for an earlier version of this classifier was between 37% and 74%, depending on the settings and background noise, while the proportion of missed clicks was not reported (Gillespie and Chappell, 2002). However, the performance of the current version remains unquantified. Additionally, the classifier requires manual verification after the identified clicks have been highlighted and clicks have to be manually selected in echolocation events in order to be extracted for further analysis (Cucknell et al., 2016; Lawrence et al., 2016). Alternatively, many researchers use custom-built classifiers (such as the KERNO classifier of the C-POD), of which neither the algorithm, nor the performance is available. As acoustic recordings continue to accumulate, assessing the performance of available classifiers for comparison purposes and automating these processes becomes essential.

The objective of this study was to develop a harbour porpoise click classifier (PorCC) that improves the performance of existing classifiers, reducing the occurrence of both false alarms and missed clicks, and that provides the user with a simple assessment of the quality of the classified click. PorCC was developed using the output of PAMGuard's Click Detector Module, and uses the coefficients of two logistic regression models to estimate a probability that a given signal was produced by a harbour porpoise. The predictor variables used to build the two logistic regression models were selected because these are the variables most commonly used to describe the temporal and spectral characteristics of harbour

3428 J. Acoust. Soc. Am. 145 (6), June 2019

porpoise vocalisations (e.g., Kyhn *et al.*, 2013). Once the probabilities are estimated, each signal is assigned to one of three categories: high-quality clicks (HQ), low-quality clicks (LQ), and high-frequency noise (N). These categories were defined based on the characteristics of the waveform, power spectrum, and spectrogram (Fig. 1). The performance of PorCC was tested against manually labelled samples from 18 h of data collected in two different seasons and in different background noise conditions. Additionally, the performance was tested against PAMGuard's Classifier in a subset of the dataset, consisting of 8 h of data from one summer day.

II. MATERIAL AND METHODS

A. Data collection

Acoustic data were collected during systematic surveys conducted in the West coast of Scotland, in the Firth of Clyde $(55.5254^\circ\ N, 4.9333^\circ\ W)$ during 25 survey days throughout all seasons, between 2016 (n=20) and 2017 (n=5), totalling over 210h of recordings. Surveys were carried out under sail or engine from the "Saorsa," a 40-ft sailing vessel. Transect lines were determined in advance and surveyed at a speed between 5 and 7 knots, in different weather conditions, during both day and night times. Surveys were terminated if the sea state reached >5. No concurrent visual observations were made. Recordings were made using a towed omnidirectional hydrophone array connected to the software PAMGuard (Gillespie et al., 2009) version 1.15.10, and digitised through a St. Andrews Instrumentation Ltd. data acquisition card with 16-bit resolution, at a sampling frequency of 500 kHz. The array included two Magrec HP03 hydrophone units, each comprising a spherical ceramic and a HP02 preamp, with a preamp high pass filter set at 2 kHz. The hydrophones had a sensitivity of -201 dB re $1 \text{ V}/\mu\text{Pa}$ at 150 kHz, and a flat response between 2 and 150 kHz. The array was towed using a Kevlarstrengthened 100 m long cable and the units were 25 cm apart.

PAMGuard's Click Detector Plug-In detects impulsive sounds (i.e., sounds of short duration with abrupt onset and rapid decay) over a given SNR threshold selected by the user (e.g., 6 dB). The detected sound is then saved as an individual audio clip, which also includes a very short recording period before and after the impulsive sound detected. All impulsive sounds detected in a given hour of recording are individually saved in one .pgdf file (for PAMGuard Data File) (Gillespie and Oswald 2017). For each audio clip, additional information is attached such as date and time time of arrival difference (i.e., delay) with respect to the reference hydrophone, and direction of arrival, estimated using trigonometric methods based on time of arrival differences (Gillespie and Chappell, 2002). By extracting individual clips from these files, two datasets were created, one to train PorCC and one to test its performance against manually labelled clips. Additionally, a subset of the testing data was used to compare the performance against PAMGuard's Classifier.

B. Training data

Three categories of signals were defined for the development of PorCC: high quality porpoise clicks (HQ), low-quality

Cosentino et al.



FIG. 1. (Color online) Examples of the categories defined to develop the harbour porpoise click classifier (PorCC). (a) High-quality harbour porpoise click (HQ), (b) low-quality harbour porpoise click (LQ), (c) high-frequency noise (N). Wigner plot (centre plot), waveform (lower plot), and power spectrum (right plot).

porpoise clicks (LQ), and high-frequency noise (N) (Fig. 1). HQ are polycyclic signals with peak frequency between 100 and 160 kHz, no spectral energy below 100 kHz, and duration around 100 μ s, matching the description of on-axis harbour porpoise clicks (Au *et al.*, 2006; Hansen *et al.*, 2008). LQ consist of signals slightly different to HQ, for example presenting notches in the power spectrum, or no clear beginning or end of the signal (low signal-to-noise ratio). Noise clips (N) are signals with peak and centroid frequencies between 100 and 160 kHz that do not share other characteristics with harbour porpoise clicks (e.g., oligocyclic, do have energy below 100 kHz).

Of the over 2 500 000 audio clips detected and saved by PAMGuard's Click Detector during the survey period, a subsample of 125416 (representing 5% of the total) was extracted using a random number generator to ensure they were independent from each other. In order to find good signals to develop the logistic regression models for the classifier (PorCC), an early version was used to assign to each clip a probability of being a harbour porpoise click. Those with high probability (≥ 0.9) were considered to be potential HQ, those with a probability between 0.5 and 0.9 were considered to be potential LQ, and those with a probability < 0.5 were considered to be potential N clips. Subsequently, from these, 5500 were randomly selected from their respective category to build two logistic regression models, thus 500 were potential HQ, 500 were potential LQ, and 4500 were potential N. In order to ensure each clip was a good representative of its respective category, all 5500 clips were then manually verified. Unrepresentative clips were discarded and replaced with clips randomly selected from the original subsample for that particular category.

J. Acoust. Soc. Am. 145 (6), June 2019

1. Logistic regression models

In simple terms, the logistic regression model estimates coefficients for each predictor variable in the model and the error term, from which a probability is derived. The predictor variables used were the duration of the signal (estimated as the 80% energy of the clip, μ s), peak frequency (PF, kHz) and centroid frequency (CF, kHz), -3 dB (BW-3dB, kHz) and root mean square bandwidths (Madsen and Wahlberg, 2007), and Q_{RMS} (ratio between CF and BW_{RMS}). Additionally, the ratio between peak and centroid frequencies (Ratio) and the peak value of a cross correlation (XC) performed against a typical harbour porpoise click were used. The click used for the cross correlation was extracted from the original dataset,¹ and was selected based on the waveform, power spectrum and spectrogram characteristics and peak-to-peak amplitude (162 dB re: 1µPa). Additionally, the waveform was consistent in both hydrophones, and the time of arrival difference between them was 0 (i.e., the orientation of the animal was perpendicular to the array). All predictor variables were explored for normality.¹ Multicollinearity, that is, when the predictor variables are correlated with each other, was tested using the Pearson χ^2 coefficient and none of the variable pairs had a correlation coefficient higher than ± 0.36 , except Q_{RMS} and XC that had a correlation of 0.49.1

The response variable for model 1 is binomial with the outcomes HQ/N and was built using 500 and 4500 clips of each, respectively. The response variable for model 2 is also binomial with the outcomes LQ/ N and was built using 500 and 4500 clips of each, respectively. The same N clips were used for both models. For each logistic regression model, a total of 63 models were tested as a series of reduced models

Cosentino et al. 3429
using all possible predictor variable combinations, and the best of each model was identified as the one with the lowest Akaike Information Criteria (AIC) value (Table I).¹

C. PorCC—Classification algorithm

The algorithm of the PorCC was written in MATLAB 2017a (The Math Works TM, Inc., Natick, MA) and runs on clips previously saved by PAMGuard's Click Detector Plug-In, analysing only those recorded by the first hydrophone on which they impinged. For each clip, the predictor variables identified in the model selection procedure are estimated and two probabilities are calculated using the coefficients obtained from the logistic regression models. Subsequently, a series of if/then statements is applied to assign the clip to one of the three categories previously defined (Fig. 2).

D. Testing data

To test the performance of PorCC, a dataset was created with all clips (n = 265 918) extracted from 5% of *.pgdf* files (i.e., 11 h of recordings, from ten survey days), which were selected randomly, and all clips (n = 284 231) from the 28th of August 2017 (i.e., 8h of recordings). Clips with peak and centroid frequencies between 100 and 160 kHz and $Q_{\rm RMS} > 4$ represented potential harbour porpoise clicks, and so these (n = 70 689) clips were extracted and manually labelled according to the three categories previously defined (Fig. 1), based on the characteristics of the waveform, power spectrum, and spectrogram. The overlap between the training and the testing data was of 442 clips. Subsequently, PorCC was used to classify the clips automatically by estimating the predictor variables and the probability-threshold values of 0.9999 and 0.55 (Fig. 2).

Confusion matrices and receiver operational characteristics (ROC) curves were used to assess the performance of PorCC. The hit rate was calculated for all categories as well as the rate of misclassification (i.e., false alarm and missed clicks) and the precision level. The hit rate is the number of HQ and LQ clips classified as HQ (strict criterion), or as

TABLE I. Series of logistic regression models for model 1 and model 2. Only the best five are shown here. See text for description of the variables used (footnote 1). The outcomes of the response variable for model 1 are high-quality harbour porpoise clicks or high frequency noise, and for model 2 are low-quality harbour (LQ) porpoise click or high-frequency noise (N). AIC = Akaike's Information Criterion.

ID	Predictor variables-Model 1	ΔΑΙΟ
1	Q _{RMS} + Duration	0
2	Q _{RMS} + Duration + Ratio	1.64
3	$Q_{RMS} + Duration + BW$	1.67
4	Q _{RMS} + Duration + CF	1.78
5	Q_{RMS} + Duration + XC	1.96
ID	Predictor variables-Model 2	ΔΑΙΟ
1	Q_{RMS} + Duration + Ratio + XC + CF + BW	0
2	Q_{RMS} + Duration + Ratio + XC + BW	1.19
3	Q _{RMS} + Duration + Ratio + CF + BW	19.19
4	Q_{RMS} + Duration + XC + CF + BW	20.07
5	$Q_{RMS} + \text{Duration} + \text{Ratio} + \text{BW}$	20.87

3430 J. Acoust. Soc. Am. 145 (6), June 2019



FIG. 2. Flowchart illustrating the decision-making pathway of the harbour porpoise click classifier (PorCC). CF = centroid frequency. PF = peak frequency. Th = probability thresholds. Prob = Probability.

either HQ or LQ (relaxed criterion), divided by the total number of non-N clips. Three analyses were performed: one with only HQ clips, one with only LQ clips, and one with both HQ and LQ clips, all against the N clips. The falsealarm rate is the number of N clips classified as HQ (strict criterion), or HQ or LQ (relaxed criterion) divided by the total number of N clips. In total, this results in six different points of operation in the ROC plot (strict or relaxed criterion combined with HQ, LQ, or both). The missed-clicks rate is defined as 1 minus the hit rate. The precision is defined as the number of clips classified invided by the total number of clips classified into that category.

E. PorCC vs PAMGuard

A subset of the testing dataset for PorCC was used to assess the performance of the porpoise click classifier builtin in PAMGuard and compared it to that of PorCC. This dataset subset contained all clips from the 28th of August 2017 (n = 284 231) of which 30 897 clips had already been manually labelled, having peak and centroid frequencies within the 100-160 kHz range, and Q>4. PAMGuard's classifier highlights potential harbour porpoise clicks and echoes that the user can manually verify and group into "acoustic events" to later extract them for further analysis. For the purpose of this study, all highlighted clicks were selected without manual verification, assigned to a unique acoustic event, and exported to an sqL database. Putative echoes were included because it was previously noted that this classifier sometimes misidentifies real harbour porpoise clicks as echoes (and vice versa) as well as for comparison purposes, as PorCC also identifies potential echoes (LQ). PAMGuard creates a table within the SQLite database, where information for each of the extracted potential harbour porpoise click is provided, including date, time, and an identification number within the .pgdf file where the waveform is saved. Using a custom-built script, and using the identification number, all clips identified by PAMGuard's classifier as potential harbour porpoise clicks were extracted from the .pgdf files and saved in a MATLAB structure array for further analysis. Subsequently, clips that were highlighted by

PAMGuard's classifier but were discarded by PorCC were manually labelled.

Confusion matrices and receiver operational characteristics (ROC) curves were used to assess the performance of PAMGuard and compare it against PorCC. False alarm, hit rates, and precision levels were also estimated, as well as the detectability index (d') (see, e.g., Egan, 1975; Tougaard, 2002).

III. RESULTS

A. Logistic regression models

According to AIC values, the best model 1 (for HQ signals) was that with only Q_{RMS} and duration as explanatory variables, while the best model 2 (for LQ signals) had five explanatory variables, Q_{RMS} , duration, ratio between peak and centroid frequency, cross-correlation coefficient, centroid frequency, and -3 dB bandwidth.¹

B. PorCC performance

PorCC classification process, including estimating all necessary parameters, takes approximately 1 ms per clip. Harbour porpoises produce between fewer than 10 and few hundred clicks per second depending on their behaviour (Clausen et al., 2010; Sørensen et al., 2018; Wright et al., 2017), PorCC shows, therefore, potential for real time application. For HQ, precision was 88.5% (4475 out of 5054, 519 of which were LQ and 60 N), false alarm (i.e., N classified as HQ) was 0.0001% (60 out of 537 591 N clips were classified as HQ), and 31.8% of clicks were missed (1710 were classified as LQ and 382 as N) (Table II). As precision increases hit rate decreases, that is fewer clicks, of the total available to the classifier, are going to be identified, demonstrating the well-known trade-off between errors: false alarms vs misses in signal detection and type I vs type II errors in conventional statistics (Fig. 3).

C. PorCC vs PAMGUARD

A total of 30 897 clips from the 28th of August met the criteria for potential harbour porpoise clicks, that is, having

TABLE II. Confusion matrices. Comparison of correct and misidentification levels between PorCC and the Porpoise Click Detector/Classifier Module in PAMGuard, and overall performance of PorCC. HQ = high-quality harbour porpoise clicks. LQ = low-quality harbour porpoise clicks. Noise = highand low-frequency noise (i.e., anything that is not a porpoise click).

Labelled		PorCC		PAMGuard			Р	orCC
	Total	HQ	Noise	HQ	Ν	Total	HQ	Noise
HQ	1833	564 ^a	1269 ^b	1209	113 ^b	6567	4475	382
Noise	965	477 + 1601 ^{a,c}	279,355	25°	280 034	537 591	60	533 228

^aOf the total of 3017 clips highlighted by PAMGuard as potential harbour porpoise clicks, 1601 had $Q_{\rm RMS}$ <4 and peak and centroid frequencies outside of the 100 and 160 kHz range, therefore they were not captured by PorCC, as they were discarded at the first step.

"False alarm (N clips classified as HQ clicks divided by the total number of N clips).

J. Acoust. Soc. Am. 145 (6), June 2019

peak and centroid frequency between 100 and 160 kHz and Q_{RMS} higher than 4. The results of the comparison of the performance of PorCC (using Th1 \geq 0.9999 and Th2 \geq 0.55) and PAMGuad's classifier for HQ are shown in Table II. Based on the detectability indexes (Fig. 3), PorCC outperforms PAMGuard's classifier in all cases, but especially for HQ clicks. The overall precision for HQ for PorCC was 30.8% for PAMGuard's classifier, assuming that PAMGuard's classifier correctly classified HQ and LQ as such in 100% of the cases, as once clicks are extracted from PAMGuard, there is no information of whether a clip was originally classified as a harbour porpoise click or an echo, which can be considered as equivalent to the HQ and LQ categories.¹

IV. DISCUSSION

The perfect classifier cannot exist, as detection always will be limited by noise, either external from the environment, or internal. For electronic systems this internal noise is in amplifiers and hydrophones, and for biological systems, this noise will be in the form of spontaneous activity in the neurons. In real-world applications, noise also comes in the form of substantial variation in the temporal and spectral characteristics of acoustic signals. These are affected by many factors, including background noise and the direction from where the signals impinges on the hydrophone, as well as by how the data were collected (e.g., hydrophone own noise, frequency characteristics of the hydrophones) (Richardson et al., 1995). Moreover, in this study, the performance of the classifier is intrinsically linked to the performance of the Click Detector Plug-In in PAMGuard, which in turn depends on the settings selected by the user (e.g., number of samples before and after the signal, SNR thresholds). Despite this, the results of this study show that a classification system based on logistic regression models to identify NBHF vocalisations produced by harbour porpoises outperforms existing classifiers. PorCC can achieve hit rates of over 90% while keeping the false alarm rate below 1% and maintaining high precision levels. The performance of PorCC is expected to be similar, or higher, in data collected using static devices, or in areas with low background noise. Moreover, it has potential for real time application, as it can analyse the equivalent of 1 h of data in under 1 min.

For both logistic regression models, one model was better than the others. It is worth noting that in both cases, the model with the cross-correlation coefficient (XC) as the only explanatory variable appears in the second position after Q, when looking at models with only one explanatory variable. To classify HQ clicks, cross-correlation analysis, which can be a time costly process, is not necessary and introduces a lot of variation as porpoise clicks are not blueprints of each other. In fact, the cross-correlation coefficient value ranged from 0.0038 to 4.5655, and thus using a threshold in a decision-making process would inevitably include HQ as well as N. The first model containing XC for click detection is fifth on the list. For LQ, on the other hand, XC explains more of the variance in the model, being necessary in the best model, and therefore helps in the classification process. There is likely to be both intra- and inter-animal variation in

^bMissed clicks (HQ clicks classified as N divided by the total number of HQ click). ^cFalse alarm (N clips classified as HQ clicks divided by the total number of



FIG. 3. Receiver operating characteristics (ROC) curves. Dots represent false alarm rates and hit rates associated with detection of HQ-clicks (solid black line), LQ-clicks (black dashed line) and both types combined (grey line), all against a background of N-clicks. Top figures show performance of PAMGuard. Curves are best fitting ROC-curves, generated under the assumption of Gaussian underlying distributions with equal variance. Bottom figures show performance by PorCC under two different criteria: strict (only clicks classified by PorCC as HQ) and relaxed (all clicks classified as either LQ or HQ). Figures to the left and right contain same data, but right figures are plotted on double probit (probability) axes.

signals, as well as substantial effects on the frequency spectrum caused by the directionality of the beam and the frequency dependent absorption in the water. This variation is illustrated by the differences in the recorded signals shown in Fig. 1, most evident by the lack of overlap in frequency spectra of the HQ and the LQ signals. For signals where the parameters are very variable, but where means may be more stable, other types of detectors can be predicted to outperform a cross-correlation receiver. One such receiver is a simple energy detector, which integrates energy within a specified frequency band and a specified duration (Green and Swets, 1966), and this is essentially what the HQclassifier of PorCC is.

3432 J. Acoust. Soc. Am. 145 (6), June 2019

Available data suggest that the variation pattern of interclick intervals within a click train is indicative of specific behaviours (Clausen *et al.*, 2010; Koschinski *et al.*, 2008; Wisniewska *et al.*, 2018; Sørensen *et al.*, 2018). This is especially true for foraging and feeding behaviour, characterised by inter-click intervals below 10 ms after a phase with much larger inter-click intervals (e.g., Koschinski *et al.*, 2008). PorCC's classification algorithm can be implemented in the output of any transient-sound detector for continuous recordings and, given the low misidentification levels, it is suitable to study the behaviour of wild harbour porpoises, as the variations in inter-click intervals can only be detected if the majority of clicks within a click train are identified.

Moreover, these studies can be carried out in data that has already been collected using continuous recordings at an adequate sampling rate, both using towed hydrophone arrays or static devices, such as SoundTrap. The PorCC classification algorithm, including the functions to estimate the different variables and the resulting coefficients, is publicly available and can be coded in other programming languages, such as Python. It could also be incorporated into PAMGuard.

PorCC, like other classifiers, is not exempt of errors, and trying to increase the hit rate would in turn lead to an increase in the false alarm rate, as seen in the change in performance going from a strict to a relaxed criterion. However, the ultimate goal in classification is not to avoid errors, but to manage them. Thus, PorCC provides the user with a general assessment of its performance through the ROC curves, as these show the changes in hit rate with false alarm variations (Tougaard, 2002), which results from using different threshold values to classify harbour porpoise clicks. Therefore, users can, a priori, manage the level of error according to their needs. Furthermore, depending on the objectives, the user can extract either or both HQ and LQ clicks as well as decide when LQ clicks should be ignored (e.g., single LQ clicks) or taken into account (e.g., studies of click train patterns).

The performance of PorCC for HQ clicks is very high, yet much lower for LQ clicks. This could be the result of some high-frequency noise clips having similar characteristics to LO clicks, which means the coefficients derived from the second logistic regression model are inefficient to distinguish between LQ clicks and high frequency noise. However, this low performance can also be the result of a level of subjectivity when assigning signals to these categories. This happens to be a fundamental limitation for almost all studies of this kind, where performance of detectors is evaluated on real world data. One must have some means of determining the "true state of the world," i.e., separating signals into those truly originating from porpoises and those that are just random noise. In this study, as in most others, we relied on the superior ability of the human brain to perform pattern recognition in noise and thus measure the performance of the detectors essentially against the performance of a skilled human observer. There is no objective way of determining whether a signal in the array recordings really originated from a porpoise or not. Only under extremely well controlled circumstances, such as when one has a single animal isolated in a pool and a recorder attached to the animal to monitor each and every vocalisation from the animal is it possible to evaluate the absolute detection performance of the detection system and even in such cases, one would suffer difficulties in transferring the experimental settings (limited depth and distance to receiver, training or habituation of the animal etc.) to the situation in real world monitoring.

V. CONCLUSIONS AND FUTURE WORK

The performance of PorCC greatly exceeds that of the currently available classifier in PAMGuard and has potential for real time application as well as to study the acoustic behaviour of harbour porpoises and other NBHF species in

J. Acoust. Soc. Am. 145 (6), June 2019

the wild, in data collected using both towed hydrophone arrays or static recorders.² Future work includes testing PorCC in data obtained using a different recording device (e.g., SoundTrap-Ocean Instruments, New Zealand) and under different survey conditions, and in recordings of harbour porpoises from another population. Additionally, the performance of PorCC will be tested against the performance of C-PODs in data collected simultaneously by a C-POD and a SoundTrap (Sarnocinska *et al.*, 2016).

ACKNOWLEDGMENTS

M.C. would like to thank M. Coscarella for his comments and advice during data analysis. The authors would like to thank P. Ramirez who helped labelling signals in early stages of development of PorCC. Data were collected by community volunteers from Clyde Porpoise CIC with assistance from NA Ventures Trust, Firstport, Bridgend Group and The Hunter Foundation. The research received funding from the Engineering and Physical Sciences Research Council (UK) through Grant No. EP/ N509760/1 and via the European Research Council under the European Union's Seventh Framework Programme (FP/ 2007-2013)/ERC Grant Agreement No. [615030]. J.T. was funded by the Danish Environmental Protection Agency. M.C. conceived the ideas, and analysed the data; M.C., J.T., J.F.C.W., and J.C.J. designed the methodology and interpreted the results; D.N. collected the data; M.C. led the writing of the manuscript. All authors contributed critically to the drafts and gave final approval for publication. Raw data were generated by the Clyde Porpoise CIC. Derived data supporting the findings of this study and the classification algorithm are available at the Pure Data Repository of the University of Strathclyde (https://doi.org/ 10.15129/e6066318-79e4-480c-9b4b-68b6771d89a8).

See supplementary material at https://doi.org/10.1121/1.5110908 for a figure (Fig. S1.1) of the click used as a model against which a crosscorrelation is performed in the PorCC classification algorithm; for the histogram and distributions of all variables used to develop the logistic regression models (Fig. S2.1); for a correlation plot (Fig. S2.2) of all variable pairs; for a complete list of all logistic regression models performed to develop the PorCC classification algorithm (Tables S1 and S2); for examples of signals misclassified by both PorCC (Fig. S3.1 to S3.6) and by PAMGuard's Click Classifier Module (FIG. 3.7).

²Preliminary results suggest that PorCC algorithm can be successfully applied to harbour porpoise data recorded with different devices and in different areas, as well as other NBHF species, such as Heaviside's dolphins (*Cephalorhynchus heavisidi*).

- Akamatsu, T., Teilmann, J., Miller, L. A., Tougaard, J., Dietz, R., Wang, D., Wang, K., Siebert, U., and Naito, Y. (2007). "Comparison of echolocation behaviour between coastal and riverine porpoises," Deep Sea Res. Pt II. 54, 290–297.
- Au, W. W. L., Kastelein, R. A., Benoit-Bird, K. J., Cranford, T. W., and McKenna, M. F. (2006). "Acoustic radiation from the head of echolocating harbor porpoises (*Phocoena phocoena*)," J. Exp. Biol. 209, 2726–2733.
- Carlström, J. A. K. (2005). "Diel variation in echolocation behavior of wild harbor porpoises," Mar. Mammal Sci. 21(1), 1–12. Carstensen, J., Henriksen, O. D., and Teilmann, J. (2006). "Impacts of off-
- Carstensen, J., Henriksen, O. D., and Teilmann, J. (2006). "Impacts of offshore wind farm construction on harbour porpoises: Acoustic monitoring of echolocation activity using porpoise detectors (T-PODs)," Mar. Ecol. Prog. Ser. 321, 295–308.

- Clausen, K. T., Wahlberg, M., Beedholm, K., Deruiter, S. L., and Madsen, P. T. (2010). "Click communication in harbour porpoises Phocoena *phocora*," Bioacoustics **20**, 1–28. Cucknell, A.-C., Boisseau, O., Leaper, R., Mclanaghan, R., and Moscrop, A
- (2016). "Harbour porpoise (*Phocoena phocoena*) presence, abundance and distribution over the Dogger Bank, North Sea, in winter," J. Mar. Biol. Assoc. U. K. 97(7), 1455-1465.
- Egan, J. P. (1975). Signal Detection Theory and ROC Analysis (Academic, New York).
- Galatius, A., Olsen, M. T., Steeman, M. E., Racicot, R. A., Bradshaw, C. D., Kyhn, L. A., and Miller, L E. E. A. (2019). "Raising your voice: Evolution of narrow-band high- frequency signals in toothed whales (Odontoceti)," Biol. J. Linn. Soc. 126, 231-224.
- Gillespie, D., Berggren, P., Brown, S., Kuklik, I., Lacey, C., Lewis, T Matthews, J., McLanaghan, R., Moscrop, A., and Tregenza, N. J. C. (2005). "Relative abundance of harbour porpoises (*Phocoena phocoena*) from acoustic and visual surveys of the Baltic Sea and adjacent waters during 2001 and 2002," J. Cetacean Res. Manage. 7(1), 51–57. Gillespie, D., and Chappell, O. (2002). "An automatic system for detecting
- and classifying the vocalisations of harbour porpoises," Bioacoustics 13, 37-61.
- Gillespie, D., Gordon, J. C. D., Mchugh, R., Mclaren, D., Mellinger, D. K., Redmond, P., Thode, A., Trinder, P., and Deng, X. Y. (2009). "PAMGUARD: Semiautomated, open source software for real-time acoustic detection and localisation of cetaceans," J. Acoust. Soc. Am. 125,
- Gillespie, D., and Oswald, M. (2017). Binary file structure for PAMGUARD detector output. Version 4. 0 1 Introduction 2 File Format. Green, D. M., and Swets, J. A. (1966). Signal Detection Theory and Psychophysics (John Wiley, Oxford, UK).
- Hammond, P. S., Berggren, P., Benke, H., Borchers, D. L., Collet, A., Heide-Jørgensen, M. P., Heimlich, S., Hiby, A. R., Leopold, M. F., and Michael M. (2002), "Abundance of harbour porpoise and other cetaceans in the North Sea and adjacent waters," J. Appl. Ecol. 39, 361–376.
 Hansen, M., Wahlberg, M., and Madsen, P. T. (2008), "Low-frequency com-
- ponents in harbor porpoise (*Phocoena Phocoena*) clicks: Communication signal, by-products, or artifacts?," J. Acoust. Soc. Am. 124(6), 4059-4068.
- Koschinski, S., Diederichs, A., and Amundin, M. (2008). "Click train patterns of free-ranging harbour porpoises acquired using T-PODs may be useful as indicators of their behavior," J. Cetacean Res. Manage. 10(2), 147_155
- Kyhn, L. A., Jensen, F. H., Beedholm, K., Tougaard, J., Hansen, H., and Madsen, P. T. (2010). "Echolocation in sympatric Peale's dolphins (Lagenorhynchus australis) and Commerson's dolphins (Cephalorhynchus mersonii) producing narrow-band high-frequency clicks," J. Exp. Biol. 213(11), 1940-1949.
- Kyhn, L. A., Tougaard, J., Beedholm, K., Jensen, F. H., Ashe, E., Williams, R., and Madsen, P. T. (2013). "Clicking in a killer whale habitat: Narrow-band,

high-frequency biosonar clicks of harbour porpoise (*Phocoena phocoena*) and Dall's porpoise (*Phocoenoides dalli*)," PLoS One **8**(5), 0063763.

- Lawrence, J. M., Armstrong, E., Gordon, J. C. D., Lusseau, S. M., and Fernandes, P. (2016). "Passive and active, predator and prey: Using acoustics to study interactions between cetaceans and forage fish," ICES J. Mar. Sci. 73(8), 2075-2084.
- Linnenschmidt, M., Teilmann, J., Akamatsu, T., Dietz, R., and Miller, L.A. (2013). "Biosonar, dive, and foraging activity of satellite tracked harbor porpoises (*Phocoena phocoena*)," Mar. Mammal Sci. 29(2), E77-E97
- Madsen, P. T., and Wahlberg, M. (2007). "Recording and quantification of ultrasonic echolocation clicks from free-ranging toothed whales," Deep Sea Res. Part I 54, 1421–1444.
- Møhl, B., and Andersen, S. (1973). "Echolocation: High-frequency compo-nent in the click of the harbour porpoise (*Phocoena phocoena*)," J. Acoust. Soc. Am. 54(5), 1368-1379.
- Pinn, E., Tasker, M., Mendes, S., and Goold, J. (2010). "Maintaining favourable conservation status of harbour porpoise in UK waters," Joint Nature Conservation Committee, Aberdeen, UK, 17 pp.
- Pirotta, E., Brookes, K. L., Graham, I. M., and Thompson, P. M. (2014). "Variation in harbour porpoise activity in response to seismic survey noise," Biol. Lett. 10, 20131090.
- Richardson, J. W., Greene, C. R., Jr., Malme, C. I., and Thomson, D. H. (1995). Marine Mammals and Noise (Academic, San Diego, CA).
- Sarnocinska, J., Tougaard, J., Johnson, M. P., Madsen, P. T., and Wahlberg, M. (2016). "Comparing the performance of C-PODs and SoundTrap/ M. (2010). Comparing the performance of CFODS and Sound rap/ PAMGUARD in detecting the acoustic activity of harbor porpoises (*Phocoena phocoena*)," Proc. Mtgs. Acoust. 27, 070013.
 Sørensen, P. M., Wisniewska, D. M., Jensen, F. H., Johnson, M. P., Teilmann, J., and Madsen, P. T. (2018). "Click communication in wild harbour porpoises (*Phocoena phocoena*)," Sci. Rep. 8, 9702.
 Sveegaard, S., Teilmann, J., Berggren, P., Mouritsen, K. N., Gillespie, D., act Taverde L. (2011). "Acoustic neuron conferent the high density acoustic act Taverde L. (2012)."
- and Tougaard, J. (2011). "Acoustic surveys confirm the high-density areas of harbour porpoises found by satellite tracking," ICES J. Mar. Sci. 68(5), 929-936.
- Tougaard, J. (2002). "Signal detection theory, detectability and stochastic resonance effects," Biol. Cybernet. 87, 79–90.
- Verfuß, U., Christopher, K., Honnef, G., Meding, A., Dähne, M., Mundry, R., and Benke, H. (2007). "Geographical and seasonal variation of harbour porpoise (*Phocoena phocoena*) presence in the German Baltic Sea revealed by passive acoustic monitoring," J. Mar. Biol. Assoc. U. K. 87, 165-176.
- Wisniewska, D. M., Johnson, M. P., Teilmann, J., Siebert, U., Galatius, A., Dietz, R., and Madsen, P. T. (2018). "High rates of vessel noise exposure on wild harbour porpoises (*Phocoena phocoena*) can disrupt foraging," Proc. R. Soc. B 285, 20172314.
- Wright, A. J., Akamatsu, T., Teilmann, J., Mouritsen, K. N., Sveegaard, S., and Dietz, R. (**2017**). "Silent porpoise: Potential sleeping behaviour identi-fied in wild harbour porpoise," Anim. Behav. **133**, 211–222.

Appendix B

Table A1. Series of logistic regression models for Model 1. See text for description of the variables used. The outcomes of the response variable for Model 1 are high-quality harbour (HQ) porpoise click or high-frequency noise (N). AIC = Akaike's Information Criterion.

ID	Predictor Variables	ΔΑΙC
1	Q _{RMS} + Duration	0
2	$Q_{RMS} + Duration + Ratio$	1.64
3	Q_{RMS} + Duration + BW	1.67
4	Q_{RMS} + Duration + CF	1.78
5	Q_{RMS} + Duration + XC	1.96
6	Q_{RMS} + Duration + Ratio + BW	3.34
7	$Q_{RMS} + Duration + CF + BW$	3.46
8	$Q_{RMS} + Duration + Ratio + CF$	3.51
9	Q_{RMS} + Duration + Ratio + XC	3.64
10	Q_{RMS} + Duration + XC + BW	3.64
11	Q_{RMS} + Duration + XC + CF	3.73
12	$Q_{RMS} + Duration + Ratio + CF + BW$	5.21
13	Q_{RMS} + Duration + Ratio + XC + BW	5.33
14	$Q_{RMS} + Duration + XC + CF + BW$	5.41
15	Q_{RMS} + Duration + Ratio + XC + CF	5.50
16	$Q_{RMS} + Duration + Ratio + XC + CF + BW$	7.20
17	$Q_{RMS} + Ratio + XC + CF + BW$	108.96
18	$Q_{RMS} + Ratio + XC + BW$	109.15

19	$Q_{RMS} + XC + CF + BW \\$	110.57
20	$Q_{\text{RMS}} + XC + BW$	111.85
21	$Q_{RMS} + Ratio + BW$	116.14
22	$Q_{RMS} + Ratio + CF + BW$	116.52
23	$Q_{RMS} + CF + BW \\$	118.26
24	$Q_{RMS} + BW$	119.09
25	$Q_{RMS} + Ratio + XC + CF$	141.89
26	$Q_{RMS} + Ratio + XC$	143.13
27	$Q_{RMS} + XC + CF$	144.38
28	$Q_{RMS} + XC$	145.08
29	$Q_{RMS} + Ratio + CF$	153.37
30	$Q_{RMS} + Ratio$	153.86
31	$Q_{RMS} + CF$	155.41
32	Qrms	155.76
33	Duration + Ratio + XC + BW	374.23
34	Duration + Ratio + XC + CF + BW	376.23
35	Duration + Ratio + XC	384.45
36	Duration + Ratio + XC + CF	386.38
37	Duration + XC + BW	455.57
38	Duration + XC + CF + BW	456.22
39	Duration + XC	472.27
40	Duration + XC + CF	473.66
41	Ratio + XC + CF + BW	632.67

42	Ratio + XC + BW	633.62
43	Ratio + XC + CF	649.11
44	Ratio + XC	649.47
45	XC + CF + BW	731.51
46	XC + BW	741.19
47	XC + CF	742.95
48	XC	754.22
49	Duration + Ratio + BW	1022.92
50	Duration + Ratio + CF + BW	1024.33
51	Duration + Ratio	1166.78
52	Duration + Ratio + CF	1168.77
53	Duration + CF + BW	1266.04
54	Duration + BW	1272.47
55	Duration + CF	1449.11
56	Duration	1449.55
57	Ratio + CF + BW	2608.89
58	Ratio + CF	2618.05
59	Ratio + BW	2756.10
60	Ratio	2768.74
61	CF + BW	2978.54
62	CF	2981.04
63	BW	3213.82

ID	Predictor Variables	ΔΑΙC
1	$Q_{RMS} + Duration + Ratio + XC + CF + BW$	0
2	$Q_{RMS} + Duration + Ratio + XC + BW$	1.19
3	$Q_{RMS} + Duration + Ratio + CF + BW$	19.19
4	$Q_{RMS} + Duration + XC + CF + BW \\$	20.07
5	Q_{RMS} + Duration + Ratio + BW	20.87
6	$Q_{RMS} + Duration + XC + BW$	22.21
7	$Q_{RMS} + Duration + CF + BW$	38.05
8	Q_{RMS} + Duration + BW	41.05
9	$Q_{RMS} + Duration + Ratio + XC + CF$	41.33
10	$Q_{RMS} + Duration + Ratio + XC$	43.84
11	Q_{RMS} + Duration + XC + CF	63.10
12	$Q_{RMS} + Duration + Ratio + CF$	63.57
13	$Q_{RMS} + Duration + Ratio$	66.89
14	Q_{RMS} + Duration + XC	67.66
15	Q_{RMS} + Duration + CF	83.99
16	Q_{RMS} + Duration	89.98
17	$Q_{RMS} + Ratio + XC + CF$	366.58
18	$Q_{RMS} + Ratio + XC + CF + BW$	368.58
19	$Q_{RMS} + Ratio + XC$	368.88

Table A2. Series of logistic regression models for Model 2. See text for description of the variables used. The outcomes of the response variable for Model 2 are low-quality harbour (LQ) porpoise click or high-frequency noise (N). AIC = Akaike's Information Criterion.

20	$Q_{RMS} + Ratio + XC + BW$	370.82
21	$Q_{\text{RMS}} + \text{Ratio} + CF$	374.82
22	$Q_{RMS} + Ratio + CF + BW$	376.78
23	$Q_{RMS} + Ratio$	376.86
24	$Q_{RMS} + Ratio + BW$	378.86
25	$Q_{RMS} + XC + CF$	410.75
26	$Q_{RMS} + XC + CF + BW \\$	412.74
27	$Q_{RMS} + XC$	413.15
28	$Q_{RMS} + XC + BW \\$	415.10
29	$Q_{RMS} + CF$	418.06
30	$Q_{\rm RMS} + CF + BW \\$	419.98
31	Qrms	419.99
32	$Q_{\rm RMS} + BW$	421.99
33	Duration + Ratio + XC + CF + BW	1042.02
34	Ratio + XC + CF + BW	1103.10
35	Duration + Ratio + XC + CF	1124.00
36	Duration + Ratio + XC + BW	1129.92
37	Ratio + XC + CF	1143.50
38	Ratio + XC + BW	1199.11
39	Duration + Ratio + XC	1206.99
40	Ratio + XC	1233.64
41	Duration + XC + CF + BW	1296.13
42	XC + CF + BW	1350.30

43	Duration + Ratio + CF + BW	1387.39
44	Duration + XC + CF	1394.01
45	Ratio + CF + BW	1394.34
46	XC + CF	1407.15
47	Duration + XC + BW	1472.01
48	Duration + Ratio + CF	1500.27
49	Ratio + CF	1500.61
50	XC + BW	1531.41
51	Duration + XC	1553.00
52	Duration + Ratio + BW	1556.81
53	Ratio + BW	1560.40
54	XC	1573.10
55	Duration + Ratio	1657.53
56	Ratio	1658.48
57	Duration + CF + BW	1679.28
58	CF + BW	1680.73
59	Duration + CF	1809.32
60	CF	1815.33
61	BW	1960.22
62	Duration + BW	1961.49
63	Duration	2061.31

Appendix C

The algorithm below extracts underlying patterns by removing unwanted sources of sounds in click trains in order to later identify behaviours.

Abbreviations:

- Amp: Amplitude (dB re: 1μ Pa).
- **CF**: Centroid frequency (kHz).
- **CPS**: Clicks per second.
- **CTTemp**: Temporary click train. This is the click train being treated.
- **Fs**: Sampling frequency (Hz).
- ICI: inter-click interval (ms).
- MaxDiffSorted: The maximum difference (*i.e.*, jump) in CPS, when sorted in ascending order. It indicates whether there are outliers (*e.g.*, multipath signals).
- NewCT: New click train. This is the resulting click train.

ALGORITHM

```
% Add row numbers for a later step
CTTemp.RowN(1,1) = 1; CTTemp.RowN(1:end,1) = 1:size(CTTemp,1);
% Save in another variable for a later step
CTTemp1 = CTTemp;
% A large maximum difference indicates the presence of outliers
SortedCPS = sort(CTTemp.CPS); DiffSorted = diff(SortedCPS);
[MaxDiffSorted, ~] = max(DiffSorted);
% A small maximum difference indicates a good click train
if MaxDiffSorted < 28
   % Do nothing
else
    Outlier = isoutlier (CTTemp.CPS);
    HighCPS = CTTemp.CPS > 50;
    Both = Outlier + HighCPS;
    CTTemp(Both == 2,:) = [];
   % The function 'NewICI' estimates the ICI and CPS of all
       clicks in the click train
    CTTemp = NewICI(CTTemp, Fs);
    SortedCPS = sort(CTTemp.CPS);
    DiffSorted = diff(SortedCPS);
   % Recalculate the maximum difference
    [MaxDiffSorted, ~] = max(DiffSorted);
end
% If after removing outliers the maximum difference is small
if MaxDiffSorted > 10 && MaxDiffSorted < 28 && size(CTTemp,1) <
   100
```

```
NewCT = CTTemp; \% This step ends here
  NewCT = NewICI(NewCT, Fs);
else
   % Remove local minimum - these are clicks with amplitude
       smaller that the previous and following clicks, with at
       least 5 dB difference with the previous click
    S1 = size(CTTemp, 1); S2 = 1;
    while S1 \tilde{} = S2
        S1 = size(CTTemp, 1);
        DiffAmps(1,1) = 0;
        DiffAmps(2:S1,1) = diff(CTTemp.Amp);
        LocMin = islocalmin (CTTemp.Amp);
        LargeDiff = DiffAmps < -5;
        DeleteRows = LocMin + LargeDiff;
        CTTemp(DeleteRows == 2, :) = [];
        CTTemp = NewICI(CTTemp, Fs);
        S2 = size(CTTemp, 1);
    end
end
SortedCPS = sort(CTTemp.CPS);
DiffSorted = diff(SortedCPS);
[MaxDiffSorted, ~] = max(DiffSorted);
if MaxDiffSorted > 35 && size(CTTemp,1) > 100
    % Finding the stable areas with low CPS variation
     CPSDiff = diff(CTTemp.CPS);
     PercChangeS = (CPSDiff./CTTemp.CPS(1:end-1))*100;
```

```
PercChangeS = abs(PercChangeS(2:end));
PercCPSDiff = movmean(PercChangeS, 4); \% Moving average
StartRow = find (PercCPSDiff < 5);
StartRow = StartRow + 2;
DiffStartRow = diff(StartRow);
Here = find (DiffStartRow == 1);
\% Go into the CT
Keep = 1;
if size (StartRow, 1) == 0
  RowN = 1:
else
 RowN = StartRow(Here(1)); % Low variability in CPS (for
     the next 4)
end
RowsToKeep(1,1) = RowN;
FirstRowN = RowN;
% Look backwards
while RowN > 10
ClickCPS = CTTemp.CPS(RowN, 1);
ClickAmp = CTTemp.Amp(RowN, 1);
ClickStartSample = CTTemp.startSample(RowN, 1);
ICIs = abs(ClickStartSample - CTTemp.startSample(RowN-9:
   RowN-1,1))/(Fs/1000);
CPSs = 1000. / ICIs;
Amps = CTTemp.Amp(RowN-9:RowN-1,1);
Amps = abs(ClickAmp - Amps);
```

```
DiffCPSs = abs(ClickCPS - CPSs);
[, ixCPS] = min(DiffCPSs); % Position of the next click
    if Amps(ixCPS) < 5
      DiffCPSs(ixCPS) = 1000;
      [, ixCPS] = min(DiffCPSs); % Position of the next
         click
      RowN = RowN - ixCPS;
    else
      RowN = RowN - ixCPS;
    end
Keep = Keep + 1;
RowsToKeep(Keep, 1) = RowN;
end
% Look forwards
RowN = FirstRowN;
while CTTemp.ICI(RowN,1) < 1000 & RowN < size(CTTemp,1)-10
ClickCPS = CTTemp.CPS(RowN, 1);
ClickAmp = CTTemp.Amp(RowN, 1);
ClickStartSample = CTTemp.startSample(RowN, 1);
ICIs = abs(CTTemp.startSample(RowN+1:RowN+9,1) -
   ClickStartSample)/(Fs/1000);
CPSs = 1000. / ICIs;
Amps = CTTemp.Amp(RowN+1:RowN+9,1);
Amps = abs(Amps - ClickAmp);
DiffCPSs = abs(CPSs - ClickCPS);
[, ixCPS] = min(DiffCPSs); % Position of the next click
```

```
if Amps(ixCPS) < 5
          DiffCPSs(ixCPS) = 1000;
          [~, ixCPS] = min(DiffCPSs); % Position of the next
              click
          RowN = RowN + ixCPS;
        else
          RowN = RowN + ixCPS;
        end
    Keep = Keep + 1;
    RowsToKeep(Keep, 1) = RowN;
    end
% Delete loose clicks
       L1 = size(CTTemp, 1);
       L2 = 1;
   while L2 \tilde{} = L1
      L1 = size(CTTemp, 1);
      CTTemp = NewICI(app, CTTemp, Fs);
      LooseClicks = find(CTTemp.ICI > 250);
      Positions = find (diff (LooseClicks) == 1);
      RowsToDelete = LooseClicks(Positions);
      CTTemp(RowsToDelete, :) = [];
      L2 = size(CTTemp, 1);
   end
RowsToKeep = sort (unique (RowsToKeep));
NewCT = CTTemp(RowsToKeep, :);
NewCT = NewICI(NewCT, Fs);
```

```
else
NewCT = CTTemp;
NewCT = NewICI(NewCT, Fs);
end
RowsLeft = NewCT.RowN(:,1);
CTTemp2 = CTTemp1;
CTTemp2(RowsLeft,:) = [];
end
```